Development of suitable *in vitro* models of the bronchial epithelium for inhalation toxicology

Der Naturwissenschaftlichen Fakultät der Gottfried Wilhelm Leibniz Universität Hannover zur Erlangung des Grades Doktor der Naturwissenschaften Dr. rer. nat. genehmigte Dissertation von

> Dipl.-Biol. Jessica Rach geboren am 08.01.1977 in Gehrden

Referent:	Prof. Dr. Thomas Scheper
	Institut für Technische Chemie
	Gottfried Wilhelm Leibniz Universität Hannover

Korreferentin:Prof. Dr. Cornelia KasperDepartment für BiotechnologieUniversität für Bodenkultur (BOKU) Wien, Österreich

Tag der Promotion: 14.06.2013

Erklärung

Ich versichere, dass ich diese Dissertation selbstständig verfasst habe und alle benutzten Hilfsmittel sowie evtl. zur Hilfeleistung herangezogene Institutionen vollständig angegeben wurden. Diese Arbeit wurde nicht als Diplomarbeit oder ähnliche Prüfungsarbeit verwendet.

Hannover, im April 2013

Danksagung

An erster Stelle möchte ich mich bei Frau Prof. Aufderheide, Herrn Prof. Mohr, Herrn Prof. Scheper und Frau Prof. Kasper, bedanken, ohne die diese Arbeit nicht möglich gewesen wäre.

Frau Prof. Aufderheide und Herrn Prof. Mohr danke ich für die Bereitstellung des sehr interessanten und aktuellen Themas und die dafür nötigen Hilfsmittel, für die fachliche Beratung und Hilfestellung und besonders dafür, dass sie stets ein offenes Ohr für mich hatten und mich unterstützt und motiviert haben.

Herrn Prof. Scheper und Frau Prof. Kasper danke ich für die Möglichkeit diese Arbeit am Institut für Technische Chemie absolvieren zu können, für die freundliche Betreuung und dafür, dass sie mir das Vertrauen geschenkt haben, diese Arbeit selbstständig und erfolgreich zu bearbeiten.

Ich danke Herrn Prof. Figuth (Abteilung für Thorax- und Gefäßchirurgie, Oststadtkrankenhaus Hannover), Dr. Pfennig (Pathologie, KRH Hannover) und ihren Mitarbeitern für die Bereitstellung von Lungengewebe und die damit verbundene Aufklärung der Patienten.

Des Weiteren möchte ich mich bei dem gesamten Cultex Team für die gute Zusammenarbeit und das angenehme Arbeitsklima in den letzten Jahren bedanken.

Ein ganz besonderer Dank geht an Birgit Heise für die hervorragende Arbeit, die sie im Labor verrichtet hat und für ihre Flexibilität und Geduld, die es mir ermöglicht haben, die vielen spontanen Ideen in die Tat umzusetzen.

Olaf Krischenowski danke ich für seine unschätzbare, vielfältige Hilfe bei technischen und handwerklichen Schwierigkeiten und Conni Wicht, Sabrina Meine und Jaqueline Scharrer für die zusätzliche Unterstützung im Labor und die Aufheiterungen in den Pausen. Ich danke Steffi Scheffler, Niklas Möhle und Jessica Budde für die lebhaften wissenschaftlichen Diskussionen und Anregungen. Steffi danke ich außerdem für das gründliche Korrekturlesen, die hilfreichen Anmerkungen und die motivierenden Worte.

Bei meinen lieben Freunden, besonders Diana Ganguin und Kim Domsalla, möchte ich mich für die aufmunternden Worte und die vielen schönen und fröhlichen Momente bedanken, die mir immer neue Kraft und Motivation gegeben haben.

Last but not least danke ich meiner Familie- meinen Eltern Karola und Hans-Werner Rach, meiner Schwester Sandra Homeier und meinen Großeltern Renate und Rudi Schreiber- und meinem Freund Kai Kamm dafür, dass sie immer an mich geglaubt haben und entscheidend dazu beigetragen haben, dass ich meine Ziele nie aus den Augen verloren habe. Danke für eure Liebe und Unterstützung!

Content

Lis	st of Figu	ures	
Lis	t of Tab	bles	
Ab	ostract		1
Ku	ırzfassur	ng	3
1	Gene	ral Introduction	5
2	Scope	e and objectives of the thesis	7
3	Theor	retical Framework	9
	3.1 T	oxicological Testing	9
	3.2 L	egislative Regulations for Animal Experiments	11
	3.3 A	Alternative Methods	
	3.3.1	Integrated Testing Strategies (ITS)	13
	3.3.2	Tissue Engineering	
	3.3.3	Cell Culture	15
	3.3	.3.1 Three-dimensional cell culture	
	3.4 T	The Respiratory System	
	3.4.1	The Lung	
	3.4.2	The lung epithelium	21
	3.4	.2.1 Function and role of the lung epithelium	24
	3.4.3	Prevalence and incidence of airway diseases	
	3.5 Ir	n vitro Models of the Airway Epithelium	
	3.5.1	Normal human bronchial epithelial cells	
	3.5.2	Human continuous airway epithelial cell lines	
	3.5	.2.1 16HBE140	
	3.5	.2.2 Calu-3	32
	3.5	.2.3 ChaGo-K1	
	3.5	.2.4 A549	

3.5.3	Three-dimensional cultures and co-culture models of the airways	33
3.5.4	Cultivation of airway epithelial cell cultures	34
3.6 Aut	tomated cultivation systems	37
3.6.1	The CULTEX [®] Long-Term Cultivation (LTC) System	39
3.6.2	Testing of materials for <i>in vitro</i> devices	41
4 Experim	nental Work	43
4.1 Eva	luation of cell lines and primary cells of the bronchial epithelium	44
4.1.1	Continuous epithelial lung cell lines	46
4.1.1.	1 ChaGo-K1	46
4.1.1.	2 Calu-3	48
4.1.1.	3 16HBE14o	53
4.1.1.	4 Uncharacterized immortalized clonal cell lines	62
4.1.2	Normal human bronchial epithelial cell lines	67
4.1.2.	1 Commercial normal human bronchial epithelial cells (HBEpC)	67
4.1.2.	2 Isolation of bronchial epithelial cells from normal human lung tissue	74
4.1.2.	3 Establishment of normal human bronchial epithelial (NHBE) cell lines	82
4.1.2.	4 Characterization of normal human bronchial epithelial (NHBE) cell line	s.84
4.1.3	Human lung fibroblasts	88
4.1.3.	1 IMR90- fetal human lung fibroblasts	88
4.1.3.	2 Establishment of normal human lung fibroblast cell lines (NHLF)	88
4.2 Dev	velopment of 3-dimensional in vitro models of the airway epithelium	92
4.2.1	Sandwich Cell Model (SCM)	93
4.2.2	Compartment Cell Model (CCM)	. 108
4.3 Stir	nulation of NHBE cells by special additives in the culture medium	. 120
4.3.1	Effects of retinoic acid deficiency in NHBE cultures	. 122
4.3.2	Effects of different EGF concentrations in NHBE cultures	. 123
4.3.3	Effects of TGF-ß1 stimulation in NHBE cultures	. 131
4.3.4	Effects of IL-13 stimulation in NHBE cultures	. 134

1.4	The	CULTEX [®] Long Term Cultivation (LTC) System	
4.4	.1	Material testing	
4.4	.2	Application of the CULTEX [®] Long Term Cultivati	on (LTC) System140
Sur	nmar	y and Final Conclusions	
Арі	pendi	x	
5.1	Met	hods	
6.1	.1	Cultivation and handling of cell cultures	
6	511	1 Cell culture media	150
6	5112	2 Subcultivation (nassaging) of cells	154
6	5.1.1.3	3 Cryopreservation of cells and tissue	
6	5.1.1.4	4 Coating of cell culture insert membranes o	r culture vessels 157
6.1	.2	Preparation of co- cultures according to the Sar	dwich Cell Model (SCM) 159
6.1	.3	Preparation of co-cultures according to the C	ompartment Cell Model (CCM)
var	iant I	I	
6.1	.4	Material testing	162
		-	
6.1	.5	Analytical methods	
6.1 6	.5 5.1.5.1	Analytical methods	
6.1 6 6	.5 5.1.5.2 5.1.5.2	Analytical methods 1 Histology 2 Immunohistochemistry	
6.1 6 6	.5 5.1.5.2 5.1.5.2	Analytical methods 1 Histology 2 Immunohistochemistry 3 Immunofluorescence	
6.1 6 6 6	.5 5.1.5.2 5.1.5.2 5.1.5.2	Analytical methods 1 Histology 2 Immunohistochemistry 3 Immunofluorescence 4 TEER measurement	162
6.1 6 6 6 6	.5 5.1.5.2 5.1.5.2 5.1.5.4 5.1.5.4	Analytical methods Analytical methods Histology Immunohistochemistry Immunofluorescence TEER measurement ELISA	162
6.1 6 6 6 6 5.2	.5 5.1.5.2 5.1.5.2 5.1.5.4 5.1.5.9 6.1.5.9 Abb	Analytical methods Histology Immunohistochemistry Immunofluorescence TEER measurement ELISA previations	162
6.1 6 6 6 5.2 5.3	.5 5.1.5.2 5.1.5.2 5.1.5.4 5.1.5.4 5.1.5.9 Labe	Analytical methods Histology Immunohistochemistry Immunofluorescence TEER measurement ELISA oreviations	162
6.1 6 6 6 5.2 5.3	.5 5.1.5.2 5.1.5.2 5.1.5.4 5.1.5.4 5.1.5.5 Abb Labo Anti	Analytical methods Histology Immunohistochemistry Immunofluorescence TEER measurement ELISA reviations oratory devices	162
6.1 6 6 6 5.2 5.3 5.4	.5 5.1.5.2 5.1.5.2 5.1.5.4 5.1.5.4 5.1.5.4 Abb Labo Anti	Analytical methods Histology Immunohistochemistry Immunofluorescence TEER measurement ELISA oreviations oratory devices ibodies	162
6.1 6 6 6 5.2 5.3 5.4 5.5 5.6	.5 5.1.5.2 5.1.5.2 5.1.5.4 5.1.5.4 5.1.5.5 Abb Labo Anti Labo	Analytical methods 1 Histology 2 Immunohistochemistry 3 Immunofluorescence 4 TEER measurement 5 ELISA oratory devices	162
6.1 6 6 6 5.2 5.3 5.4 5.5 5.6 5.7	.5 5.1.5.2 5.1.5.2 5.1.5.4 5.1.5.4 5.1.5.4 6.1.5.4 Abb Labo Labo Labo Labo	Analytical methods 1 Histology 2 Immunohistochemistry 3 Immunofluorescence 4 TEER measurement 5 ELISA oratory devices	162 163 163 163 166 168 170 171 174 174 176 176 177 178 179
	4.4 4.4 Sur Ap 6.1 6.1 6.1 6.1 var 6.1	4.4.1 4.4.2 Summar Appendi 5.1 Met 6.1.1 6.1.1.3 6.1.1.3 6.1.1.4 6.1.2 6.1.3 <i>variant l</i> 6.1.4	 4.4.1 Material testing

List of Figures

FIGURE 3-1: STRUCTURE OF THE BRONCHIAL TREE AND THE INDIVIDUAL ALVEOLAR SACS (ADAPTED FROM WIKIPEDIA)
FIGURE 3-2: STRUCTURE OF THE BRONCHIAL EPITHELIUM. A: HISTOLOGICAL SECTION OF THE HUMAN LUNG SHOWING THE BRONCHIAL
EPITHELIUM [HEMALAUM-EOSIN (HE) STAINED]; B: SCHEMATIC DRAWING OF THE BRONCHIAL EPITHELIUM;1: MUCUS-
SECRETING GOBLET CELL; 2: CILIATED CELL; 3: BASAL CELL; 4: BASEMENT MEMBRANE (COLLAGEN TYPE IV); 5: FIBROBLASTS; 6:
LAMINA PROPRIA MAINLY COMPOSED OF ECM PROTEINS
FIGURE 3-3: HISTOLOGICAL SECTION OF THE HUMAN LUNG SHOWING THE BRONCHIAL EPITHELIUM AND A SUBMUCOSAL GLAND,
INCLUDING A PRIMARY COLLECTING DUCT AND SECONDARY TUBULE (HE STAINING, MAGNIFICATION: 200X);1: BRONCHIAL
EPITHELIUM; 2: PRIMARY COLLECTING DUCT OF SUBMUCOSAL GLAND; 3: TUBULE OF SUBMUCOSAL GLAND
FIGURE 3-4: CULTIVATION OF AIRWAY EPITHELIAL CELLS ON MICROPOROUS MEMBRANES OF COMMERCIAL CELL CULTURE INSERTS. A:
THE CELLS ARE CULTIVATED SUBMERGED IN CELL CULTURE MEDIUM TO ENHANCE THE PROLIFERATION; B: THE AIR-LIQUID
INTERFACE CULTIVATION METHOD IS INDISPENSIBLE FOR THE DIFFERENTIATION INTO CILIATED AND MUCOUS CELLS (1= CILIATED
CELL; 2= GOBLET CELL; 3= BASAL CELL; 4= MUCUS)
FIGURE 3-5: OVERVIEW OF THE CULTEX [®] LTC (LONG-TERM CULTIVATION) MODULE (A) AND A DETAILED DESCRIPTION OF THE
INDIVIDUAL AUTOMATED VESSELS FOR THE CELL CULTURE INSERTS (B). (ILLUSTRATION PROVIDED BY HALTER ENGINEERING
GмвH) 40
Figure 4-1: Growth pattern of ChaGo-K1 cells cultivated on collagen IV coated microporous membranes (0.4 μ m
PORES) AFTER A TOTAL CULTIVATION TIME OF 14 days at the air-liquid interface with basic differentiation medium
(HE STAINED HISTOLOGICAL SECTION; MAGNIFICATION: 200x)
Figure 4-2: Morphology of ChaGo-K1 cells cultivated on collagen IV coated microporous membranes (0.4 μ m pores)
AT THE AIR-LIQUID INTERFACE WITH BASIC DIFFERENTIATION MEDIUM AFTER A CULTIVATION TIME OF 20 DAYS (HE STAINED
HISTOLOGICAL SECTION; MAGNIFICATION: 630x)
FIGURE 4-3: GROWTH PATTERN OF CALU-3 CELLS CULTIVATED IN CELL CULTURE FLASKS (75 CM ²) WITH EMEM INCLUDING 10 % FBS
FIGURE 4-4: MORPHOLOGY AND GROWTH PATTERN OF CALU-3 CELLS CULTIVATED ON COLLAGEN I COATED MICROPOROUS INSERT
Figure 4-4: Morphology and growth pattern of Calu-3 cells cultivated on collagen I coated microporous insert membranes (0.4 μ m) at the air-liquid interface with basic differentiation medium after a cultivation time of 31
FIGURE 4-4: MORPHOLOGY AND GROWTH PATTERN OF CALU-3 CELLS CULTIVATED ON COLLAGEN I COATED MICROPOROUS INSERT MEMBRANES (0.4 μM) AT THE AIR-LIQUID INTERFACE WITH BASIC DIFFERENTIATION MEDIUM AFTER A CULTIVATION TIME OF 31 DAYS (25 DAYS UNDER ALI CONDITIONS; HE STAINED HISTOLOGICAL SECTION; MAGNIFICATION: 400x)
FIGURE 4-4: MORPHOLOGY AND GROWTH PATTERN OF CALU-3 CELLS CULTIVATED ON COLLAGEN I COATED MICROPOROUS INSERT MEMBRANES (0.4 μM) AT THE AIR-LIQUID INTERFACE WITH BASIC DIFFERENTIATION MEDIUM AFTER A CULTIVATION TIME OF 31 DAYS (25 DAYS UNDER ALI CONDITIONS; HE STAINED HISTOLOGICAL SECTION; MAGNIFICATION: 400x)
FIGURE 4-4: MORPHOLOGY AND GROWTH PATTERN OF CALU-3 CELLS CULTIVATED ON COLLAGEN I COATED MICROPOROUS INSERT MEMBRANES (0.4 μM) AT THE AIR-LIQUID INTERFACE WITH BASIC DIFFERENTIATION MEDIUM AFTER A CULTIVATION TIME OF 31 DAYS (25 DAYS UNDER ALI CONDITIONS; HE STAINED HISTOLOGICAL SECTION; MAGNIFICATION: 400x)
FIGURE 4-4: MORPHOLOGY AND GROWTH PATTERN OF CALU-3 CELLS CULTIVATED ON COLLAGEN I COATED MICROPOROUS INSERT MEMBRANES (0.4 μM) AT THE AIR-LIQUID INTERFACE WITH BASIC DIFFERENTIATION MEDIUM AFTER A CULTIVATION TIME OF 31 DAYS (25 DAYS UNDER ALI CONDITIONS; HE STAINED HISTOLOGICAL SECTION; MAGNIFICATION: 400X)
 Figure 4-4: Morphology and growth pattern of Calu-3 cells cultivated on collagen I coated microporous insert Membranes (0.4 μm) at the air-liquid interface with basic differentiation medium after a cultivation time of 31 Days (25 days under ALI conditions; HE stained histological section; magnification: 400x)
 Figure 4-4: Morphology and growth pattern of Calu-3 cells cultivated on collagen I coated microporous insert Membranes (0.4 μm) at the air-liquid interface with basic differentiation medium after a cultivation time of 31 Days (25 Days under ALI conditions; HE stained histological section; magnification: 400x)
 FIGURE 4-4: MORPHOLOGY AND GROWTH PATTERN OF CALU-3 CELLS CULTIVATED ON COLLAGEN I COATED MICROPOROUS INSERT MEMBRANES (0.4 μM) AT THE AIR-LIQUID INTERFACE WITH BASIC DIFFERENTIATION MEDIUM AFTER A CULTIVATION TIME OF 31 DAYS (25 DAYS UNDER ALI CONDITIONS; HE STAINED HISTOLOGICAL SECTION; MAGNIFICATION: 400X)

FIGURE 4-7: IMMUNOHISTOCHEMICAL STAINING OF THE BASAL CELL MARKER P63 IN A CALU-3 CULTURE CULTIVATED ON COLLAGEN I
coated microporous insert membranes (0.4 μ m) at the air-liquid interface with basic differentiation medium
AFTER A CULTIVATION TIME OF 31 DAYS (25 DAYS UNDER ALI CONDITIONS). THE CELL NUCLEI WERE COUNTERSTAINED WITH
HEMATOXYLIN (MAGNIFICATION: 1000x)
FIGURE 4-8: IMMUNOHISTOCHEMICAL STAINING OF THE CILIA MARKER ACETYLATED A-TUBULIN IN A CALU-3 CULTURE CULTIVATED ON
collagen I coated microporous insert membranes (0.4 μ m) at the air-liquid interface with basic differentiation
MEDIUM AFTER A CULTIVATION TIME OF 50 DAYS (43 DAYS UNDER ALI CONDITIONS). THE CELL NUCLEI WERE COUNTERSTAINED
with hematoxylin (Magnification: 630x)
FIGURE 4-9: GROWTH PATTERN OF 16HBE140- CELLS CULTIVATED IN CELL CULTURE FLASKS (75 CM ²) WITH MEM INCLUDING 10%
FBS
FIGURE 4-10: MORPHOLOGY OF 16HBE140- CELLS CULTIVATED ON COLLAGEN/FIBRONECTIN COATED MICROPOROUS INSERT
MEMBRANES (0.4μ M) at the air-liquid interface with basic differentiation medium after a cultivation time of 21
DAYS (14 DAYS UNDER ALI CONDITIONS; HE-STAINED HISTOLOGICAL SECTION; MAGNIFICATION: 400x)
FIGURE 4-11: INCREASE OF SECRETORY MATERIAL OF 16HBE140- CELLS CULTIVATED ON COLLAGEN/FIBRONECTIN COATED
MICROPOROUS INSERT MEMBRANES (0.4μ M) at the air-liquid interface with basic differentiation medium after a
CULTIVATION TIME OF 31 DAYS (24 DAYS UNDER ALI CONDITIONS; HE-STAINED HISTOLOGICAL SECTION; MAGNIFICATION:
400x)
FIGURE 4-12: INCREASED STABILITY AND LIFE SPAN OF 16HBE140- CULTURES AFTER RINSING WITH PBS ONCE OR TWICE A WEEK. THE
cultures were cultivated on collagen/fibronectin coated microporous insert membranes (0.4μ m) at the air-
LIQUID INTERFACE WITH BASIC DIFFERENTIATION MEDIUM AFTER A CULTIVATION TIME OF 56 DAYS (50 DAYS UNDER ALI
conditions; HE stained histological section; magnification: 400x).
FIGURE 4-13: IMMUNOHISTOCHEMICAL STAINING OF THE MAJOR SECRETED AIRWAY MUCINS MUC5AC (A) AND MUC5B (B) IN A
$16HBE140$ - culture cultivated on collagen/fibronectin coated microporous insert membranes (0.4 μ m) at the
AIR-LIQUID INTERFACE WITH BASIC DIFFERENTIATION MEDIUM AFTER A CULTIVATION TIME OF 56 DAYS (51 DAYS UNDER ALI
conditions). The cell nuclei were counterstained with hematoxylin (Magnification: 400x)
FIGURE 4-14: IMMUNOHISTOCHEMICAL STAINING OF THE CILIA MARKER ACETYLATED A-TUBULIN IN A 16HBE14O- CULTURE
cultivated on collagen/fibronectin coated microporous insert membranes (0.4 μ m) at the air-liquid interface
with basic differentiation medium after a cultivation time of 56 days (51 days under ALI conditions). The cell
NUCLEI WERE COUNTERSTAINED WITH HEMATOXYLIN (MAGNIFICATION: 400x)
FIGURE 4-15: IMMUNOFLUORESCENCE ANALYSES OF THE PROLIFERATION MARKER KI-67 ON A HISTOLOGICAL SECTION OF A
$16HBE140$ - culture cultivated on collagen/fibronectin coated microporous insert membranes (0.4 μ m) at the
AIR-LIQUID INTERFACE WITH BASIC DIFFERENTIATION MEDIUM AFTER A CULTIVATION TIME OF 37 DAYS (32 DAYS UNDER ALI
conditions). The KI-67 staining was visualized through a TRITC-labelled secondary antibody (red; A). The cell
NUCLEI WERE STAINED WITH DAPI (BLUE; B). SIGNALS OF BOTH STAININGS WERE MERGED TO OBTAIN AN OVERVIEW OF THE
RATIO OF PROLIFERATING, KI-67 POSITIVE CELLS (PURPLE; C) TO THE TOTAL NUMBER OF CELLS.

FIGURE 4-16: ANALYSIS OF TIGHT JUNCTION FORMATION IN SIX 16HBE140- INSERT CULTURES BY MONITORING THE TEER VALUES OVER A CULTURE PERIOD OF 11 (A) AND 18 DAYS (B). THE CELLS WERE CULTIVATED UNDER SUBMERGED CONDITIONS WITH CELL-SPECIFIC GROWTH MEDIUM (MEM + 10 % FBS) UNTIL THEY FORMED A CONFLUENT LAYER AFTER 8 DAYS. STARTING FROM DAY NINE, THE CULTURES WERE CULTIVATED AT THE AIR-LIQUID INTERFACE BUT STILL SUPPLIED WITH THE GROWTH MEDIUM UNTIL DAY 11; A: AT DAY 11, THREE 16HBE14O- INSERT CULTURES WERE TREATED WITH TRITON X-100 TO INDUCE APOPTOSIS. THE TEER VALUES WERE DETERMINED 2 H AFTER THE ADDITION OF TRITON X-100 AND THESE VALUES WERE CONSIDERED AS BLANK VALUES. B: THE OTHER THREE INSERT CULTURES WERE ADAPTED TO THE BASIC DIFFERENTIATION MEDIUM (BEBM/ DMEM (1:1) INCLUDING SINGLEQUOTS® (W/O RETINOIC ACID), 1 MM CACL₂, 2 % ULTROSER® G AND FIGURE 4-17: GROWTH PATTERN OF IMMORTALIZED NHBE CELLS OF CLONE 1 (7F4167), PASSAGE 7, CULTIVATED IN CELL CULTURE FIGURE 4-18: INVERSE MICROSCOPIC IMAGES (A+C) AND HE STAINED HISTOLOGICAL SECTIONS (B+D) OF CULTURES OF TWO CLONAL CELL LINES. THE CELL MORPHOLOGY OF THE TWO CLONAL CELL LINES DIFFERS WHILE THE STRATIFIED GROWTH IS SIMILAR. A+B: MONOCULTURES OF CLONE 1 (P07) AFTER A CULTIVATION TIME OF 11 (A) OR 8 DAYS, RESPECTIVELY (B). C+D: FIGURE 4-19: IMMUNOHISTOCHEMICAL STAINING OF THE BASAL CELL MARKER P63 IN A CULTURE OF THE CLONAL CELL LINE 7 (99498) CULTIVATED ON COLLAGEN IV COATED MICROPOROUS INSERT MEMBRANES (0.4 μM) UNDER ALI CONDITIONS WITH BASIC DIFFERENTIATION MEDIUM AFTER A CULTIVATION TIME OF 29 DAYS (23 DAYS UNDER ALI CONDITIONS). THE CELL NUCLEI WERE FIGURE 4-20: TEST OF CELL VIABILITY OF HBEPC908 IN DIFFERENT PASSAGES. THE CELLS WERE CRYOPRESERVED IN PASSAGE 5 AND, AFTER THAWING, SUBCULTIVATED UNTIL PASSAGE 10. The cells were cultivated in 25 cm 2 cell culture flasks at a seeding density of $3.5 \times 10^5/25$ cm^{2.} Subcultivation was performed every 7 days and the number of viable cells FIGURE 4-21: ALTERED MORPHOLOGY OF HBEPC908 CELLS OF PASSAGE 10, CULTIVATED IN A 25CM² CULTURE FLASK AND SUPPLIED WITH AIRWAY EPITHELIAL CELL GROWTH MEDIUM (PROMOCELL). THE CELLS WERE FROZEN IN LIQUID NITROGEN IN PASSAGE FIGURE 4-22: GROWTH PATTERN AND CELL MORPHOLOGY OF HBEPC812 (A) AND HBEPC901 (B) CULTURES CULTIVATED ON Collagen IV coated microporous membranes (0.4 μ m pores) at the air-liquid interface with basic DIFFERENTIATION MEDIUM WITHOUT ULTROSER® G AFTER A CULTIVATION TIME OF 13 (A) OR 28 (B) DAYS. (HE STAINED HISTOLOGICAL SECTIONS; MAGNIFICATION: 400x). FIGURE 4-23: IMMUNOHISTOCHEMICAL STAINING OF THE SQUAMOUS CELL MARKER CYTOKERATIN 6 (CK6) IN A HBEPC812 CULTURE cultivated on collagen IV coated microporous insert membranes (0.4 μ m) at the air-liquid interface with basic DIFFERENTIATION MEDIUM WITHOUT ULTROSER® G AFTER A CULTIVATION TIME OF 13 DAYS (9 DAYS UNDER ALL CONDITIONS). FIGURE 4-24: PSEUDOSTRATIFIED STRUCTURE OF A HBEPC901 CULTURE CULTIVATED ON COLLAGEN IV COATED MICROPOROUS MEMBRANES (0.4 μM PORES) AT THE AIR-LIQUID INTERFACE WITH BASIC DIFFERENTIATION MEDIUM INCLUDING 2 % ULTROSER® G AFTER A CULTIVATION TIME OF 54 DAYS. (49 DAYS UNDER ALI CONDITIONS; HE STAINED HISTOLOGICAL

FIGURE 4-25: IMMUNOHISTOCHEMICAL STAINING OF THE BASAL CELL MARKER P63 IN A HBEPC908 CULTURE CULTIVATED ON
collagen IV coated microporous insert membranes (0.4 μ m) at the air-liquid interface with basic
differentiation medium including 2 % Ultroser $^{\circ}$ G after a cultivation time of 54 days (49 days under ALI
CONDITIONS). THE CELL NUCLEI WERE COUNTERSTAINED WITH HEMATOXYLIN (MAGNIFICATION: 400x)
FIGURE 4-26: IMMUNOHISTOCHEMICAL STAINING OF ACETYLATED A-TUBULIN IN A HBEPC908 CULTURE CULTIVATED ON COLLAGEN
IV coated microporous insert membranes (0.4 μ m) at the air-liquid interface with basic differentiation medium
including 2 % Ultroser® G after a cultivation time of 54 days (49 days under ALI conditions). The cell nuclei
WERE COUNTERSTAINED WITH HEMATOXYLIN (MAGNIFICATION: 400x)
FIGURE 4-27: IMMUNOHISTOCHEMICAL STAINING OF THE MUCIN 5AC IN A HBEPC908 CULTURE CULTIVATED ON COLLAGEN IV
coated microporous insert membranes (0.4 $_{\mu}$ m) at the air-liquid interface with basic differentiation medium
INCLUDING 2 % ULTROSER® G AFTER A CULTIVATION TIME OF 54 DAYS (49 DAYS UNDER ALI CONDITIONS). THE CELL NUCLEI
WERE COUNTERSTAINED WITH HEMATOXYLIN (MAGNIFICATION: 630x)72
FIGURE 4-28: NORMAL HUMAN BRONCHIAL EPITHELIAL CELLS, RESUSPENDED IN BEGM, APPROX. 15 MIN AFTER ISOLATION FROM
FRESH LUNG TISSUE. THE CELLS UNDERWENT THREE MORPHOLOGICAL STAGES BEFORE THEY ADHERED TO THE CULTURE VESSEL.
Cells with strong beating cilia (1) were only detectable within the first 30 min. Afterwards the beating slowed
down and the cells became more roundish (2). In the last stage, the cilia were lost and the cells had a round
SHAPE (3)
FIGURE 4-29: NORMAL HUMAN BRONCHIAL EPITHELIAL CELL CULTURE IN BEGM, APPROX. 2 H AFTER ISOLATION FROM FRESH LUNG
TISSUE. THE CULTURE IS HIGHLY CONTAMINATED WITH ERYTHROCYTES (ROUND BICONCAVE DISC-SHAPED CELLS)
ACCUMULATING AT THE BOTTOM OF THE CULTURE VESSEL
FIGURE 4-30: NORMAL HUMAN BRONCHIAL EPITHELIAL CELLS ADHERED TO COLLAGEN/FIBRONECTIN COATED CULTURE VESSELS 16
DAYS AFTER ISOLATION. CELLS THAT UNDERWENT AN ADDITIONAL DISSOCIATION STEP WITH "RESCUE MEDIUM" (B) FORMED A
CONFLUENT LAYER EARLIER THAN THE "NON-RESCUED" CELLS (A) IN SOME CASES
FIGURE 4-31: AN ADDITIONAL CENTRIFUGATION OF THE CELL SUSPENSION OF NORMAL HUMAN BRONCHIAL EPITHELIAL CELLS WITHIN
APPROX. 5 DAYS AFTER ISOLATION OFTEN RESULTED IN THE ADHERENCE OF LARGE CELL AGGREGATES. THE CELLS GROWING OUT
FROM THESE ADHERED CELL AGGREGATES OFTEN HAD AN ALTERED MORPHOLOGY
FIGURE 4-32: NHBE CELLS FORMING A CONFLUENT CELL LAYER IN WELLS OF A 6-WELL PLATES APPROXIMATELY 17 DAYS AFTER
ISOLATION FROM TISSUE
FIGURE 4-33: NORMAL HUMAN BRONCHIAL EPITHELIAL CELLS OF PASSAGE 1 CULTIVATED IN A COLLAGEN/FIBRONECTIN COATED
CULTURE FLASKS (25 CM ²) AND SUPPLIED WITH BEGM
FIGURE 4-34: DIFFERENTIATED CULTURE OF NHBE CELLS OF PASSAGE 2 CULTIVATED ON COLLAGEN IV COATED MICROPOROUS
MEMBRANES (0.4 μ M PORES) AT THE AIR-LIQUID INTERFACE WITH BASIC DIFFERENTIATION MEDIUM AFTER A CULTIVATION TIME
of 42 days (38 days under ALI conditions). The cell population is characterized by a pseudostratified structure
WITH BASAL CELLS, CILIATED CELLS AND NON-CILIATED COLUMNAR CELLS (POTENTIALLY SECRETORY CELLS; HE STAINED
HISTOLOGICAL SECTIONS; MAGNIFICATION: 1000x)

FIGURE 4-35: IMMUNOHISTOCHEMICAL STAINING OF THE MUCINS 5AC (A) AND 5B IN A NHBE CULTURE (PASSAGE 2) CULTIVATED)
on collagen IV coated microporous insert membranes ($0.4~\mu m$) at the air-liquid interface with basic	
DIFFERENTIATION MEDIUM AFTER A CULTIVATION TIME OF 42 DAYS (38 DAYS UNDER ALI CONDITIONS). THE CELL NUCLEI WE	RE
COUNTERSTAINED WITH HEMATOXYLIN (MAGNIFICATION: 400x).	85
FIGURE 4-36: IMMUNOHISTOCHEMICAL STAINING OF THE BASAL CELL MARKER P63 IN A NHBE CULTURE (PASSAGE 2) CULTIVATED O	лс
COLLAGEN IV COATED MICROPOROUS INSERT MEMBRANES (0.4μ M) at the air-liquid interface with basic	
differentiation medium after a cultivation time of 42 days (38 days under ALI conditions). The cell nuclei we	RE
COUNTERSTAINED WITH HEMATOXYLIN (MAGNIFICATION: 400x).	86
FIGURE 4-37: TWO CULTURES OF THE SAME NHBE LOT BEFORE (A) AND AFTER CRYOPRESERVATION (B). BOTH CULTURES WERE	
CULTIVATED UNDER IDENTICAL CONDITIONS (ON COLLAGEN IV COATED INSERT MEMBRANES, AT THE AIR-LIQUID INTERFACE AI	١D
WITH BASIC DIFFERENTIATION MEDIUM) FOR 26 DAYS (21 DAYS UNDER ALI CONDITIONS; HE STAINED HISTOLOGICAL SECTION	NS;
magnification: 400x)	87
FIGURE 4-38: FIBROBLASTS GROWING OUT FROM HUMAN BRONCHIAL TISSUE CULTIVATED IN PETRI DISHES (3 CM DIAMETER) FOR 2	0
DAYS	89
FIGURE 4-39: TWO DIFFERENT CELL TYPES GROWING OUT FROM HUMAN BRONCHIAL TISSUE CULTIVATED IN PETRI DISHES (3 CM	
DIAMETER) FOR 14 DAYS. AFTER REMOVAL OF THE TISSUE SAMPLE, A LARGE NUMBER OF PRESUMED MACROPHAGES WERE	
DETECTED, WHILE ONLY A FEW FIBROBLASTS GREW OUT FROM THE TISSUE AT THE OUTER EDGES.	90
Figure 4-40: Normal human lung fibroblasts of passage 2 after isolation from lung tissue cultivated in 75 cm ²	
culture flasks and supplied with RPMI incl. $10\%FBS,L$ -glutamine, sodium pyruvates and	
PENICILLIN/STREPTOMYCIN	92
PENICILLIN/STREPTOMYCIN Figure 4-41: Basic protocol for the Sandwich Cell Model (SCM). A: Lung fibroblasts are suspended in a pH neutral	92
PENICILLIN/STREPTOMYCIN FIGURE 4-41: BASIC PROTOCOL FOR THE SANDWICH CELL MODEL (SCM). A: LUNG FIBROBLASTS ARE SUSPENDED IN A PH NEUTRAL COLLAGEN SOLUTION AND APPLIED ONTO THE MEMBRANE OF A CELL CULTURE INSERT. B: AFTER THE COMPLETE	92
PENICILLIN/STREPTOMYCIN FIGURE 4-41: BASIC PROTOCOL FOR THE SANDWICH CELL MODEL (SCM). A: LUNG FIBROBLASTS ARE SUSPENDED IN A PH NEUTRAL COLLAGEN SOLUTION AND APPLIED ONTO THE MEMBRANE OF A CELL CULTURE INSERT. B: AFTER THE COMPLETE POLYMERIZATION OF THE COLLAGEN AT 37°C, LUNG EPITHELIAL CELLS ARE APPLIED ON THE COLLAGEN MATRIX AND THE CO-	92
PENICILLIN/STREPTOMYCIN FIGURE 4-41: BASIC PROTOCOL FOR THE SANDWICH CELL MODEL (SCM). A: LUNG FIBROBLASTS ARE SUSPENDED IN A PH NEUTRAL COLLAGEN SOLUTION AND APPLIED ONTO THE MEMBRANE OF A CELL CULTURE INSERT. B: AFTER THE COMPLETE POLYMERIZATION OF THE COLLAGEN AT 37°C, LUNG EPITHELIAL CELLS ARE APPLIED ON THE COLLAGEN MATRIX AND THE CO- CULTURES ARE CULTIVATED UNDER SUBMERGED CONDITIONS. C: AS SOON AS THE EPITHELIAL CELL LAYER REACHED 100%	92
PENICILLIN/STREPTOMYCIN FIGURE 4-41: BASIC PROTOCOL FOR THE SANDWICH CELL MODEL (SCM). A: LUNG FIBROBLASTS ARE SUSPENDED IN A PH NEUTRAL COLLAGEN SOLUTION AND APPLIED ONTO THE MEMBRANE OF A CELL CULTURE INSERT. B: AFTER THE COMPLETE POLYMERIZATION OF THE COLLAGEN AT 37°C, LUNG EPITHELIAL CELLS ARE APPLIED ON THE COLLAGEN MATRIX AND THE CO- CULTURES ARE CULTIVATED UNDER SUBMERGED CONDITIONS. C: AS SOON AS THE EPITHELIAL CELL LAYER REACHED 100% CONFLUENCY, THE CULTURES ARE CULTIVATED AT THE AIR-LIQUID INTERFACE TO PROMOTE THE CELL DIFFERENTIATION	92
PENICILLIN/STREPTOMYCIN FIGURE 4-41: BASIC PROTOCOL FOR THE SANDWICH CELL MODEL (SCM). A: LUNG FIBROBLASTS ARE SUSPENDED IN A PH NEUTRAL COLLAGEN SOLUTION AND APPLIED ONTO THE MEMBRANE OF A CELL CULTURE INSERT. B: AFTER THE COMPLETE POLYMERIZATION OF THE COLLAGEN AT 37°C, LUNG EPITHELIAL CELLS ARE APPLIED ON THE COLLAGEN MATRIX AND THE CO- CULTURES ARE CULTIVATED UNDER SUBMERGED CONDITIONS. C: AS SOON AS THE EPITHELIAL CELL LAYER REACHED 100% CONFLUENCY, THE CULTURES ARE CULTIVATED AT THE AIR-LIQUID INTERFACE TO PROMOTE THE CELL DIFFERENTIATION (1= COLLAGEN MATRIX; 2= FIBROBLASTS; UNDIFFERENTIATED EPITHELIAL CELLS; 4= CILIATED CELL; 5= BASAL CELL; 6=	92
PENICILLIN/STREPTOMYCIN FIGURE 4-41: BASIC PROTOCOL FOR THE SANDWICH CELL MODEL (SCM). A: LUNG FIBROBLASTS ARE SUSPENDED IN A PH NEUTRAL COLLAGEN SOLUTION AND APPLIED ONTO THE MEMBRANE OF A CELL CULTURE INSERT. B: AFTER THE COMPLETE POLYMERIZATION OF THE COLLAGEN AT 37°C, LUNG EPITHELIAL CELLS ARE APPLIED ON THE COLLAGEN MATRIX AND THE CO- CULTURES ARE CULTIVATED UNDER SUBMERGED CONDITIONS. C: AS SOON AS THE EPITHELIAL CELL LAYER REACHED 100% CONFLUENCY, THE CULTURES ARE CULTIVATED AT THE AIR-LIQUID INTERFACE TO PROMOTE THE CELL DIFFERENTIATION (1= COLLAGEN MATRIX; 2= FIBROBLASTS; UNDIFFERENTIATED EPITHELIAL CELLS; 4= CILIATED CELL; 5= BASAL CELL; 6= GOBLET CELL; 7= MUCUS).	92 93
PENICILLIN/STREPTOMYCIN FIGURE 4-41: BASIC PROTOCOL FOR THE SANDWICH CELL MODEL (SCM). A: LUNG FIBROBLASTS ARE SUSPENDED IN A PH NEUTRAL COLLAGEN SOLUTION AND APPLIED ONTO THE MEMBRANE OF A CELL CULTURE INSERT. B: AFTER THE COMPLETE POLYMERIZATION OF THE COLLAGEN AT 37°C, LUNG EPITHELIAL CELLS ARE APPLIED ON THE COLLAGEN MATRIX AND THE CO- CULTURES ARE CULTIVATED UNDER SUBMERGED CONDITIONS. C: AS SOON AS THE EPITHELIAL CELL LAYER REACHED 100% CONFLUENCY, THE CULTURES ARE CULTIVATED AT THE AIR-LIQUID INTERFACE TO PROMOTE THE CELL DIFFERENTIATION (1= COLLAGEN MATRIX; 2= FIBROBLASTS; UNDIFFERENTIATED EPITHELIAL CELLS; 4= CILIATED CELL; 5= BASAL CELL; 6= GOBLET CELL; 7= MUCUS). FIGURE 4-42: 12-WELL RECEIVER PLATE WITH TEN CELL CULTURE INSERTS CARRYING SCM CULTURES PREPARED FROM DIFFERENT	92 93
PENICILLIN/STREPTOMYCIN FIGURE 4-41: BASIC PROTOCOL FOR THE SANDWICH CELL MODEL (SCM). A: LUNG FIBROBLASTS ARE SUSPENDED IN A PH NEUTRAL COLLAGEN SOLUTION AND APPLIED ONTO THE MEMBRANE OF A CELL CULTURE INSERT. B: AFTER THE COMPLETE POLYMERIZATION OF THE COLLAGEN AT 37°C, LUNG EPITHELIAL CELLS ARE APPLIED ON THE COLLAGEN MATRIX AND THE CO- CULTURES ARE CULTIVATED UNDER SUBMERGED CONDITIONS. C: AS SOON AS THE EPITHELIAL CELL LAYER REACHED 100% CONFLUENCY, THE CULTURES ARE CULTIVATED AT THE AIR-LIQUID INTERFACE TO PROMOTE THE CELL DIFFERENTIATION (1= COLLAGEN MATRIX; 2= FIBROBLASTS; UNDIFFERENTIATED EPITHELIAL CELLS; 4= CILIATED CELL; 5= BASAL CELL; 6= GOBLET CELL; 7= MUCUS). FIGURE 4-42: 12-WELL RECEIVER PLATE WITH TEN CELL CULTURE INSERTS CARRYING SCM CULTURES PREPARED FROM DIFFERENT EPITHELIAL CELLS (HBEPC, 16HBE140-) AND TWO HBEPC MONOCULTURES (MARKED). ALL SCM CULTURES CLEARLY SHOW	92 - 93 v
PENICILLIN/STREPTOMYCIN FIGURE 4-41: BASIC PROTOCOL FOR THE SANDWICH CELL MODEL (SCM). A: LUNG FIBROBLASTS ARE SUSPENDED IN A PH NEUTRAL COLLAGEN SOLUTION AND APPLIED ONTO THE MEMBRANE OF A CELL CULTURE INSERT. B: AFTER THE COMPLETE POLYMERIZATION OF THE COLLAGEN AT 37°C, LUNG EPITHELIAL CELLS ARE APPLIED ON THE COLLAGEN MATRIX AND THE CO- CULTURES ARE CULTIVATED UNDER SUBMERGED CONDITIONS. C: AS SOON AS THE EPITHELIAL CELL LAYER REACHED 100% CONFLUENCY, THE CULTURES ARE CULTIVATED AT THE AIR-LIQUID INTERFACE TO PROMOTE THE CELL DIFFERENTIATION (1= COLLAGEN MATRIX; 2= FIBROBLASTS; UNDIFFERENTIATED EPITHELIAL CELLS; 4= CILIATED CELL; 5= BASAL CELL; 6= GOBLET CELL; 7= MUCUS). FIGURE 4-42: 12-WELL RECEIVER PLATE WITH TEN CELL CULTURE INSERTS CARRYING SCM CULTURES PREPARED FROM DIFFERENT EPITHELIAL CELLS (HBEPC, 16HBE140-) AND TWO HBEPC MONOCULTURES (MARKED). ALL SCM CULTURES CLEARLY SHOW CONTRACTION OF THE COLLAGEN MATRIX.	92 - 93 v 94
PENICILLIN/STREPTOMYCIN	92 - 93 v 94
 PENICILLIN/STREPTOMYCIN FIGURE 4-41: BASIC PROTOCOL FOR THE SANDWICH CELL MODEL (SCM). A: LUNG FIBROBLASTS ARE SUSPENDED IN A PH NEUTRAL COLLAGEN SOLUTION AND APPLIED ONTO THE MEMBRANE OF A CELL CULTURE INSERT. B: AFTER THE COMPLETE POLYMERIZATION OF THE COLLAGEN AT 37°C, LUNG EPITHELIAL CELLS ARE APPLIED ON THE COLLAGEN MATRIX AND THE CO-CULTURES ARE CULTIVATED UNDER SUBMERGED CONDITIONS. C: AS SOON AS THE EPITHELIAL CELL LAYER REACHED 100% CONFLUENCY, THE CULTURES ARE CULTIVATED AT THE AIR-LIQUID INTERFACE TO PROMOTE THE CELL DIFFERENTIATION (1= COLLAGEN MATRIX; 2= FIBROBLASTS; UNDIFFERENTIATED EPITHELIAL CELLS; 4= CILIATED CELL; 5= BASAL CELL; 6= GOBLET CELL; 7= MUCUS). FIGURE 4-42: 12-WELL RECEIVER PLATE WITH TEN CELL CULTURE INSERTS CARRYING SCM CULTURES PREPARED FROM DIFFERENT EPITHELIAL CELLS (HBEPC, 16HBE140-) AND TWO HBEPC MONOCULTURES (MARKED). ALL SCM CULTURES CLEARLY SHOW CONTRACTION OF THE COLLAGEN MATRIX. FIGURE 4-43: SCHEMATIC REPRESENTATION OF A SCM CULTURE COMPOSED OF A BASAL COLLAGEN LAYER (A: 100 μL COLLAGEN SOLUTION); AND A MAIN COLLAGEN LAYER IN WHICH THE FIBROBLAST ARE EMBEDDED (B: 170 – 200 μL COLLAGEN SOLUTION) 	92 - 93 v 94
 PENICILLIN/STREPTOMYCIN FIGURE 4-41: BASIC PROTOCOL FOR THE SANDWICH CELL MODEL (SCM). A: LUNG FIBROBLASTS ARE SUSPENDED IN A PH NEUTRAL COLLAGEN SOLUTION AND APPLIED ONTO THE MEMBRANE OF A CELL CULTURE INSERT. B: AFTER THE COMPLETE POLYMERIZATION OF THE COLLAGEN AT 37°C, LUNG EPITHELIAL CELLS ARE APPLIED ON THE COLLAGEN MATRIX AND THE CO-CULTURES ARE CULTIVATED UNDER SUBMERGED CONDITIONS. C: AS SOON AS THE EPITHELIAL CELL LAYER REACHED 100% CONFLUENCY, THE CULTURES ARE CULTIVATED AT THE AIR-LIQUID INTERFACE TO PROMOTE THE CELL DIFFERENTIATION (1= COLLAGEN MATRIX; 2= FIBROBLASTS; UNDIFFERENTIATED EPITHELIAL CELLS; 4= CILIATED CELL; 5= BASAL CELL; 6= GOBLET CELL; 7= MUCUS). FIGURE 4-42: 12-WELL RECEIVER PLATE WITH TEN CELL CULTURE INSERTS CARRYING SCM CULTURES PREPARED FROM DIFFERENT EPITHELIAL CELLS (HBEPC, 16HBE140-) AND TWO HBEPC MONOCULTURES (MARKED). ALL SCM CULTURES CLEARLY SHOW CONTRACTION OF THE COLLAGEN MATRIX. FIGURE 4-43: SCHEMATIC REPRESENTATION OF A SCM CULTURE COMPOSED OF A BASAL COLLAGEN LAYER (A: 100 µL COLLAGEN SOLUTION); AND A MAIN COLLAGEN LAYER IN WHICH THE FIBROBLAST ARE EMBEDDED (B: 170 – 200 µL COLLAGEN SOLUTION 1= CILIATED CELL; 2= BASAL CELL; 3= GOBLET CELL; 4= MUCUS). 	92 - 93 v 94 4; 97
 PENICILLIN/STREPTOMYCIN FIGURE 4-41: BASIC PROTOCOL FOR THE SANDWICH CELL MODEL (SCM). A: LUNG FIBROBLASTS ARE SUSPENDED IN A PH NEUTRAL COLLAGEN SOLUTION AND APPLIED ONTO THE MEMBRANE OF A CELL CULTURE INSERT. B: AFTER THE COMPLETE POLYMERIZATION OF THE COLLAGEN AT 37°C, LUNG EPITHELIAL CELLS ARE APPLIED ON THE COLLAGEN MATRIX AND THE CO-CULTURES ARE CULTIVATED UNDER SUBMERGED CONDITIONS. C: AS SOON AS THE EPITHELIAL CELL LAYER REACHED 100% CONFLUENCY, THE CULTURES ARE CULTIVATED AT THE AIR-LIQUID INTERFACE TO PROMOTE THE CELL DIFFERENTIATION (1= COLLAGEN MATRIX; 2= FIBROBLASTS; UNDIFFERENTIATED EPITHELIAL CELLS; 4= CILIATED CELL; 5= BASAL CELL; 6= GOBLET CELL; 7= MUCUS). FIGURE 4-42: 12-WELL RECEIVER PLATE WITH TEN CELL CULTURE INSERTS CARRYING SCM CULTURES PREPARED FROM DIFFERENT EPITHELIAL CELLS (HBEPC, 16HBE140-) AND TWO HBEPC MONOCULTURES (MARKED). ALL SCM CULTURES CLEARLY SHOW CONTRACTION OF THE COLLAGEN MATRIX. FIGURE 4-43: SCHEMATIC REPRESENTATION OF A SCM CULTURE COMPOSED OF A BASAL COLLAGEN LAYER (A: 100 µL COLLAGEN SOLUTION); AND A MAIN COLLAGEN LAYER IN WHICH THE FIBROBLAST ARE EMBEDDED (B: 170 – 200 µL COLLAGEN SOLUTION 1= CILIATED CELL; 2= BASAL CELL; 3= GOBLET CELL; 4= MUCUS). FIGURE 4-44: SCM CULTURES COMPOSED OF CALU-3 AND IMR90 CELLS (EMBEDDED IN COLLAGEN I) AFTER A CULTIVATION TIME COLLAGEN IN COLLAGEN LAYER IN WHICH THE FIBROBLAST ARE EMBEDDED (B: 170 – 200 µL CULLAGEN SOLUTION) 	92 - 93 v 94 v; 97 »F
 PENICILLIN/STREPTOMYCIN FIGURE 4-41: BASIC PROTOCOL FOR THE SANDWICH CELL MODEL (SCM). A: LUNG FIBROBLASTS ARE SUSPENDED IN A PH NEUTRAL COLLAGEN SOLUTION AND APPLIED ONTO THE MEMBRANE OF A CELL CULTURE INSERT. B: AFTER THE COMPLETE POLYMERIZATION OF THE COLLAGEN AT 37°C, LUNG EPITHELIAL CELLS ARE APPLIED ON THE COLLAGEN MATRIX AND THE CO-CULTURES ARE CULTIVATED UNDER SUBMERGED CONDITIONS. C: AS SOON AS THE EPITHELIAL CELL LAYER REACHED 100% CONFLUENCY, THE CULTURES ARE CULTIVATED AT THE AIR-LIQUID INTERFACE TO PROMOTE THE CELL DIFFERENTIATION (1= COLLAGEN MATRIX; 2= FIBROBLASTS; UNDIFFERENTIATED EPITHELIAL CELLS; 4= CILIATED CELL; 5= BASAL CELL; 6= GOBLET CELL; 7= MUCUS). FIGURE 4-42: 12-WELL RECEIVER PLATE WITH TEN CELL CULTURE INSERTS CARRYING SCM CULTURES PREPARED FROM DIFFERENT EPITHELIAL CELLS (HBEPC, 16HBE140-) AND TWO HBEPC MONOCULTURES (MARKED). ALL SCM CULTURES CLEARLY SHOV CONTRACTION OF THE COLLAGEN MATRIX. FIGURE 4-43: SCHEMATIC REPRESENTATION OF A SCM CULTURE COMPOSED OF A BASAL COLLAGEN LAYER (A: 100 µL COLLAGEN SOLUTION); AND A MAIN COLLAGEN LAYER IN WHICH THE FIBROBLAST ARE EMBEDDED (B: 170 – 200 µL COLLAGEN SOLUTION); AND A MAIN COLLAGEN LAYER IN WHICH THE FIBROBLAST ARE EMBEDDED (B: 170 – 200 µL COLLAGEN SOLUTION) FIGURE 4-44: SCM CULTURES COMPOSED OF CALU-3 AND IMR90 CELLS (EMBEDDED IN COLLAGEN I) AFTER A CULTIVATION TIME COLLAGEN LAYER (A: 100 µL COLLAGEN SOLUTION); AND A MAIN COLLAGEN LAYER IN WHICH THE FIBROBLAST ARE EMBEDDED (B: 170 – 200 µL COLLAGEN SOLUTION) 	92 - 93 v 94 v; 97 »F E.
 PENICILLIN/STREPTOMYCIN FIGURE 4-41: BASIC PROTOCOL FOR THE SANDWICH CELL MODEL (SCM). A: LUNG FIBROBLASTS ARE SUSPENDED IN A PH NEUTRAL COLLAGEN SOLUTION AND APPLIED ONTO THE MEMBRANE OF A CELL CULTURE INSERT. B: AFTER THE COMPLETE POLYMERIZATION OF THE COLLAGEN AT 37°C, LUNG EPITHELIAL CELLS ARE APPLIED ON THE COLLAGEN MATRIX AND THE CO-CULTURES ARE CULTIVATED UNDER SUBMERGED CONDITIONS. C: AS SOON AS THE EPITHELIAL CELL LAYER REACHED 100% CONFLUENCY, THE CULTURES ARE CULTIVATED AT THE AIR-LIQUID INTERFACE TO PROMOTE THE CELL DIFFERENTIATION (1= COLLAGEN MATRIX; 2= FIBROBLASTS; UNDIFFERENTIATED EPITHELIAL CELLS; 4= CILIATED CELL; 5= BASAL CELL; 6= GOBLET CELL; 7= MUCUS). FIGURE 4-42: 12-WELL RECEIVER PLATE WITH TEN CELL CULTURE INSERTS CARRYING SCM CULTURES PREPARED FROM DIFFERENT EPITHELIAL CELLS (HBEPC, 16HBE140-) AND TWO HBEPC MONOCULTURES (MARKED). ALL SCM CULTURES CLEARLY SHOW CONTRACTION OF THE COLLAGEN MATRIX. FIGURE 4-43: SCHEMATIC REPRESENTATION OF A SCM CULTURE COMPOSED OF A BASAL COLLAGEN LAYER (A: 100 µL COLLAGEN SOLUTION); AND A MAIN COLLAGEN LAYER IN WHICH THE FIBROBLAST ARE EMBEDDED (B: 170 – 200 µL COLLAGEN SOLUTION 1= CILIATED CELL; 2= BASAL CELL; 3= GOBLET CELL; 4= MUCUS). FIGURE 4-44: SCM CULTURES COMPOSED OF CALU-3 AND IMR90 CELLS (EMBEDDED IN COLLAGEN I) AFTER A CULTIVATION TIME COLLAGEN LAYER (A: 100 µL COLLAGEN LAYER IN WHICH THE FIBROBLAST ARE EMBEDDED (B: 170 – 200 µL COLLAGEN SOLUTION 1= CILIATED CELL; 2= BASAL CELL; 3= GOBLET CELL; 4= MUCUS). FIGURE 4-44: SCM CULTURES COMPOSED OF CALU-3 AND IMR90 CELLS (EMBEDDED IN COLLAGEN I) AFTER A CULTIVATION TIME COLLAGEN LAYER (A: 100 µL COLLAGEN LAYER IN WHICH THE FIBROBLAST ARE EMBEDDED (B: 170 – 200 µL COLLAGEN SOLUTION TIME COLLAGEN LAYER IN WHICH THE FIBROBLAST ARE EMBEDDED (B: 170 – 200 µL COLLAGEN SOLUTION 1= CILIATED CELL; 2= BASAL CELL; 3= GOBLET CELL; 4= MUCUS). 	92 - 93 v 94 v; 97 of E. AL
 PENICILLIN/STREPTOMYCIN FIGURE 4-41: BASIC PROTOCOL FOR THE SANDWICH CELL MODEL (SCM). A: LUNG FIBROBLASTS ARE SUSPENDED IN A PH NEUTRAL COLLAGEN SOLUTION AND APPLIED ONTO THE MEMBRANE OF A CELL CULTURE INSERT. B: AFTER THE COMPLETE POLYMERIZATION OF THE COLLAGEN AT 37°C, LUNG EPITHELIAL CELLS ARE APPLIED ON THE COLLAGEN MATRIX AND THE CO-CULTURES ARE CULTIVATED UNDER SUBMERGED CONDITIONS. C: AS SOON AS THE EPITHELIAL CELL LAYER REACHED 100% CONFLUENCY, THE CULTURES ARE CULTIVATED AT THE AIR-LIQUID INTERFACE TO PROMOTE THE CELL DIFFERENTIATION (1= COLLAGEN MATRIX; 2= FIBROBLASTS; UNDIFFERENTIATED EPITHELIAL CELLS; 4= CILIATED CELL; 5= BASAL CELL; 6= GOBLET CELL; 7= MUCUS). FIGURE 4-42: 12-WELL RECEIVER PLATE WITH TEN CELL CULTURE INSERTS CARRYING SCM CULTURES PREPARED FROM DIFFERENT EPITHELIAL CELLS (HBEPC, 16HBE140-) AND TWO HBEPC MONOCULTURES (MARKED). ALL SCM CULTURES CLEARLY SHOV CONTRACTION OF THE COLLAGEN MATRIX. FIGURE 4-43: SCHEMATIC REPRESENTATION OF A SCM CULTURE COMPOSED OF A BASAL COLLAGEN LAYER (A: 100 µL COLLAGEN SOLUTION); AND A MAIN COLLAGEN LAYER IN WHICH THE FIBROBLAST ARE EMBEDDED (B: 170 – 200 µL COLLAGEN SOLUTION); AND A MAIN COLLAGEN LAYER IN WHICH THE FIBROBLAST ARE EMBEDDED (B: 170 – 200 µL COLLAGEN SOLUTION) FIGURE 4-44: SCM CULTURES COMPOSED OF CALU-3 AND IMR90 CELLS (EMBEDDED IN COLLAGEN I) AFTER A CULTIVATION TIME COLLAGEN LAYER (A: 100 µL COLLAGEN SOLUTION); AND A MAIN COLLAGEN LAYER IN WHICH THE FIBROBLAST ARE EMBEDDED (B: 170 – 200 µL COLLAGEN SOLUTION) FIGURE 4-44: SCM CULTURES COMPOSED OF CALU-3 AND IMR90 CELLS (EMBEDDED IN COLLAGEN I) AFTER A CULTIVATION TIME COLLAGEN LAYER (ALL CONDITIONS). THE MONOLAYERED CALU-3 CELLS ARE CHARACTERIZED BY A COLUMNAR-SHAP A: SLIGHT DEGRADATION OF THE COLLAGEN IS INDICATED BY SMALL CRUMBLES WITHIN THE MATRIX (HE STAINED HISTOLOGIC SECTIONS). B: THE PRESENCE OF MUCUS PRODUCING CELLS WAS VERIFIED BY IMMUNOHISTOCHEMICAL ANALYSES OF 	92 - 93 v 94 v; 97 FE. :AL

FIGURE 4-45: SCM CULTURES PREPARED WITH NHBE CELLS OF PASSAGE 1 AFTER A CULTIVATION TIME OF 29 DAYS. THE CULTURES DID NOT SHOW A CONTRACTION OF THE COLLAGEN MATRIX BUT A HIGH DEGRADATION OR DRYING OUT OF THE COLLAGEN. . 100 FIGURE 4-46: IMMUNOHISTOCHEMICAL ANALYSIS OF THE BASAL CELL MARKER P63 (A) AND THE SQUAMOUS CELL MARKER CK13 (B)

- FIGURE 4-58: NHBE CELLS CULTIVATED IN CCM VARIANT // CO-CULTURE WITH NHLF AFTER A CULTIVATION TIME OF 84 DAYS (79
 DAYS UNDER ALI CONDITIONS). THE CULTURE SHOWED AN EXTENSIVE MUCUS SECRETION AND THE MUCUS ACCUMULATED ON
 TOP OF THE EPITHELIAL CELLS. THIS MUCUS LAYER COULD BE PRESERVED DURING FIXATION AND PARAFFIN EMBEDDING AND WAS
 THUS STILL VISIBLE IN THE HISTOLOGICAL SECTION. IMMUNOHISTOCHEMICAL ANALYSES REVEALED A STRONG STAINING OF
 MUC5AC IN THIS MUCUS LAYER (THE CELL NUCLEI WERE COUNTERSTAINED WITH HEMATOXYLIN; MAGNIFICATION: 400X). 116
 FIGURE 4-59: NHBE CELLS CULTIVATED IN MONOCULTURE (A; 29 DAYS; 25 DAYS UNDER ALI CONDITIONS) AND CCM VARIANT // CO CULTURE WITH IMR90 FIBROBLASTS (B; 38 DAYS; 33 DAYS UNDER ALI CONDITIONS). NHBE CELLS OF THE SAME DONOR AND

FIGURE 4-60: IMMUNOFLUORESCENCE ANALYSIS OF NHBE CELLS (P01) IN CCM VARIANT II CO-CULTURES WITH IMR90 AFTER A
CULTIVATION TIME OF 55 DAYS. CILIA ARE STAINED WITH ACETYL-A-TUBULIN (RED) AND CELL NUCLEI WITH DAPI (BLUE;
MAGNIFICATION: 400x)
FIGURE 4-61: CALU-3 CELLS CULTIVATED IN CCM VARIANT II CULTURES WITH IMR90 (A) OR AS MONOCULTURE (B) AFTER A
CULTIVATION TIME OF 29 DAYS (23 DAYS UNDER ALI CONDITIONS; HE STAINED HISTOLOGICAL SECTIONS; MAGNIFICATION:
400x)
FIGURE 4-62: SCM CULTURE OF HBEPC 908 CELLS (P03) WITH IMR90 FIBROBLASTS AFTER A CULTIVATION TIME OF 19 DAYS (14
days under ALI conditions). The epithelial cells show a stratified growth (A; HE stained histological section)
WITH MULTIPLE LAYERS OF P63 POSITIVE BASAL CELLS (B). THE UPPER LAYERS ARE CHARACTERIZED BY CELLS WITH A SQUAMOUS
MORPHOLOGY WHICH SHOW A HIGH EXPRESSION OF THE SQUAMOUS CELL MARKERS CK13 (C) AND CK6 (D; THE CELL NUCLEI
were counterstained with hematoxylin; magnifications: A-C: 630x; D: 400x) 121
FIGURE 4-63: MONOCULTURES OF NHBE CELLS, LOT 005 (P02) CULTIVATED UNDER IDENTICAL CONDITIONS FOR 23 DAYS (19 DAYS
under ALI conditions). A: In some insert cultures, a stratified growth, squamous morphology and
KERATINIZATION OF THE NHBE CELLS WAS DETECTED. B: IN OTHER INSERT CULTURES, THE NHBE CELLS SHOWED A NORMAL
growth and differentiation pattern (HE stained histological sections; magnification: 400x)
FIGURE 4-64: NHBE MONOCULTURES AFTER A CULTIVATION TIME OF 30 DAYS (26 DAYS UNDER ALL CONDITIONS). THE CULTURES
WERE EITHER CULTIVATED UNDER THE CONVENTIONAL CULTURE CONDITIONS (WITH BASIC DIFFERENTIATION MEDIUM (A) OR
WITHOUT RETINOIC ACID IN THE DIFFERENTIATION MEDIUM (B). CILIATED CELLS WERE DETECTED IN BOTH CULTURES (HE
STAINED HISTOLOGICAL SECTIONS; MAGNIFICATION: 630x)
FIGURE 4-65: DIFFERENTIATED NHBE CULTURES AFTER A CULTIVATION TIME OF 29 DAYS (25 DAYS UNDER ALI CONDITIONS). THE
CULTURES WERE EITHER CULTIVATED UNDER CONVENTIONAL CULTURE CONDITIONS (WITH BASIC DIFFERENTIATION MEDIUM; A)
or with an increased EGF concentration of 25 Ng/mL leading to a severe impairment of the normal
differentiation (B; HE stained histological sections; magnification: 400x)
FIGURE 4-66: DIFFERENTIATED NHBE CULTURES AFTER A CULTIVATION TIME OF 35 DAYS. THE CULTURES WERE EITHER CULTIVATED
under conventional culture conditions (with basic differentiation medium; A) or with an increased EGF
CONCENTRATION OF 50 NG/ML (B-D). THE STIMULATION WITH 50 NG/ML EGF RESULTED IN THE FORMATION OF STRATIFIED
AREAS WHICH WERE IRREGULARLY DISTRIBUTED OVER THE CULTURE SURFACE [(B); A-B: HE STAINED HISTOLOGICAL SECTIONS;
magnification: 200x]. The middle layers of these stratified areas were composed of squamous cells which
SHOWED A POSITIVE STAINING IN IMMUNOHISTOCHEMICAL ANALYSES OF CK13 AND CK6 [(C-D); THE CELL NUCLEI WERE
COUNTERSTAINED WITH HEMATOXYLIN; MAGNIFICATION: 400x]
FIGURE 4-67: RESULTS OF ELISA ANALYSIS OF IL-8 SECRETION (PG/ML) IN NHBE015 MONO- OR CCM VARIANT II CO-CULTURES
and IMR90 monocultures (embedded in collagen) after stimulation with 50 ng/mL compared to control
CULTURES
FIGURE 4-68: NHBE MONOCULTURES AFTER A CULTIVATION TIME OF 28 DAYS (24 DAYS UNDER ALL CONDITIONS). THE CULTURES
WERE EITHER CULTIVATED UNDER CONVENTIONAL CULTURE CONDITIONS (WITH BASIC DIFFERENTIATION MEDIUM; A) OR WITH
an increased EGF concentration of 40 ng/mL leading to basal cell hyperplasia (B; HE stained histological
SECTIONS; MAGNIFICATION: 400x)

Figure 4-69: Immunohistochemical analyses of various markers in NHBE monocultures stimulated with 40 ng/mL
AND CONTROL CULTURES. FURTHER INFORMATION ARE PROVIDED IN THE MAIN TEXT (THE CELL NUCLEI WERE COUNTERSTAINED
WITH HEMATOXYLIN; MAGNIFICATION: 400x)
FIGURE 4-70: RESULTS OF ELISA ANALYSIS OF IL-8 SECRETION (PG/ML) IN NHBE011 MONOCULTURES AFTER STIMULATION WITH
40 NG/ML COMPARED TO CONTROL CULTURES
FIGURE 4-71: NHBE MONOCULTURES OF TWO DIFFERENT LOTS (023 AND 011) AFTER A CULTIVATION TIME OF 28 (NHBE011) AND
29 (NHBE023) days, respectively. The cultures were either cultivated under conventional culture conditions
(with basic differentiation medium; A-B) or with the addition of 40 ng/mL TGF-ß1 for 7 (NHBE023; C) or 8
(NHBE011, D) DAYS LEADING TO ALTERATIONS OF THE CELL MORPHOLOGY FROM COLUMNAR TO SQUAMOUS (HE STAINED
HISTOLOGICAL SECTIONS; MAGNIFICATION: 400x)
FIGURE 4-72: NHBE MONOCULTURES OF TWO DIFFERENT LOTS (015 AND 008) AFTER A CULTIVATION TIME OF 26 (NHBE008) AND
42 (NHBE015) days, respectively. The cultures were either cultivated under the conventional culture
CONDITIONS (WITH BASIC DIFFERENTIATION MEDIUM; A-B) OR WITH THE ADDITION OF 3 NG/ML TGF-B1 FOR 22 (NHBE015;
C) OR 13 (NHBE008, D) DAYS LEADING TO THE FORMATION OF STRATIFIED AREAS WITH AN UPPER LAYER COMPOSED OF
COLUMNAR CELLS AND $2-3$ UNDERLYING LAYERS WITH SQUAMOUS CELLS (HE STAINED HISTOLOGICAL SECTIONS;
MAGNIFICATION: 400x)
FIGURE 4-73: NHBE MONOCULTURES AFTER A CULTIVATION TIME OF 29 DAYS (24 DAYS UNDER ALL CONDITIONS). THE CULTURES
were either cultivated under conventional culture conditions (with basic differentiation medium; A) or with
THE ADDITION OF 10 NG/ML IL-13 FOR 7 DAYS LEADING TO GOBLET CELL HYPERPLASIA (B; HE STAINED HISTOLOGICAL
SECTIONS; MAGNIFICATION: 400x)
FIGURE 4-74: NHBE MONOCULTURES AFTER A CULTIVATION TIME OF 37 DAYS (32 DAYS UNDER ALL CONDITIONS). THE CULTURES
were either cultivated under conventional culture conditions (with basic differentiation medium; A) or with
THE ADDITION OF 10 NG/ML IL-13 FOR 15 DAYS LEADING TO GOBLET CELL HYPERPLASIA (B; A-B: HE STAINED HISTOLOGICAL
sections; magnification: 400x). A strong staining of the mucin 5AC was verified in the IL-13 stimulated
CULTURES BY IMMUNOHISTOCHEMISTRY (C; THE CELL NUCLEI WERE COUNTERSTAINED WITH HEMATOXYLIN; MAGNIFICATION:
400x)
FIGURE 4-75: INVERSE MICROSCOPICAL IMAGES OF 16HBE140- MONOCULTURES AFTER A TOTAL CULTIVATION TIME OF 15 DAYS. THE
CULTURES WERE EITHER CULTIVATED UNDER CONVENTIONAL CONDITIONS WITH MANUAL MEDIUM RENEWAL (A) OR IN THE
CULTEX® LTC FOR 7 DAYS WITH A COMPUTER-CONTROLLED MEDIUM RENEWAL (B) EVERY 48 H. THE CULTURES THAT WERE
CULTIVATED IN THE LTC SHOWED LARGE GAPS IN THE CELL LAYER (B) WHILE THE CONTROL CULTURES HAD A NORMAL, UNIFORM
CELL LAYER (A)
FIGURE 4-76: PROCEDURE OF BIOMATERIAL TESTING. A. THE CELLS ARE SEEDED ONTO THE BOTTOM OF 6-WELL RECEIVER PLATES AND
CULTIVATED FOR 24 H IN THEIR CELL SPECIFIC GROWTH MEDIUM PRIOR TO THE EXPERIMENT. B. THE MEMBRANES OF
COMMERCIAL CELL CULTURE INSERTS (6-WELL FORMAT) ARE DISSECTED WITH A NEEDLE. C. THE TEST SPECIMENS ARE PLACED
ONTO THE PERFORATED MEMBRANE AND THE CELL CULTURE INSERTS ARE APPLIED INTO THE WELLS OF THE RECEIVER PLATE. D.
THE SPECIMEN IS COVERED WITH CELL CULTURE MEDIUM AND THE CELL CULTURES ARE THUS CULTIVATED IN PRESENCE OF THE
TEST SAMPLE FOR SEVERAL DAYS

TO PROMOTE A NORMAL MONOLAYERED GROWTH AND THE COLUMNAR SHAPE OF THE CALU-3 CELLS (C + F). 143

List of Tables

TABLE 4-1: LIST OF ROUTINELY USED MARKERS (ANTIBODIES) FOR THE DETECTION OF IMPORTANT DIFFERENTIATION CHARACTERISTICS
OF BRONCHIAL EPITHELIAL CELLS BY IMMUNOHISTOCHEMISTRY OR IMMUNOFLUORESCENCE
TABLE 4-2: TEER VALUES OF CULTURES PREPARED FROM DIFFERENT IMMORTALIZED CLONAL CELL LINES DEPENDENT ON PASSAGE
NUMBER AND CULTURE AGE
TABLE 4-3: COMPARISON OF TEER VALUES OF EACH THREE INSERTS OF NHBE CELLS EITHER CULTIVATED IN CCM VARIANT I CO-
CULTURES WITH IMR90 FIBROBLASTS OR AS MONOCULTURE AFTER A CULTIVATION TIME OF 19 DAYS
TABLE 4-4: COMPARISON OF TEER VALUES OF CALU-3 CELLS EITHER CULTIVATED IN CCM VARIANT I CO-CULTURES WITH IMR90
FIBROBLASTS OR AS MONOCULTURE AFTER A CULTIVATION TIME OF 15 AND 29 DAYS. ONE INSERT EACH WAS ANALYZED 111
TABLE 4-5: COMPARISON OF TEER VALUES OF NHBE CELLS EITHER CULTIVATED IN CCM VARIANT II CO-CULTURES WITH NHLF
FIBROBLASTS OR AS MONOCULTURE (AVERAGE OF THREE INSERTS EACH ±SD)
TABLE 4-6: COMPARISON OF TEER VALUES OF CALU-3 CELLS EITHER CULTIVATED IN CCM VARIANT II CO-CULTURES WITH IMR90
FIBROBLASTS OR AS MONOCULTURE. ONE INSERT EACH WAS ANALYZED.
TABLE 4-7: LIST OF DIFFERENT APPROACHES FOR THE STIMULATION OF NHBE CELLS WITH INCREASED EGF CONCENTRATIONS. TWO
DIFFERENT NHBE CELL LOTS (011 AND 015) WERE STIMULATED WITH VARYING CONCENTRATIONS OF EGF. THE STARTING
POINT (CULTURE DAY) AS WELL AS THE DURATION OF STIMULATION (DAYS) WAS VARIED. THE STIMULATION OF THE NHBE015
CELLS WITH 50 NG/ML WAS PERFORMED WITH MONO- AND CCM VARIANT II CO-CULTURES (WITH IMR90 CELLS)
TABLE 4-8: TEER VALUES OF NHBE CULTURES (CULTIVATED AS MONO- OR CCM VARIANT II CO-CULTURES) AFTER STIMULATION WITH
INCREASED EGF CONCENTRATIONS COMPARED TO CONTROL CULTURES. THE TEER VALUES OF THE TEST CULTURES ARE LISTED
FOR EACH INSERT INDIVIDUALLY, WHILE THOSE OF THE RESPECTIVE CONTROL CULTURES ARE SHOWN AS MEAN VALUES WITH
STANDARD DEVIATION (±)
TABLE 4-9: LIST OF DIFFERENT APPROACHES FOR THE STIMULATION OF NHBE CELLS WITH TGF-B1. FOUR DIFFERENT NHBE LOTS
(008, 011, 015 and 023) were stimulated with varying concentrations of TGF-B1. The starting point (culture
DAY) AS WELL AS THE DURATION OF STIMULATION (DAYS) WAS VARIED. THE STIMULATION OF THE NHBE008 AND NHBE015
CELLS WITH 3 NG/ML WAS PERFORMED WITH MONO- AND CCM VARIANT II CO-CULTURES (WITH IMR90 CELLS)
TABLE 4-10: LIST OF DIFFERENT APPROACHES FOR THE STIMULATION OF NHBE CELLS WITH IL-13. TWO DIFFERENT NHBE LOTS (008)
and 023) were stimulated with 10 Ng/mL IL- 13 . The starting point (culture day) as well as the duration of
STIMULATION (DAYS) WAS VARIED. THE STIMULATION OF THE NHBE015 CELLS WITH 50 NG/ML WAS PERFORMED WITH
MONO- AND CCM VARIANT II CO-CULTURES (WITH IMR90 CELLS)
TABLE 6-1: OVERVIEW OF CELL LINE-SPECIFIC GROWTH MEDIA, INCLUDING SPECIAL INGREDIENTS OR ADDITIONAL SUPPLEMENTS 151
TABLE 6-2: INFORMATION ON THE TRYPSIN/ EDTA CONCENTRATIONS AND VOLUMES AS WELL AS THE TRYSPIN STOPPING SOLUTIONS
USED FOR THE SUBCULTIVATION OF DIFFERENT CELL LINES
TABLE 6-3. LIST OF ABBREVIATIONS 174
TABLE 6-4: LIST OF LABORATORY DEVICES 176
TABLE 6-5: LIST OF PRIMARY AND SECONDARY ANTIBODIES 176
TABLE 6-6: LIST OF LABORATORY EQUIPMENT
TABLE 6-7: LIST OF LABORATORY CONSUMABLES

ABLE 6-8: LIST OF CHEMICALS AND SOLUTIONS

Abstract

The respiratory tract forms the boundary between the external environment and the body and consequently its lining epithelium is permanently exposed to various pathogens and particles. Major consequences are acute and chronic effects, ranging from simple reversible irritations to persistent inflammation and remodeling of the airway epithelium, which often result in chronic diseases like chronic obstructive pulmonary diseases (COPD), asthma or pulmonary fibrosis. In recent years, the incidence rate of respiratory disorders has steadily increased and they now rank second (after cardiovascular diseases) with regard to morbidity and mortality worldwide.

Numerous cellular and molecular factors are involved in the repair and regeneration of the airway epithelium after damage. Improving our understanding of epithelial function in normal and pathologic conditions may help to develop novel and efficient therapeutics and treatments.

For this purpose, *in vitro* models of the airway epithelium are valuable tools. Unlike animal models, cell and tissue cultures can, in general, be used for a variety of toxicological studies without legislative restrictions and ethical concerns. Moreover, *in vitro* models allow the simultaneous and economic analysis of several cell types specifically under various conditions with numerous analytical methods.

This work deals with the development of *in vitro* tissue equivalents of the bronchial epithelium which exhibit natural differentiation and structural patterns. Critical factors are the use of suitable cell lines and the establishment of appropriate culture conditions. To simulate *in vivo*-like conditions, co-culture system are generated. Here, epithelial cells and fibroblasts are co-cultivated in order to allow their interactions to be comparable to the *Epithelial-Mesenchymal Trophic Unit*. A computer-controlled cultivation system is tested that was specifically designed to enable the maintenance of stable conditions and thus to enhance the reproducibility of well-differentiated 3D models.

To study aspects of acute inflammation and chronic airway diseases, damage of the airway epithelium is tried to be simulated by inducing structural alterations in the developed *in vitro* models using special cultivation conditions.

The developed 3D cell model can be employed for a variety of toxicological studies and may deliver important insight into cellular mechanisms of inflammation, repair and remodeling of the airway epithelium.

Keywords: In vitro models, airway diseases, co-culture cell systems, inhalation toxicology

Kurzfassung

Der Atemtrakt bildet die Grenze zwischen der Außenwelt und dem Körper und folglich ist sein auskleidendes Epithel zahlreichen Pathogenen und Partikeln permanent ausgesetzt. Häufige Folgen sind akute und chronische Schäden, die von einer einfachen und reversiblen Irritation bis hin zu chronischen Erkrankungen, wie chronisch obstruktive Lungenerkrankungen (COPD), Asthma oder Lungenfibrose reichen. In den letzten Jahren ist die Häufigkeit von Lungenerkrankungen kontinuierlich gestiegen und sie sind heute die zweithäufigste Todesursache (nach Herz-Kreislauf-Erkrankungen) weltweit.

Zahlreiche zelluläre und molekulare Faktoren sind an der Wiederherstellung und Remodellierung des Atemwegepithels nach Schädigung beteiligt. Ein verbessertes Verständnis der Funktionen des Epithels unter normalen und pathologischen Bedingungen kann dazu beitragen, neue, effiziente Behandlungsmethoden zu entwickeln.

Zu diesem Zweck sind *in vitro* Modelle nützliche Werkzeuge, da sie für zahlreiche toxikologische Untersuchungen eingesetzt werden können, ohne dass gesetzliche Einschränkungen und ethische Bedenken berücksichtigt werden müssen, wie es bei Tierversuchen der Fall ist. Darüber hinaus bieten *In vitro* Modelle die Möglichkeit, verschiedene Zelltypen unter variierenden Bedingungen mit einer Vielzahl von analytischen Methoden gleichzeitig und kostengünstig untersuchen zu können.

Die vorliegende Arbeit befasst sich mit der Entwicklung von *in vitro* Gewebenachbildungen des Bronchialepithels, welche eine natürliche Differenzierung und Struktur aufzeigen. Entscheidende Faktoren hierbei sind zum einen die Verwendung von tauglichen Zelllinien und zum anderen die Erzeugung von geeigneten Kulturbedingungen. Damit *in vivo-* nahe Bedingungen simuliert werden können, werden Co-Kultursysteme erzeugt, bei denen Epithelzellen mit Fibroblasten gemeinsam kultiviert werden. Auf diese Weise sollen Interaktionen zwischen den beiden Zelltypen wie in der *Epithelial-Mesenchymal Trophic Unit* ermöglicht werden. Ein computergesteuertes Kultivierungssystem wird getestet, welches speziell entwickelt wurde, um stabile Kulturbedingungen zu erzeugen und so die Reproduzierbarkeit der ausdifferenzierten 3D-Modelle zu verbessern.

Um Aspekte einer akuten Entzündung und chronischer Atemwegerkrankungen studieren zu können, wird versucht, eine Schädigung des Atemwegepithels zu simulieren, indem

Strukturveränderungen in den entwickelten *in vitro* Modellen gezielt durch spezielle Kulturbedingungen induziert werden.

Die entwickelten 3D Zellmodelle können für eine Vielzahl von toxikologischen Untersuchungen eingesetzt werden und so wichtige Einblicke in zelluläre Mechanismen während Entzündungs- und Reparaturprozessen und daraus resultierenden Strukturveränderungen liefern.

Schlagworte: *In vitro*-Modelle, Co-Kultur Zellsysteme, Inhalationstoxikologie, Atemwegerkrankungen

1 General Introduction

Legislative restrictions on the use of animals for toxicological studies in the European Union require the development of alternative methods for the evaluation of chemicals and complex atmospheres for human health risk assessment.

Despite the construction of technical devices for such *in vitro* studies, the development of tissue and cell models resembling parts of human organs remains a key factor.

The airways are the boundary between the environment and the body and consequently the airway epithelium plays an important role as a protective physical and functional barrier for inhaled pathogens and toxicants. Impairment of the barrier function, e.g. caused by persistent inflammation, may lead to chronic airway diseases, like COPD. The incidence of such diseases is currently increasing and they are already the leading cause of death in many countries ⁽¹⁾.

The development of realistic *in vitro* models of the human epithelium is of great value for the understanding of barrier properties and physiological processes during inflammation, repair and remodeling of this tissue. The cell models can serve as a complement to or, to a certain degree, as a replacement for traditional animal models in various toxicological studies. Unlike animal models, the effects of different agents on single cell types can be comprehensively analyzed by means of molecular biological or biochemical methods.

Today, cell cultures composed of readily differentiated epithelial cells isolated from different parts of the respiratory tract (upper and lower airways) are commercially available from different suppliers (e.g. Epithelix Sàrl, MatTek Corp.). However, these models generally only consist of epithelial cells, while *in vivo* the airway epithelium comprises different components: The *Epithelial-Mesenchymal Trophic Unit* (EMTU) is an anatomical and functional unit that includes (a) the epithelial layer (b) the basal lamina to which the epithelial cells are attached and (c) the connective tissue (*lamina propria*) with its fibroblasts embedded in an extracellular matrix ⁽²⁾. The *in vitro* simulation of this complex structure is still a major challenge.

The extracellular matrix (ECM) of the connective tissue in the airways is mainly made up of collagen fibers. Fibroblasts, the main cell type of the connective tissue, are scattered in this matrix. These mesenchymal cells have been shown to play a crucial role in epithelial growth,

differentiation and wound repair by secreting cytokines and chemokines in response to various stimuli ⁽²⁻⁴⁾. Thus, to meet *in vivo* conditions as close as possible, *in vitro* models of human airway epithelium optimally include both, epithelial cells and fibroblasts.

The most critical factors for the implementation of 3D tissue equivalents that resemble the airway epithelium, with regard to cell type composition and structure, are the use of appropriate cell lines and the establishment of optimal culture conditions, including nutrition supply and the surrounding atmosphere. Primary epithelial cells isolated from human tissue explants are able to retain their natural characteristics and to form a structure similar to normal epithelia *in vitro* when cultivated under appropriate conditions. However, these cells have a limited lifespan and can only be subcultivated for a few passages until they become senescent. Thus, the reproducibility of established 3D tissue equivalents with these cells might be limited and the use of primary cells obtained from different donors probably leads to heterogeneous results ⁽⁵⁻⁷⁾.

Permanent cell lines are either isolated from carcinomas or they are generated by immortalization of normal cells, e.g. through viral transformation. These cell lines can be subcultivated for a long time and thus establish continuity in the system to be studied. There are several permanent epithelial cell lines of the respiratory tract which are commercially available. However, only a few of them are able to form functional tight junctions and to express characteristics similar to differentiated cells *in vivo* ⁽⁸⁾.

For the differentiation and realistic structuring of lung epithelial cells into basal, mucus secreting and ciliated cells *in vitro*, the use of certain growth factors, vitamins and hormones and their concentrations are decisive factors ⁽⁹⁾. In addition, the type of cultivation plays a key role in the mucociliary differentiation. The air-liquid interface (ALI) has been proven to be essential for airway epithelial cells for mucus secretion and cilia formation (mucociliary differentiation) while the proliferation is enhanced in submerged cultures ⁽⁵⁾.

In summary, the realization of *in vitro* models simulating the structural and functional characteristics of the airway epithelium is dependent on the choice of appropriate cell lines and a balanced cultivation protocol. Moreover, to study the effects of test substances on the airway epithelium, the interactions between mesenchymal and epithelial cells should be taken into account and thus co-culture systems are indispensable for toxicological studies in order to provide near-realistic conditions.

2 Scope and objectives of the thesis

This work deals with the establishment of realistic and reproducible *in vitro* models of the bronchial epithelium for toxicological studies. The achievement of such cell models showing natural differentiation patterns is strongly dependent on the source of epithelial cells. There are some permanent bronchial epithelial cell lines, either immortalized or tumor-derived, which are described to have retained several important characteristics of their normal (non-cancerous) *in vivo*- counterparts when cultivated under appropriate conditions. Some of the most commonly used lung epithelial cell lines, the 16HBE140- and Calu-3, and a rather unknown cell line, ChaGo-K1, are tested in this thesis for their suitability to serve as a *in vitro* model of the human bronchial epithelium.

In addition, newly developed immortalized cell lines that have not been characterized before, are comprehensively tested for their (1) general cell viability, (2) life span and (3) ability to develop natural differentiation patterns.

Since primary (non-transformed) cells isolated from the target tissue are the best source for cell culture models with regard to the reflection of realistic *in vivo* conditions, an easy and efficient protocol for the isolation of normal human bronchial epithelial (NHBE) cells from lung tissue is developed. Primary cells are known to show donor-specific variations and thus, all isolated NHBE cells are characterized for their cell viability and differentiation capacity. In order to guarantee the availability of these cells, the suitability of cryopreservation of NHBE cells and bronchial tissue samples is evaluated.

Subsequently, the most suitable epithelial cell lines are used for the generation of 3dimensional tissue equivalents of the bronchial epithelium. These cell models are composed of bronchial epithelial cells and fibroblasts and are intended to simulate natural interactions of the EMTU. Two different approaches are tested for the implementation of such 3D coculture models: The Sandwich Cell Model (SCM) reproduces the structure of the bronchial epithelium and provides a direct contact between the two cell types. In the Compartment Cell Model (CCM), on the other hand, only an indirect contact between the lung epithelial cells and fibroblasts is enabled but here, both cell types can be analyzed separately. The commercial IMR90 fetal human lung fibroblast strain is primarily used as mesenchymal component in the bronchial epithelial tissue equivalents. In addition, a protocol for the isolation of normal human lung fibroblasts (NHLF) from bronchial tissue samples is developed, in order to test whether the co-cultivation with adult fibroblasts shows distinct effects on epithelial cells.

The automated Long Term Cultivation module (CULTEX[®] LTC) developed by Cultex Laboratories GmbH was specifically designed for the cultivation of cell cultures on insert membranes that are cultivated at the air-liquid-interface. The system enables a computer-controlled medium supply and is intended to increase the stability and reproducibility of the culture conditions. The practicability and suitability of this cultivation method for different cell lines is tested for the first time in this thesis.

The ability to simulate natural pathologic features of airway inflammation after stimulation with certain substances is another requirement for suitable *in vitro* models for toxicological studies. Therefore, a further aim of this thesis is the development of protocols for the induction of inflammation states in bronchial epithelial cells *in vitro*. For this purpose, different signaling molecules (cytokines) or growth factors that are known to be related to inflammation or irritation of the airway epithelium are added to the culture medium during the differentiation phase of the cell cultures. The cellular response to these additives is analyzed by means of several analytical methods.

3 Theoretical Framework

3.1 Toxicological Testing

Toxicology is "the study of the adverse effects of chemical, physical or biological agents on living organisms and the ecosystem, including the prevention and amelioration of such adverse effects" ⁽¹⁰⁾. Most developed countries today have enacted laws and regulations for the marketing of drugs, food additives and other substances of toxicological concern. To evaluate the risk to human health or the environment, guidelines for toxicity testing are prescribed to enable public authorities to calculate the risks of particular substances ⁽¹¹⁾. For example, REACH is the European Community Regulation on chemicals and their safe use which came into force on 1 June 2007 ([EC] No 1907/2006). It deals with the **R**egistration, **E**valuation, **A**uthorisation and Restriction of **Ch**emical substances. This regulation places greater responsibility on industry to manage the risks from chemicals and to provide safety information on the substances.

Chemicals can have many types of harmful effects: some cause allergic reactions, others irritate the eyes or skin. More severely, chemicals might also be genotoxic, carcinogenic or reproductively toxic. The latter effects are mostly not immediately detectable but require long-term studies. The most critical factor for the risk assessment of substances is the selection of appropriate toxicity test systems. The use of living animals for the evaluation of adverse effects of chemicals began in the 1920s when the LD₅₀ test was introduced by the British pharmacologist J.W. Trevan ⁽¹²⁾. This test aims to classify of the acute toxicity of substances by determining its single dose that would kill half of the animals exposed to it. Later, animal testing gained more and more attention in the 1940s with the development of the standardized test for eye and skin irritation on albino rabbits, the "Draize test" ⁽¹³⁾. A few years later, the US National Cancer Institute established the first standardized test for the identification of carcinogenicity of chemicals through the daily exposure of rats and mice to the test substances for up to two years ⁽¹¹⁾.

Due to the increasing importance of chemicals and pharmaceuticals during the 80s and consequently the augmentation of animal testing, the Organisation for Economic Cooperation and Development (OECD) devised harmonized test guidelines that should regulate the use of animals for toxicological assessment internationally. At the same time, the call for alternatives to animal testing for toxicological investigations increased steadily for several reasons. Firstly, the general public and scientific community have raised concerns about the suffering of experimental animals since some of the conventional test methods included highly painful procedures and sometimes consumed large numbers of animals for testing a single substance. The latter aspect has secondly led to a debate about the enormous costs and the low throughput of animal experiments. Another point that has raised concerns is the reliability and relevance of animals to the human health risk is a critical point ⁽¹⁴⁻¹⁵⁾. It is impossible to exactly know whether a test on rodents or dogs will provide a reliable prediction of toxic effects in humans ⁽¹⁶⁾. Moreover, the relevance of the tested doses of the substances is questionable with regard to realistic human exposure levels ⁽¹⁷⁾.

Alternative toxicological test methods are those that enable the reduction, refinement or replacement of animal experiments ⁽¹⁸⁾. The "Refinement Alternatives" include methods that eliminate or minimize pain and distress, for example through enhancing the experimental design and handling of the animals ⁽¹⁹⁾. "Reduction Alternatives" are strategies that will result in lower number of animals that are necessary to obtain sufficient data ⁽¹⁸⁾. This might be achieved by using novel exposure technologies and by implementing statistical methods to select sample sizes ⁽¹⁹⁻²⁰⁾. Moreover, duplication of experiments should be avoided by providing public access to the information of animal studies ⁽²¹⁾. Acceptable "Replacement Alternatives" are non-animal approaches that allow the classification and labeling of substances in the same way as current animal experiments allow ⁽²²⁾. Here, the most promising test systems are cell and tissue culture models that enable the rebuilding of organs *in vitro*. Advances in cell culture techniques and tissue engineering gave rise to emerging high-throughput technologies, like automated cultivation systems and robotics, providing an cost- and time- efficient methodology for toxicity testing.

The present work focuses on the development of *in vitro* cell culture models of the airway epithelium that can be used for inhalation toxicological studies as an alternative method to current animal models. Inhalation of airborne substances is one of the major contributors to the human health problems. Humans are permanently exposed to particulate, vapor and gaseous contaminants ⁽²³⁾. Due to the fact that the incidence of lung diseases is increasing

⁽²⁴⁻²⁵⁾ and is the major cause of lethality in many countries (see section 3.4.3), the studies in the field of inhalation toxicology today mainly concentrate on the understanding of mechanisms of lung injury and repair after exposure to toxicants ⁽²⁶⁾. Studying the adverse effects of inhaled substances is still a major challenge with regard to the design of exposure devices allowing the generation of controlled atmospheres, as well as the development of appropriate test systems for providing relevant data ⁽²⁷⁻²⁸⁾.

3.2 Legislative Regulations for Animal Experiments

In 1986, the OECD agreed to promote the refinement and reduction of animal testing and the development of *in vitro* techniques as alternatives for toxicological assessments. *Directive 86/609/EEC* and in particular its revised version 2010/63/EU⁽²⁹⁾ regulates the use of laboratory animals for scientific purposes in the EU in concordance with the "Three Rs" principle postulated by Russell and Burch, the *Refinement, Reduction and Replacement* of animal testing whenever possible ⁽¹⁸⁾. Article 7.2, for example, declares "an experiment shall not be performed if another scientifically satisfactory method of obtaining the result sought, not entailing the use of an animal, is reasonably and practicably available". Moreover, article 47 states that the Commission and the member countries should encourage the development and validation of alternatives to animal testing.

In 1991, the European Centre for the Validation of Alternative Methods (ECVAM) was established to promote the development and to coordinate the validation of such alternative methods in the Member States ⁽³⁰⁾. In 1997, the USA followed this example and established the Interagency Coordinating Center for the Validation of Alternative Methods (ICCVAM).

In Europe, the Member States also have national organizations that encourage and support the development and implementation of alternative test methods, like the Zentralstelle zur Erfassung und Bewertung von Ersatz- und Ergänzungsmethoden zum Tierversuch (ZEBET) in Germany. Promising methods are submitted as a proposal to the ECVAM which decides if it is suitable to enter first the small-scale prevalidation and, if successful, afterwards the validation phase. This official acceptance requires the existence of an agreed protocol for the method. The validation process includes different stages and for each stage a set of criteria

has to be fullfilled ⁽³¹⁾. The starting point is the test definition which includes the description of the scientific purpose of the test and the determination of appropriate endpoints and protocols. The next few stages should show the reliability of the method by evaluating (1) the within-laboratory variability, (2) the transferability and finally (3) the inter-laboratory variability. Subsequently the relevance is tested by applying the test method to reference standards (e.g. results of previous *in vivo* studies) and by determination of the applicability domain. As soon as all stages have been passed successfully, a final independent peer review process is entered ⁽³¹⁾. Once an alternative method has been validated, it has to be accepted by ECVAM for regulatory purposes.

International acceptance can be achieved by presenting the ECVAM-validated test methods to the OECD. Here, the method is evaluated according to the OECD validation principles and acceptance criteria to ensure that the new test will allow chemicals to be classified and labeled in the same way as current animal tests ^(22, 32). Internationally accepted alternative methods are recorded in OECD guidelines and will be implemented by regulatory agencies in all OECD Member States. The adoption of an alternative method in OECD guidelines is the last stage in the regulatory acceptance process and it is a significant step towards the replacement of animals for this particular purpose.

The development of alternative methods for toxicology has gained more attention since the adoption of the Seventh Amendment to the EU Cosmetic Directive 2003/15/EC which prescribed a phase-out of animals for testing finished cosmetics products from 2004 and for cosmetic ingredients latest from March 2009. A total ban of animal experiments for cosmetic evaluation, including for repeated-dose toxicity, reproductive toxicity and toxicokinetics, has only recently came into force on 11 March 2013. The time estimation of the phase-out was based on the presence of accepted alternative *in vitro* methods for cosmetic-related toxicity testing (e.g. skin sensitization/ corrosion, skin absorption/ penetration, genotoxicity or mutagenicity) or the progress in their development ⁽³³⁾.

Several animal rights organizations call for the expansion of the ban of animal testing to other industries, like tobacco products. It is conceivable that the enormous progress being made in the development of alternative methods might lead to the implementation of an animal phase-out also for other industry branches in the near future.

3.3 Alternative Methods

Recent advances in tissue and cell culture methods, molecular biology and computational techniques have led to a variety of alternative toxicological test methods that are under validation or have already been accepted by ECVAM or ICCVAM or even internationally by the OECD according to the respective validation criteria. These alternative tests include, among others, *in vivo* methods that require a reduced number of animals compared to previous tests for a particular purpose, for example due to the support of *in vitro* data. Furthermore, there are currently a number of accepted protocols for *in vitro* tests available which make use of reconstituted tissue (i.e. skin models), human or other mammalian cell lines or microorganisms, depending on the respective purpose (e.g. microorganisms are used for testing the genotoxic potential of chemicals).

The use of cell and tissue cultures for studying cellular aspects outside of the animal body started in the early 20th century. Today, cell culture has become one of the major tools used in many different fields of the life sciences. The recent changes in the legislative regulations for toxicological testing have led to an increased usage of cultured cells or reconstituted tissue for a variety of tests even outside the regulatory framework.

The overall aim of toxicology today is to maximize the information of the toxic potential of chemicals while reducing financial costs and testing time as well as minimizing the use of laboratory animals. This can be achieved by so-called *Integrated Testing Strategies* (ITS) which combine a variety of existing data from *in vitro* tests and computational toxicology ⁽³⁴⁾.

In the following sections, some of the most important non-animal-based approaches for toxicological assessment are described with special focus on cell culture methods.

3.3.1 Integrated Testing Strategies (ITS)

ITS are combinations of test batteries which include relevant mechanistic steps and are organized in a logical, hypothesis-driven decision scheme ⁽³⁵⁾.

The first step of each ITS is to assess all prior data that are related to the chemical to be tested. These data might already lead to the decision whether a chemical can be rejected as likely being too hazardous or not ⁽³⁴⁾. If not, in the next step, the bioavailability of the chemical and the resultant target organs are determined by relevant algorithms or tests on

available co-culture cell systems. This information is then used to define the type of hazard more specifically. This is mostly first performed by computational methods, followed by *in vitro* tests ⁽³⁴⁾. Only in the ultimate step, if necessary, will animals be employed for a final decision on the toxic potential of the tested substance.

The need for more research on ITS and the usefulness of this approach has recently been discussed ⁽³⁵⁾. However, the practicability and suitability of ITS, of course, is strongly dependent on the availability of existing data and *in vitro* test systems for the respective chemical or target organ.

3.3.2 Tissue Engineering

Tissue engineering was developed to create biological tissue substitutes that can be used for repair and replacement of damaged tissue and organs ⁽³⁶⁾.

The principle of tissue engineering is to isolate cells and to expand them in culture on an appropriate scaffold. The differentiated tissue constructs are afterwards reimplanted into the body ⁽³⁷⁾.

The scaffold materials have to fulfill different requirements depending on the tissue that is intended to be engineered. In general, they have to be biocompatible and should support the attachment, growth and differentiation of the respective cell type ⁽³⁸⁾. Different scaffold materials show varying levels of mechanical stability and porosity and are selected according to the requirements of the target organ. Ideally, the materials are biodegradable, which means that they will degrade as cells deposit their extracellular matrix molecules ⁽³⁶⁾.

Engineered skin, cartilage and bone tissue are amongst the most advanced areas of tissue engineering and some products are currently in clinical use [Apligraf[®](skin), Dermagraft[®] (skin), Orcel[®] (skin)^r, CARTICEL[®](cartilage), Hyalograft[®] C (bone)]. Several other tissue-engineered products are in clinical trials ⁽³⁶⁾.

However, besides their use for clinical purposes, engineered tissues can also be employed for *in vitro* testing, e.g in toxicology and cell biology. At this point, the fields of tissue engineering and recent cell culture techniques, like 3D cell cultures, overlap.

3.3.3 Cell Culture

Most plant and animal cells can be maintained, expanded and even be induced to express differentiation properties in a culture dish, given appropriate conditions ⁽³⁹⁾. This was first shown in 1885, when Roux achieved the maintenance of embryonic chicken cells in a saline solution outside the animal body ⁽⁴⁰⁾.

At the beginning of the 20th century, researchers started to use small tissue explants to study cellular behaviour and functions under normal and stress conditions outside the animals. In the following years, the use of disaggregated explanted cells obtained by outgrowth from these isolated tissue cultures became more prominent for these experiments ⁽⁴¹⁾.

The first continuous cell line was introduced in 1943 by Earle and colleagues. This cell line was established by isolating single cells from the mouse L cell line and cloning them in culture. The cell line, known as L929 fibroblastic cell line, is today still widely used for standard testing, such as biomaterial testing (see section 3.6.2) ⁽⁴²⁾. The first human continuous cell line was established from a human cervical carcinoma in 1952 by Gey and colleagues and later became the well-known HeLa cell line ⁽³⁹⁾.

In 1952 Dulbecco described a procedure for the generation of replicate subcultures by using the enzyme trypsin which cleaves the proteins bonding the cultured cells to the dish, so that the cells can be suspended and transferred to fresh dishes. This method for subcultivation of cells and the development of chemically defined media promoted the use of cell cultures routinely for a diversity of testing until today ⁽⁴¹⁾.

Besides clonal cell lines of primary cells, a variety of continuous cell lines, either cancerderived or immortalized, of different cell types from nearly all organs are commercially available today.

Primary cells are those that have been directly isolated from tissue of an organism without further passaging *in vitro*. Subcultivation of these cells results in so-called secondary cultures which can be, depending on the cell type, repeatedly subcultivated for weeks or months ⁽³⁹⁾. Primary cells are able to successfully imitate the properties of their *in vivo* –counterparts when appropriate surroundings are established ⁽⁷⁾. However, in contrast to cell lines, the number of population doublings is highly restricted in primary cells before they reach replicative senescence ⁽⁴³⁾. Thus, their availability is considerably limited and, when using

cells from different tissue samples, donor-specific variations are likely to occur ⁽⁴⁴⁾. In general, the reproducibility of the experiments performed with primary cells is limited and thus, the use of primary cells for routine testing, like screening assays, is not favorable ⁽⁷⁾.

Replicative cell senescence is caused by the structure of the telomeres in many human somatic cells. Telomere shortening occurs in cells that are deficient for the enzyme called telomerase and leads to cell cycle arrest. Most cancer cells have regained the ability to produce telomerase and therefore have an almost unlimited proliferation ability ⁽³⁹⁾. For that reason, isolated cancer cells are often used for *in vitro* studies, especially in standard testing and screening assays. These cells bring continuity into the test system and thus provide reproducible data. However cancer-derived cells often show characteristics distinct from normal cells ⁽⁴⁵⁾. For example, they can often proliferate in a higher density in culture dishes and are not dependent on ECM (extracellular matrix) proteins for adherence. In addition, they mostly lack many important differentiation properties ^(8, 39).

Besides cancer cells, permanent (or continuous) cell lines can also be established from primary cells by genetic transformation of viral oncogenes, like the Simian Virus 40 (SV40) large T antigen, or by introducing the catalytic subunit of human telomerase (hTERT) into the cell ⁽⁴⁶⁻⁴⁸⁾. These immortalized cell lines have a prolonged life span and are therefore especially suitable for routine tests. However, they might have lost some important characteristics of primary cells and often show a decreased sensitivity to certain agents or conditions. But, given appropriate culture conditions, some of these cell lines show a similar phenotype to primary cells and have been proven to be suitable for particular studies ⁽⁵⁾.

3.3.3.1 Three-dimensional cell culture

In the conventional 2D cell cultures, the cells are grown on flat surfaces in monolayers. This type of cultivation is easy and convenient and was shown to be a valuable tool for the fast identification of toxic compounds, as in the biomaterial testing (see section 3.6.2). Moreover, for the maintenance of cells in culture, they are also grown under 2D conditions.

However, numerous studies revealed that cells under 2D conditions have some limitations with regard to their physiological relevance ⁽⁴⁹⁻⁵⁰⁾. *In vivo*, cells grow in a three-dimensional manner and they interact with other cell types or their surrounding environment, like the
extracellular matrix (ECM) ⁽⁵¹⁻⁵²⁾. 2D cell culture conditions affect the natural phenotype of cells and accordingly also their response to certain stimuli ⁽⁴⁹⁾.

As a consequence of the increasing recognition of the limitations of 2D cell cultures, a variety of 3D cell cultures has recently been developed. These advanced cell culture systems have been proven to mimic *in vivo* conditions more closely and to considerably enhance toxicological and drug absorption studies ^(49, 51, 53). There are different methods for the cultivation of cells in a 3-dimensional manner. The most commonly used systems are multicellular spheroids and hydrogels. Spheroids are spherical clusters of cells that are formed under distinct conditions, for example when the cells are cultivated in spinner flasks, on agar plates or in hanging drop ⁽⁵⁴⁻⁵⁷⁾. Under these conditions, the cells can survive for several weeks and they develop many properties of normal tissue *in vivo*, amongst others, the formation of a natural ECM ^(49, 56). Multicellular spheroids are often employed for cancer research, since they are able to develop distinct regions under certain conditions that are also found in tumors, with a proliferating outer circle, a quiescent intermediate region and a necrotic inner core ^(49, 55).

As in tissue engineering, 3D cell cultures can also be established by using 3D scaffolds into which the cells can invade. Hydrogels were proven to be highly suitable for the 3D cultivation of many cell types since these crosslinked networks share many properties of the ECM *in vivo* ⁽⁵⁸⁾. Hydrogels can be composed of natural components, such as ECM proteins, or prepared synthetically. Synthetic hydrogels, however, lack important biological characteristics and cells grown in these matrices fail to develop certain surface receptors ⁽⁵⁸⁾.

The use of hydrogels prepared from natural ECM proteins is therefore advantageous in order to promote the natural cell behaviour. Collagens are the major extracellular components of the connective tissue and collagen type I is the most abundant ECM protein in mammals ⁽³⁹⁾. Matrices prepared from collagen type I are commonly used for the cultivation of various cells in a 3-dimensional manner (see 3.3.3.1.1).

In a more sophisticated approach of 3D cell cultures, different cell types are combined in coculture systems in order to additionally consider their interactions. For example, to study the function of epithelia under certain conditions, the inclusion of fibroblasts in the *in vitro* models is advantageous, since they play a critical role for the growth, differentiation and regeneration of epithelial cells *in vivo* ⁽³⁻⁴⁾. For that purpose, collagen matrices with

17

embedded fibroblasts are often applied since this structure resembles the connective tissue which underlies the epithelial layer *in vivo*. This co-culture model was successfully used for reconstruction of the skin and cornea ⁽⁵⁹⁻⁶¹⁾.

Special co-culture models with cell types of the respiratory system are further detailed in section 3.5.3.

3.3.3.1.1 Collagen matrices for 3D cell culture

Matrices prepared from collagen type I are often used for 3D cell cultures in order to mimic the ECM of the connective tissue where collagen I is a predominant molecule ⁽⁶²⁾.

In vivo, the cells are embedded into the ECM which does not only provide mechanical support but also plays a central role in their proliferation, differentiation and morphology and influences the gene expression ^(58, 63). These interactions are mediated through signaling via specific surface receptors, primarily integrin receptors, of the collagen fibrils ⁽⁶⁴⁾.

There are many published protocols for the isolation and preparation of collagen I from various organisms and tissues. Moreover, a variety of collagen type I products are commercially available. For these reasons, collagen type I provides a good source as a natural scaffold and has been widely used for the establishment of 3D cultures. For example, several studies have shown the suitability of collagen-based hydrogels for the cultivation of fibroblasts. Here, reciprocal interactions between the matrix and the fibroblasts were observed and it was shown that this cell culture system provides a good model for studying matrix remodeling, an event which is important for the maintenance of tissue homeostasis (65-66).

Collagen matrices form a meshwork of fibrils which polymerizes from collagen monomers. The polymerization occurs when the collagen solution is brought to physiological conditions, with a neutral pH and a temperature of 20-37 °C. The preparation and storage of collagen solutions, on the other hand, requires an acidic pH (around 2) and the cooling to 2-8 °C ⁽⁶²⁾. There are numerous factors that determine the biochemical and physical characteristics of the collagen matrix and consequently influence the cell-ECM interactions. These factors include (a) the collagen source (tissue and species of origin, pepsin or acid extraction) ⁽⁶⁷⁾, (b)

the collagen concentration and (c) the polymerization protocol (pH and temperature variations) ⁽⁶⁸⁾. For example, the presence or absence of telopeptides in the collagen

molecules highly influences the fibril growth and assembly. The pepsin extraction of collagen type I cleaves off the telopeptides from the molecules whereas they are preserved in the acid extraction method ^(62, 64). Another factor that has to be taken into account is the lot-to-lot and product-to-product variation, due to a lack of standardized requirements for the characterization of collagen products.

In summary, the preparation of suitable collagen matrices for 3D cell cultures is a challenging task and mostly requires a time-intensive optimization of the protocol. However, the use of collagen enables the simulation of a natural environment for cells and is thus valuable for the construction of realistic 3D tissue equivalents *in vitro*.

3.4 The Respiratory System

The respiratory system is mainly designed to deliver inspired air to the parts where gas exchange takes place and to transport gaseous waste, like carbon dioxide out of the body. At the same time, the inspired air is humidified, warmed and filtered ⁽⁶⁹⁾. The major parts of the respiratory system through which air passes from proximal to distal are the nasal cavity, the pharynx, the larynx, the trachea, the bronchi, the bronchioles, the terminal bronchioles, the respiratory bronchioles, the alveolar ducts, the alveolar sacs, and finally the alveoli ⁽⁷⁰⁾.

The lung, with special focus on its conducting portion, the bronchi and proximal bronchioles, will further be described in the following sections.

3.4.1 The Lung

The lung is the boundary between an organism and its surrounding environment. It comprises the bronchi, bronchioles and the alveoli. The bronchi and proximal bronchioles are conducting airways, which means, that their major task is the conduction of air to and from the respiratory portion of the lung. The respiratory airways include the remaining parts of the lung where the gas exchange takes place - the respiratory bronchioles, alveolar ducts, alveolar sacs, and alveoli ⁽⁶⁹⁾. Here, the inspired oxygen is transferred to the blood circulation.

In humans, the trachea divides into the two main bronchi that enter the roots of the lungs. The bronchi continue to divide like branches of a reversed tree within the lung ⁽⁷¹⁻⁷²⁾. The

trachea and bronchial tubes are surrounded by cartilage and smooth muscle. The amount of cartilage progressively decreases with continuing branching of the bronchi and is absent in the bronchioles. On the other hand, smooth muscle increases with decreasing cartilage, enabling the constriction of the bronchioles and thus controlling the airflow ⁽⁷³⁻⁷⁵⁾.

The next generation of branches following the main bronchi are the lobar bronchi. These bronchi each enter one of the five lung lobes, three into the right and two into the left lung (Figure 3-1). The lobar bronchi divide into the segmental bronchi which supply the lung segments, ten in the right and eight in the left lung. With each branching, the diameter of the bronchi decreases. After further multiple divisions, the bronchi give rise to the bronchioles which have a diameter of 1 mm or less and which in turn divide into bronchioles with a diameter of 0.5 mm or less ⁽⁷¹⁻⁷²⁾.

Terminal bronchioles branch into respiratory bronchioles which merge into alveolar ducts. Terminal bronchioles mark the end of the conducting airways while the respiratory bronchioles are the beginning of the respiratory division. The alveolar ducts are formed by a series of alveoli clusters, the alveolar sacs. The alveoli are ordered like individual grapes within a bunch. The individual alveoli are tightly wrapped in blood vessels which enable the gas exchange between the lungs and the blood circulation (Figure 3-1)⁽⁷⁰⁾. The alveoli contain neither cartilage nor smooth muscle⁽⁶⁹⁾.

The luminal surface of the airways is lined by the airway epithelium. Each region of the respiratory system is characterized by a unique composition of epithelial cell types. The epithelium of the lung, comprising the bronchi, bronchioles and alveoli, is detailed in the following section.



Figure 3-1: Structure of the bronchial tree and the individual alveolar sacs (adapted from Wikipedia).

3.4.2 The lung epithelium

The larger airways, e.g. the trachea and bronchi, are lined by a pseudostratified, columnar epithelium mainly consisting of ciliated cells interspersed with mucus-producing goblet cells and the multipotent basal cells ⁽⁷⁶⁾. The base of each cell reaches the basement membrane, and therefore it is a simple, not stratified epithelium (Figure 3-2). In healthy humans, the ciliated cells are the predominant cell type in the bronchi followed by the mucus-producing goblet cells ⁽⁷⁷⁾. During inflammation, this ratio is often found to shift. Here, goblet cells or dedifferentiated squamous cells are mostly more abundant than ciliated cells (goblet cell hyperplasia or squamous metaplasia, see section 3.4.2.1).



Figure 3-2: Structure of the bronchial epithelium. A: Histological section of the human lung showing the bronchial epithelium [hemalaum-eosin (HE) stained]; B: Schematic drawing of the bronchial epithelium;1: Mucus-secreting goblet cell; 2: Ciliated cell; 3: Basal cell; 4: Basement membrane (collagen type IV); 5: Fibroblasts; 6: *Lamina propria* mainly composed of ECM proteins.

In the small bronchioles (\leq 1mm), the ciliated epithelium is columnar to cuboidal-shaped and shows no pseudostratification ⁽⁸⁾. Here, basal cells are infrequent and the number of goblet cells decreases in correlation with the diameter of the bronchiole and are lacking in the terminal bronchioles ⁽⁶⁹⁾. By contrast, in these small airways, the epithelium exhibits another secretory cell type, the Clara cells, which are most numerous in the terminal bronchioles ⁽⁷⁸⁾. Clara cells produce the surfactant apoproteins A and B and presumably contribute to the host defense and airway regeneration ⁽⁷⁹⁾.

The bronchial and bronchiolar epithelial cells are anchored to the basement membrane (basal lamina) by adhesion molecules. A major component of the basement membrane is collagen type IV ⁽⁸⁰⁾. Besides the anchorage of epithelial cells, the basement membrane has a variety of other functions and plays a central role in the development of the phenotype of the epithelium ⁽⁷⁶⁾. The *lamina propria*, a thin layer of loose connective tissue, lies beneath the basement membrane. It is composed of ECM proteins, mainly collagen type I fibers and a cellular component, the fibroblasts, which are embedded in the ECM. The epithelial cells and the *lamina propria*, including the ECM and fibroblasts, form the EMTU which is characterized by complex reciprocal interactions between the three components ^(2, 81-82).

Beneath the *lamina propria* lies the submucosa, a thick layer of connective tissue which includes the serous und mucous glands. Submucosal glands are complex structures comprising multiple tubules that feed into a large primary collecting duct (Figure 3-3) ⁽⁸³⁾. The tubules are lined with mucous cells and serous cells. The primary duct passes from the

surface epithelium into the submucosa and collects the secretion products of the mucous and serous cells ⁽⁸⁴⁾. Mucous cells of the submucosal glands primarily secrete the mucin 5B while the watery secretory product of the serous cells is rich in antimicrobial agents and antioxidants ⁽⁸³⁻⁸⁴⁾. Submucosal glands contribute greatly to the mucociliary clearance of the airway epithelium (see section 3.4.2.1). Their number progressively decreases from the proximal to distal airways ⁽⁷⁵⁾.



Figure 3-3: Histological section of the human lung showing the bronchial epithelium and a submucosal gland, including a primary collecting duct and secondary tubule (HE staining, Magnification: 200x);1: Bronchial epithelium; 2: Primary collecting duct of submucosal gland; 3: Tubule of submucosal gland

The human alveolar epithelium with its large surface area of approximately 102 m² mainly consists of two cell types, type I and type II pneumocytes (alveolar epithelial cells, AEC). Although the AEC type II are more numerous, the AEC type I constitute >93% of the alveolar surface due to their large size and flat morphology and accordingly they form the structure of the epithelium ⁽⁸⁵⁾. The AEC type II secrete the pulmonary surfactant that decreases the surface tension of the alveoli and thus prevents them from collapsing and furthermore contains antioxidants and molecules for the defense against microorganisms. The AEC type II also serve as a progenitor for both AEC type I and II ⁽⁸⁶⁾. Another cell type found in the alveolar epithelium is the alveolar macrophages that play an essential role in host defense and tissue remodeling ⁽⁸⁷⁾.

In the alveoli, the basal lamina fuses with that of the capillaries. This ultrathin barrier enables the gas exchange (< 0.5 μ m) and is potentially the main portal for pharmaceuticals into the circulatory system.

3.4.2.1 Function and role of the lung epithelium

As a boundary between the outer environment and the respiratory portion of the lung, the epithelium lining the conducting airways forms a barrier for pathogens and particles ⁽⁸⁸⁻⁸⁹⁾.

This barrier function is driven by two major mechanisms, (1) the mucociliary clearance and (2) the formation of a physical barrier by adherence molecules.

Nevertheless, the permanent exposure to airborne substances and pathogens makes the airway epithelium vulnerable to inflammation. Mostly, inflammation processes and remodeling of the epithelium are reversible states. However, a permanent burden may lead to the development of chronic airway diseases, such as Chronic Obstructive Pulmonary Diseases (COPD) and asthma ⁽⁹⁰⁾.

The following paragraphs give a brief overview of the barrier function mechanisms and important processes during inflammation, repair and remodeling of the airway epithelium.

Mucociliary clearance

The apical surface of the epithelium in the bronchi and proximal bronchioles is lined by a liquid which is composed of two distinct layers, the mucus layer and the periciliary layer ⁽⁹¹⁾. The mucus is a viscoelastic gel which is mainly composed of water (~95%), salts and carbohydrate-rich glycoproteins (mucins) ⁽⁹²⁾. The main airway mucins are the MUC5AC and MUC5B ⁽⁹³⁾. MUC5AC is predominantly secreted by the goblet cells in the proximal airways, whereas the secretory cells of the submucosal glands mainly produce MUC5B ⁽⁹⁴⁾. The thickness of the mucus layer decreases in correlation with the airway diameter and is thickest on top of the pseudostratified epithelium of large airways (up to 50 μ m). The mucus layer functions as a trap for inhaled particles. The periciliary layer, on the other hand, has a low viscosity and thus enables the cilia movement ⁽⁹¹⁾. Ciliated cells account for more than 50% of all epithelial cells in the airways. They each have up to 300 cilia at their apical surface, moving in a coordinated manner from the distal airways to the pharynx, to transport mucus-trapped particulate matter out of the lungs ⁽⁷⁶⁾. The effective mucociliary clearance is

essential not only to keep the airways clean but also to maintain the normal mucus balance. Consequently, failure of the mucociliary clearance can cause various airway diseases ⁽⁹¹⁾.

Physicochemical barrier

As a complement to the mucociliary clearance, the epithelial cells form a physical barrier for inhaled particles through various cell adhesion molecules which form the cell-cell junctions (intercellular bridges), including tight junctions (TJ), adherence junctions, gap junctions and desmosomes ⁽⁹⁵⁻⁹⁶⁾. Among them, TJs are most important for maintaining the epithelial integrity ⁽⁹⁶⁾. These structures are located at the apicolateral borders of adjacent epithelial cells. Besides providing a physical barrier, TJs are also responsible for the selective regulation of the passage of ions and other molecules through the paracellular space and for the maintenance of an ionic gradient in order to allow the directional secretion of TJs during inflammation may play a critical role in the development of chronic lung diseases ⁽⁹⁷⁾.

Inflammatory response, repair and remodeling

There are a variety of factors that can cause injury to the lung epithelium, such as pathogen infection, inflammation, allergic reactions, physical trauma or the exposure to xenobiotics, like cigarette smoke or diesel exhaust ⁽⁹⁸⁾. Epithelial repair is rapidly induced after injury and the initial response can be highly diverse, depending on the type of injury ⁽⁹⁸⁻⁹⁹⁾. This response may include the release of cytokines and recruitment of inflammatory cells. A common event that is described to occur early after epithelial injury is the de-differentiation and migration of epithelial cells in order to close the wound. Proliferation is induced afterwards to replace the lost epithelial cells ^(81, 88, 98-100).

Remodeling is a normal response to injury and inflammation in the airway epithelium ^(90, 101). However, repeated events of injury can lead to persistent alterations, which are often precursors for chronic airway diseases ⁽¹⁰²⁾. The different diseases are each characterized by specific features, including unique remodeling patterns and cytokine and chemokine secretion profiles ⁽¹⁰³⁾.

Common transitory structural alterations during repair and regeneration of the injured epithelium are the goblet- or basal cell hyperplasia and squamous metaplasia ⁽¹⁰⁴⁻¹⁰⁵⁾. In the airways of COPD patients, these alterations are persistent ^(100, 102)

Basal cell hyperplasia is characterized by an extensive increase of basal cells up to a multitude of layers. The superficial layer generally consists of normal differentiated, columnar cells ⁽¹⁰⁶⁻¹⁰⁷⁾. Basal cell hyperplasia may be developed as a response to the exposure to a number of toxins and is very common in the airways of smokers ⁽¹⁰⁶⁻¹⁰⁷⁾. It is thought to be reversible but might also be an early stage of squamous metaplasia ^(106, 108).

Goblet cell hyperplasia is characterized by an increased number of goblet cells. Together with the enlargement of mucous glands, which is another common feature in COPD, goblet cell hyperplasia leads to a hypersecretion of mucus and consequently to the failure of the mucociliary clearance ^(90, 109-110). Mucus hypersecretion is a major cause of airway obstruction in COPD patients ⁽¹¹¹⁻¹¹²⁾. The goblet cell hyperplasia accompanied by the enlargement of submucosal glands is also a pathologic feature in mild, moderate and severe asthma and a major cause of airway obstruction in this disease ^(90, 113-116). Moreover, the loss of epithelial integrity and shedding of the epithelium are common characteristics in asthma ^(81, 98, 109).

In airways that exhibit squamous metaplasia, the normal differentiated columnar epithelium is completely replaced by a full thickness squamous epithelium ^(107, 117). Although squamous metaplasia is a common response to inflammation and infection, it is also correlated with the severity of pulmonary obstruction in COPD patients and it is believed to be a prestage for the development of carcinoma ⁽¹⁰²⁾.

Besides alterations of the epithelial layer, remodeling after injury of the airways also extends to other components of the mucosa ^(90, 98, 109, 118). The interactions between epithelial cells, fibroblasts and the ECM play a central role in the regulation of inflammation and repair of damaged tissue ^(3-4, 81, 89). Fibroblasts, which are in a quiescent state under normal conditions, start to proliferate as a response to injury of the epithelium ⁽⁹⁰⁾. This cell type is the primary source of several ECM proteins and consequently the higher number of fibroblasts leads to an increased deposition of ECM proteins ^(81, 119). In asthma and COPD, the increased deposition of ECM proteins, like collagen or proteoglycan, is a common pathologic feature, leading to a narrowing of the affected airways ^(90, 120-121).

Fibroblasts express a variety of proteins of the matrix metalloproteinase (MMP) family. MMPs have a regulatory function in matrix remodeling and degradation ^(82, 118). Epithelial cells were also shown to produce some of these proteases, like MMP-9 and MMP-2. MMP-9

26

is elevated in the airways of COPD and asthma patients, suggesting a contributory role in inflammation processes ⁽¹²²⁾. Although MMPs are primarily matrix-degrading proteases, recent studies also indicated that they affect various processes during inflammation by influencing several cytokines ⁽¹²³⁾.

Depending on the type of injury or disease, there are numerous signaling molecules, including growth factors and pro-inflammatory cytokines, that regulate the immune response, repair and remodeling in the airway epithelium. Some of the most important and best studied signaling proteins are described in the following paragraphs. The induction or effects of some of these proteins in cultured bronchial epithelial cells were also investigated in this work (see 4.3).

The transforming growth factor- β 1 (TGF- β 1) is ubiquitously expressed by several cell types but in a latent form ^(117, 124). For active signaling, TGF- β 1 needs to be activated by dissociation from the latency associated peptide (LAP) to which it is covalently linked in its latent form. This process is highly controlled by different mechanisms. Activated TGF- β 1 is critically involved in numerous cellular processes, including differentiation, apoptosis and proliferation. Moreover, it plays a central role in tissue homeostasis, ECM deposition and in the development of several pathogenic conditions ^(117, 125-127). For example, TGF- β 1 was found to be increased in the lungs of asthma and COPD patients. It is moreover described to be a major factor in driving squamous metaplasia of airway epithelial cells *in vitro* ^(124-125, 128-131).

Interleukin-13 (IL-13) is a potent mediator for mucus secretion in the airways. This was shown *in vivo* using a mouse model *and in vitro* on cultured human airway epithelial cells ^(111, 132-135). The enhanced mucus secretion is associated with the increase of goblet cells while ciliated cells are reduced ⁽¹³²⁾. IL-13 production is elevated in asthma and is likely to mediate the mucus hypersecretion in this disease ⁽¹¹¹⁾.

Interleukin-1 β (IL-1 β) and tumor necrosis factor- α (TNF- α) are prototypical proinflammatory cytokines that act on respiratory epithelial cells ⁽¹³⁶⁾. Secreted IL-1 β and TNF- α are central to the events of inflammation and can initiate transcription of chemokine genes via the transcription factor NF- κ B (Nuclear Factor κ B) activation ⁽¹³⁷⁾. A central role of IL-1 β and TNF- α in the induction of squamous metaplasia is suggested ⁽¹⁰²⁾.

27

The cytokine interleukin-8 (IL-8), which is secreted by a variety of cell types, is primarily a chemoattractant for neutrophils and accordingly it is involved in the immune response ⁽¹³⁸⁾. In airway epithelial cells, the IL-8 expression is elevated after the exposure to a variety of environmental factors, including cigarette smoke, allergens and air pollutants ⁽¹³⁹⁾. *In vitro*, IL-8 is released as a response to the stimulation of human airway epithelial cells with inflammatory mediators, like TNF- α ⁽¹⁴⁰⁾.

Interleukin–6 (IL-6) is a pleiotropic cytokine that is involved in the adaptive immune response by contributing to the determination of the type and intensity of the response ⁽¹⁴¹⁾. Results from animal studies concerning the role of IL-6 in inflammation processes are contradictory but it was shown that its expression is elevated in patients with severe COPD and in asthmatics ⁽¹⁴¹⁾. Moreover, the IL-6 secretion was increased in cultured human bronchial epithelial cells after stimulation with TGF- β 1 ⁽¹⁴²⁾. Together with IL-1 β , TNF- α and IL-8, IL-6 is widely used as a biomarker for inflammation.

The interplay of several proteins in complex signaling pathways is critical for tissue homeostasis and for the response to inflammation. Accordingly, studying the effects of the induction of key molecules can deliver important information on the development of acute and chronic lung diseases.

3.4.3 Prevalence and incidence of airway diseases

During breathing, the airways are constantly exposed to external insults, like pathogens or particles, and consequently the respiratory system, or more specifically its lining epithelium, is extraordinarily vulnerable to diseases ⁽⁵³⁾.

In recent years, respiratory disorders, e.g. chronic obstructive pulmonary disease (COPD), asthma, tuberculosis and lung cancer were the leading causes of death worldwide. A further increase in mortality is expected in the future. In Europe, respiratory diseases rank second in terms of mortality (after cardiovascular diseases), incidence, prevalence and cost. In some countries (e.g. the UK), they are already the leading killer ⁽¹⁴³⁾.

Asthma and COPD are two of the most common chronic diseases affecting millions of people worldwide ⁽¹⁴⁴⁾. COPD, for example, was the fifth leading cause of death in high-income countries and the sixth leading cause of death in low- and middle-income countries in 2001,

as reported by the World Health Organization. Risk factors for both COPD and asthma can be categorized as host and environmental factors. Environmental factors that serve as risk factors for COPD include inhalation exposure to cigarette smoke, occupational dusts and chemicals as well as indoor and outdoor pollution ⁽¹⁴⁵⁾. Cigarette smoke is potentially the major cause for COPD (more than 90% of COPD patients are smokers) but only 15 to 20% of smokers develop this disease ^(118, 146).

COPD and asthma are complex and multifactorial diseases with many disease phenotypes existing individually or simultaneously ^(90, 144, 147). Moreover, there are several overlapping phenotypes of asthma and COPD which makes their discrimination often barely possible ^(90, 144).

The symptoms of asthma are often treatable with anti-inflammatory medications, like inhalable corticosteroids, which can also lead to a permanent improvement in the airway inflammation. However, asthma is underdiagnosed and undertreated and consequently, the number of people developing severe asthma is still increasing ^(144, 148). In contrast, COPD patients generally do not respond to anti-inflammatory medications and permanent damage of the airways is evident in this disease ⁽¹⁴⁸⁾. Moreover, COPD is a progressive disease and continued smoking worsens the symptoms and accelerates the progress of airway destruction ⁽¹⁴⁹⁾.

Improving the understanding of the diverse phenotypes of COPD and asthma and identifying early disease markers might help to decrease the number of severe cases in the future. *In vitro* models of the airway epithelium can help to unravel cellular aspects and thus contribute to the prevention and treatment of chronic airway diseases.

3.5 In vitro Models of the Airway Epithelium

Although the number of *in vitro* studies on inhalation toxicology is continuously increasing, there are still no standardized *in vitro* models of the airway epithelium. Consequently, many researchers are currently engaged in the establishment of such models.

Many criteria have to be considered when searching for a suitable cell model. Ideally, the cells should be derived from the species of interest, in order to preclude species-specific characteristics.

Depending on the question to be answered, special additional requirements have to be fulfilled. For example, to study drug absorption or particle deposition, the cell model should show an intact barrier function comparable to the airway epithelium *in vivo*, through the formation of functional tight junctions. Moreover, the development of cilia and the secretion of airway mucins are desirable to simulate the mucociliary clearance which plays a critical role in the protection of the epithelium against toxicants and pathogens and which also represents an additional barrier for drug absorption.

For studies on the repair and regeneration capacity of the airway epithelium, the presence of basal cells and/or Clara cells in the respective *in vitro* model might be required. These cell types are believed to be the progenitor cells for all other cell types in the bronchi or bronchioles, respectively ⁽¹⁵⁰⁻¹⁵¹⁾. Therefore, they potentially play a key role in the regeneration of the lung epithelium after inflammation or mechanical damage.

For high throughput screening (HTS), e.g. for large-scale toxicity testing, a cell line that is characterized by a high doubling rate and robustness is required ⁽¹⁵²⁾. Moreover, it should have a long life span and its permanent availability has to be guaranteed.

Ready-to-use cell culture systems composed of primary epithelial cells from different portions of the airway epithelium (nasal cavity, trachea or bronchus) are commercially available (EpiAirway[™], MatTek Corporation; MucilAir[™], Epithelix) ⁽¹⁵³⁻¹⁵⁴⁾. These models are fully differentiated and exhibit all important features of airway epithelial cells *in vivo*. Moreover, they are made up either of cells from healthy persons, smokers or non-smokers, or from COPD or asthma patients and thus may cover a variety of aspects of interest. A long shelf-life of these cultures is also predicted. However, due to their commercial nature, these models are less economical and sample size is severely limited ^(78, 155). A lot-to-lot variation may be assumed due to donor-specific variations and consequently the reproducibility of data is not predictable. In addition, the shipping procedure might have adverse effects on the cultures and may lead to false results.

Therefore, the generation of in-house cell cultures either from normal airway epithelial cells or permanent cell lines is advantageous.

3.5.1 Normal human bronchial epithelial cells

Primary cells that have been isolated from normal, non-cancerous tissue provide the most realistic *in vitro* models of the respective organ since, when cultivated under appropriate conditions, they are able to retain most characteristics of their *in vivo* counterparts ⁽¹⁵⁶⁾. However, these differentiation properties get lost during subcultivation within a short time period and the cell viability decreases ^(69, 157-158). Airway epithelial cells of the first three passages have been shown to be best suited as *in vitro* models ^(69, 158). To overcome limitations in the availability of primary cells in low passages, a comprehensive pool of cryopreserved cells or lung tissue is required. However, donor-specific variations might have an effect on the reproducibility of data and thus, for repeating experiments or routine testing, the use of cell lines is more favorable ^(45, 69).

3.5.2 Human continuous airway epithelial cell lines

As described in section 3.3.3, permanent (immortal) cell lines, either cancer or immortalized cells, have an extended life span and can thus be subcultivated over many passages. There are several commercial lung epithelial cell lines of the respiratory tract which are available at public cell collections, like ATCC (American Type Culture Collection) or DSMZ (Deutsche Sammlung von Mikroorganismen und Zellkulturen). However, only a few of them have been proven to provide suitable *in vitro* models of the respective part of the airway epithelium in previous studies. For others, data are unavailable or they have been found to be of no use for inhalation or drug absorption studies due to the lack of functional tight junctions or the loss of other important differentiation properties ^(5, 8). The most commonly used bronchial epithelial cell lines are the carcinoma-derived Calu-3 and the two virus-transformed cell lines 16HBE14o- and BEAS-2B ⁽⁶⁹⁾. The adenocarcinoma-derived A549 is a well-established cell line for studies on AEC type II cells.

The following paragraphs give brief descriptions of the cell lines that were used in this study.

3.5.2.1 16HBE14o-

The 16HBE14o- is an immortalized cell line derived from the 2nd generation bronchus of a one-year old male. Immortalization has been achieved by transformation with the large T antigen of the SV40 virus ⁽⁴⁶⁾. Although licensed and not commercially available, these cells have widely been used and are well-established. They show many characteristics of bronchial epithelial cells *in vivo* when cultivated under appropriate conditions ⁽⁵⁾. For example, the 16HBE14o- cells are described to form functional tight junctions and to exhibit an active ion transport. Moreover, they show a high viability over many passages. However, their ability to form cilia and to secrete mucus is controversial ^(46, 69, 157, 159).

3.5.2.2 Calu-3

Calu-3 is an adenocarcinoma-derived cell line with properties of human submucosal gland epithelia cells. The cells are commercially available from ATCC (product no. HTB-55) ⁽¹⁶⁰⁾. The Calu-3 cells form polarized monolayers and exhibit functional tight junctions and an active ion transport comparable to the bronchial epithelium *in vivo* when cultivated under appropriate culture conditions. These characteristics make them valuable, for example, for drug absorption studies ^(8, 161-162). Moreover, these cells have been reported to produce mucus and to form apical cilia, although irregularly ^(69, 157).

3.5.2.3 ChaGo-K1

ChaGo-K-1 is a hyperdiploid human cell line isolated from a bronchogenic carcinoma of a 45year-old male ⁽¹⁶³⁾. The adherent cells have an epithelial morphology and the main characteristic is the production of high levels of MUC1 mucin ⁽¹⁶⁴⁾. The cell line is available at ATCC (product no HTB-168TM).

These cells were shown to respond to the exposure to titanium oxide nanoparticles with an increased mucin secretion, the induction of ROS (reactive oxygen species) production and finally with cell death by the apoptotic pathway. Similar reactions occur in normal human airway epithelial cells *in vivo* ⁽¹⁶⁵⁻¹⁶⁶⁾.

However, there are no data available on the differentiation properties of this cell line.

3.5.2.4 A549

There are currently no well-established cell lines for *in vitro* models of the alveolar epithelium. The most prominent cell line is A549, an adenocarcinoma-derived cell line isolated from a 58-year old patient in 1972, sharing metabolic and transport characteristics of *in vivo* AEC type II cells ⁽¹⁶⁷⁻¹⁶⁸⁾. However, A549 show barrier properties distinct from *in vivo* AEC type II and they do not undergo transition into AEC type I-like cells ⁽¹⁶⁹⁾. The suitability of A549 as an alveolar absorption model is also questionable due to the fact that more than 90% of the alveolar surface is covered by AEC type I.

Their robustness and high proliferation rate, nevertheless, recommend the A549 cell line for high-throughput screening assays and studies on acute toxicity ^(27, 170-172).

3.5.3 Three-dimensional cultures and co-culture models of the airways

The "three-dimensional (3D) cell culture" is inconsistently described in the literature. In the case of cultured airway epithelial cells, the term is sometimes used to describe the fully differentiated and heterogeneous population of epithelial cells cultivated at the air-liquid interface and showing a three-dimensional (polarized), pseudostratified and columnar structure, like the ready-to-use cultures MucilAir[™] or EpiAirway[™] (see section 3.5). However, in other studies, the "3D cell culture" specifies the co-culturing of different cell types which enhances the physiological relevance of *in vitro* tissue models with regard to the complex interactions between the cell types. The latter definition is adopted in this work.

In recent years, various co-culture models simulating different parts of the respiratory tract have been developed. For example, *in vitro* models of the bronchial epithelium have been realized by co-culturing human epithelial cells (cell lines or primary cells) with human fibroblasts. Here different methods have been used. In the simplest model, bronchial epithelial cells are supplied with fibroblast-conditioned medium or vice versa ^(4, 173). In a more sophisticated approach, the two cell types are co-cultured, either in direct or indirect contact. An indirect contact is enabled by seeding the fibroblasts on the opposite side of a microporous insert membrane or in wells of an insert receiver plate, while the epithelial cells are grown on the apical side of insert membranes ^(3, 174-175). A direct contact is provided, by seeding epithelial cells onto fibroblasts, that were either grown as monolayers or embedded

into a collagen matrix ⁽¹⁷⁶⁻¹⁷⁸⁾. Thus, the natural structure of the airway epithelium is simulated.

There are also co-culture models that mimic the alveolar epithelium, for example by combining A549 cells with cells of the immune system, like macrophages alone or combined with dendritic cells in a triple co-culture system ⁽¹⁷⁹⁻¹⁸¹⁾. Another approach utilizes endothelial cells for co-cultures with lung epithelial cells, mostly A549, to simulate the airblood barrier of the alveoli ⁽¹⁸²⁻¹⁸⁴⁾.

All of the above mentioned studies consistently demonstrated that different results were obtained between co-cultures and their respective monocultures due to natural intercellular interactions. These 3D lung tissue models were shown to be promising tools for providing more realistic conditions in inhalation, toxicological and drug absorption studies *in vitro*.

3.5.4 Cultivation of airway epithelial cell cultures

In vivo, there are complex interactions of numerous physiological parameters, which are responsible for the structure and functions of the respective tissue. To obtain nearly realistic tissue models *in vitro* with well-differentiated and long-living cells, an optimal balanced culture environment, approximating the *in vivo* conditions, is essential.

The air-liquid interface (ALI) was previously shown to be indispensable for the development of an *in vivo*-like phenotype in cultured human airway epithelial cells ^(44, 158, 185-187). Here the cell cultures are supplied with nutrients only from the basal side while their apical surface is exposed to air, comparable to the conditions in the body. For that purpose, cell culture inserts are used which have a microporous membrane, mostly made of polyethylene terephthalate (PET) or polycarbonate, on which the cells are grown. The inserts are placed into wells of receiver plates which are filled with cell culture medium. The membrane pores enable the diffusion of the nutrients from the basal side to the cells (Figure 3-4). The ALI promotes the mucociliary differentiation of airway epithelium *in vitro* and is essential for the formation of TJs ⁽⁴⁴⁾.



Figure 3-4: Cultivation of airway epithelial cells on microporous membranes of commercial cell culture inserts. A: The cells are cultivated submerged in cell culture medium to enhance the proliferation; B: The air-liquid interface cultivation method is indispensible for the differentiation into ciliated and mucous cells (1= Ciliated cell; 2= Goblet cell; 3= Basal cell; 4= Mucus).

Other critical parameters for the cultivation of epithelial cells are the (1) environmental conditions (temperature, humidity), (2) nutrition composition and supply (3) and substrate (culture surface).

Environmental conditions simulating the physiological conditions of the body with regard to temperature and humidity can today easily be established in the laboratory by means of incubators that are available from several companies. These incubators additionally allow the adjustment of a physiological relevant CO_2 level which is also important for maintaining a constant pH in the cell culture medium. The pH is a critical factor and therefore most cell culture media contain buffers which are dependent on certain CO_2 levels ⁽¹⁸⁸⁾.

The airway epithelium is characterized by a diverse phenotype with several specialized cell types and functions (see section 3.4.2). Thus, the recreation of this complexity *in vitro* is challenging and requires a balanced interplay of various factors and an optimal medium composition.

Retinoic acid (RA) is an oxidized derivative and active form of vitamin A. RA is known to influence numerous processes in the cell and therefore plays an important role in cell growth and differentiation both *in vivo* and *in vitro* ^(158, 189-190). Especially in the airway epithelium, retinoic acid is crucial for mucociliary differentiation ^(158, 190-191). Vitamin A deficiency, on the other hand, leads to the development of squamous metaplasia and keratinization ^(190, 192-195). However, retinoic acid is also known to reduce or inhibit

proliferation in some cells, including human bronchial epithelial cells in concentrations higher than $1 \mu M^{(196-197)}$. Accordingly, the use of low concentrations of RA during the growth phase is advantageous while higher concentrations are required for the induction of the mucociliary differentiation in the later differentiation phase. Mostly, a concentration of 50-100 nM is used for the induction of differentiation in airway epithelial cells ⁽⁹⁾.

Calcium is known to affect cell signaling, differentiation and proliferation of certain cell types $^{(198)}$. Low concentrations of CaCl₂ were found to enhance cell proliferation but to suppress the differentiation of airway epithelial cells *in vitro*. Therefore, medium with a low CaCl₂ concentration is used during the growth phase while the concentration is later increased to promote the differentiation.

Epidermal growth factor (EGF) promotes the growth of epithelial cells but has an inhibitory effect on the normal differentiation of airway epithelial cells. High EGF concentrations were furthermore shown to induce basal or goblet cell hyperplasia in cultured airway epithelial cells ⁽¹⁹⁹⁻²⁰⁰⁾. Consequently, low concentrations are required to enhance the development of cilia and the mucus secretion ^(9, 158, 198, 201).

There are several other factors that promote the growth and differentiation of cultured airway epithelial cells, including insulin, transferrin and hydrocortisone. Optimized cell culture media that were specifically designed for the growth of airway epithelial cells are commercially available from different suppliers. Ideally, the supplements are provided in individual tubes, like the BEGM[™] Bulletkit[™] provided by Lonza. In this case, the concentrations of critical factors for the cell differentiation can be added in user-defined concentrations. Optimal concentrations of these supplements resulting in well-differentiated cultures of airway epithelial cells can be found in the literature ⁽⁹⁾.

Animal serum is commonly used as a supplement to the cell culture medium since it contains numerous hormones, growth and attachment factors that are essential for the cultivation of many cell types ⁽²⁰²⁾. The most widely used sera are derived from fetal bovine calves ^(156, 203). There are some reasons that promote the use of serum-free cell culture media. Besides the ethical concerns, serum is a potential source for microbiological contaminations, like bovine viruses or mycoplasma. Moreover, the composition of serum is highly variable and consequently differs between lots ⁽¹⁵⁶⁾. Several serum substitutes are available from different suppliers, like Ultroser[®] G (Pall Corp.) that was also used in this work. Although these serum

substitutes are also delivered as an undefined mixture, their composition is highly consistent between lots.

Epithelial cells are adherent growing cells. Their adherence to the cell culture vessel depends on the nature of the surfaces. *In vivo*, the attachment of the cells to their substrate or to each other is enabled through the ECM which is composed of a variety of proteins secreted by cells organized into a structural network to which the cell surface integrin receptors bind ⁽⁵⁸⁾. The ECM also promotes the growth and differentiation of epithelial cells *in vivo*. The extracellular matrix is mainly composed of different types of collagens. Further important components are attachment molecules, like fibronectin or laminin ⁽¹⁸⁸⁾.

In cell cultures, normal attachment, growth, and differentiation of many cell types also depend on attachment factors and other ECM components ^(188, 204). While some cells are able to synthesize these components, others require an exogenous source, particularly when grown in serum-free cultures. Thus, many cell cultures are seeded on surfaces that are coated with ECM components ^(45, 188, 198). Collagen type I is predominantly used for this purpose because it is the most common protein in the human body and it can easily be isolated, for example from rat tails or bovine skin, and is thus economically available. For the cultivation of primary airway epithelial cells, collagen type IV is also frequently used as main component of the basal membrane to which the epithelial cells are attached *in vivo* ⁽²⁰¹⁾. Many protocols recommend a mixture of collagen and fibronectin and/or laminin for surface coating.

The optimal conditions for the adherence, growth and differentiation were established individually for each cell line and cell type used in this work in a stepwise process.

3.6 Automated cultivation systems

The maintenance of cell cultures is laborious work made up of many repetitive steps that are necessary to guarantee the provision of viable cells. This routine work is vulnerable to person-specific variations which might be disadvantageous for the reproducibility of cell culture studies. Moreover, the static medium supply leads to a nutrition gradient over the cultivation time and shifts in the pH of the medium. To increase the consistency of cell culture processes and to establish standardized conditions, automated cell culture systems are increasingly employed, especially in pharmaceutical companies ⁽²⁰⁵⁾.

There are different fields of application for automated cell culture systems, each demanding the automation of particular working processes ⁽²⁰⁶⁾. For example, for high-throughput screening assays, the large-scale expansion and subcultivation of cell lines as well as the preparation of high numbers of cell cultures in multiwell plates are required. Here, fully-automated cultivation systems with integrated pipetting robots are advantageous, like the CompacT SelecT provided by TAP Biosystems ^(205, 207).

For the maintenance and expansion of cells or tissue-engineered cultures under stable conditions, bioreactors are often used. These systems allow a dynamic medium supply and enable the control of pH, oxygen and transport of metabolic waste ^(38, 208). Moreover, the stirring or agitation of the cell culture medium not only offers an optimal nutrition and oxygen supply but also induces a mechanical stimulation and thus provides more realistic environmental conditions which enhance the growth and differentiation of the cells ⁽²⁰⁹⁾. Many different types of bioreactors are available which are either suitable for suspension cultures (e.g. spinner flasks) or, in the case of anchorage dependant cells, for cell cultures on special matrices (e.g. the rotation bed bioreactor) ^(38, 209).

There are two automated cultivation systems that were specifically designed for the small scale cultivation of cell cultures grown on 3D scaffolds ⁽²¹⁰⁻²¹¹⁾. These systems are based on a modular concept, whereby the cell cultures are cultivated in individual chambers that can be connected to each other ⁽²¹¹⁾. A continuous medium supply through each chamber is provided leading to a constant provision of nutrients and, at the same time, to a mechanical stimulation of the cells. Thus, a more realistic culture environment is established ⁽²¹⁰⁻²¹¹⁾.

Despite the great advances made in the field of automated cell cultivation, there are currently no systems that enable the controlled cultivation of cells on cell culture inserts and, moreover, that are suitable for the air-liquid interface cultivation which is indispensable for the differentiation of airway epithelial cells.

3.6.1 The CULTEX[®] Long-Term Cultivation (LTC) System

The manual cultivation of cell cultures grown on insert membranes is a time-consuming task and, for the successful and reproducible establishment of fully differentiated cell layers, a multitude of factors has to be considered.

For example, the handling of a large number of cell culture inserts requires the frequent opening of the incubator which causes repeated fluctuations in temperature, humidity and CO₂ content. Moreover, the manual renewal of the cell culture medium may lead to slight deviations in the adjusted medium level which could have severe effects on the integrity of cell layers in the case of air-liquid interface cultures. In addition, a static medium supply at defined time-points leads to variations in the nutrition supply and pH throughout the cultivation period, which further depends on the metabolic rate of the respective cell type.

To overcome the numerous difficulties of the complex manual cultivation of insert cell cultures at the air-liquid interface, Cultex Laboratories GmbH developed and constructed a system that was specifically designed for the computer-controlled long-term cultivation of such cultures, the CULTEX[®] LTC.

In order to establish dynamic and stable cultivation conditions, the CULTEX[®] LTC module incorporates a computer-controlled medium supply and mixing device. The system provides a controllable intermittent or continuous individual medium supply for three insert cell cultures. The medium circulation through one insert holder is driven by two peristaltic pumps. The medium enters each vessel via the medium inlet and is sucked out through the medium outlet. At the base of the vessels, the medium passes through a perforated metal plate which enables the mixing of the medium and is involved in providing a mechanical stimulus induced by a teflon membrane beneath the metal plate. The teflon membrane can be moved up and down by a lifter that is driven by an electromagnet (Figure 3-5). The medium can thus be stirred thoroughly and, at the same time, a mechanical stimulation is provided. The medium level is controlled by an ultrasonic sensor that is able to detect level deviations of ≥ 0.1 mm (Figure 3-5). The adjustment of the medium levels, the medium renewal rate and the mixing frequency are computer-controlled by special software.

The cell culture medium and medium pumps are located outside the incubator and are connected to the CULTEX[®] LTC inside the incubator via silicone tubing and cables. This setup

allows the control of the cultivation system without interfering with the adjusted conditions of the incubator.



Figure 3-5: Overview of the CULTEX[®] LTC (Long-Term Cultivation) module (A) and a detailed description of the individual automated vessels for the cell culture inserts (B). (Illustration provided by Halter Engineering GmbH)

The integration of this technical device should optimize several processes of the cultivation of cell cultures on insert membranes at the air-liquid interface. The intermittent or continuous medium renewal and the mixing of the medium lead to a stable nutrition supply and pH. At the same time, a mechanical stimulation is provided which reflects the *in vivo* situation and may thus have a positive effect on the growth and differentiation of the cell cultures. The computer-controlled cultivation increases the maintenance and reproducibility of stable conditions by allowing the exact adjustment of the medium level and by eliminating

the necessity to open the incubator. Last but not least, such a system is a time-saving alternative to the manual handling of cell cultures cultivated on insert membranes.

3.6.2 Testing of materials for *in vitro* devices

There are many factors that have to be considered when constructing devices for *in vitro* techniques, especially when utilizing living cell or tissue cultures. For example, exposure devices should optimally be designed with special regard to providing stable and reproducible conditions and physiologically relevant exposure times and concentrations depending on the respective test agent ⁽²¹²⁻²¹³⁾. Moreover, especially for long-term exposure experiments, sterility and physiological conditions (temperature, pH) have to be guaranteed. The latter parameters are also most relevant for automated cultivation systems.

Another critical point is the use of appropriate materials. The necessity to test the materials used for the construction of *in vitro* devices to exclude any influence on the cell cultures has already been declared in the late 1950s by Giardinello and colleagues ⁽²¹⁴⁾.

The International Standard (ISO) 10993-5:2009(E) describes three test methods to assess the biocompatibility of materials *in vitro* using cell cultures. This standard was developed to meet the requirements of testing materials for medical devices which is regulated in the European Directive 93/42/EEC and its amendments ⁽²¹⁵⁻²¹⁶⁾.

The "direct contact method "provides the most realistic and efficient test procedure. Here, the material sample is placed in the culture vessel and incubated with the cell culture during the entire cultivation time under standard conditions. Potential toxic components may thus be released into the culture medium and can be detected through alterations in the cell cultures. This method generally enables the qualitative (cell morphology) and quantitative (cell viability) analysis of cytotoxicity. However, movement of the specimens might cause a physical trauma to the cells and thus hinder the quantitative interpretation ⁽²¹⁷⁾.

The "indirect contact method" typically utilizes an agar layer to separate the specimen from the cells (agar diffusion test), thus preventing a physical impact. Here, the selection of a suitable agar is crucial. A low melting temperature is required to avoid a thermal shock to the cells and the molecular weight must be appropriate to allow the diffusion of all extractable chemicals. The agar layer should be constant since the diffusion distance might affect the concentration of toxicants in the cells ⁽²¹⁷⁾. This test method allows qualitative analysis but not quantitative assessment of cytotoxicity.

In the "elution test", extracts of the test materials are prepared. For that purpose the samples are incubated in cell culture medium for a defined time ⁽²¹⁸⁾. The medium should then contain all leachables from the test material and can be used for the cultivation of the cell cultures. This method also allows the qualitative and quantitative analysis of cytotoxicity. However it is more time-intensive than the direct contact test since the extract has to be prepared prior to testing.

In this thesis, an improved variant, which combines the advantages of all three standardized test methods, is established. This procedure can be used for the rapid evaluation of biocompatibility of materials that are intended to be used for the construction of newly developed *in vitro* devices.

4 Experimental Work

The experimental work of this thesis started with the characterization and evaluation of different available human airway epithelial cell lines and commercial primary cells of the respiratory tract. Later, a protocol for the isolation of bronchial epithelial cells from human lung tissue was developed and these cells were then also included in the analyses. The aim was to find an appropriate cell model of the airway epithelium that (1) reflects important *in vivo* characteristics and (2) is suitable for routine toxicological testing.

In the next step, selected appropriate airway epithelial cell models were combined with lung fibroblasts- which mostly were embedded in a collagen matrix- in order to reflect the reciprocal interactions of the EMTU. Here different protocols for the implementation of such 3D co-culture systems were developed. The mesenchymal component was mostly represented by a common human fetal lung fibroblast strain, the IMR90, which is commercially available. However, to also include adult fibroblasts, a protocol for the isolation of fibroblasts from human lung tissue was developed.

For the analyses of the 3-dimensional growth structure and the presence of certain differentiation characteristics, several analytical methods, like histology, immunohistochemistry and immunofluorescence were applied. Since these techniques were originally developed for tissue cultures (histology and immunohistochemistry) or cells grown in a 2-dimensional manner on glass or plastic (immunofluorescence), the protocols had to be adapted specifically for cell cultures grown on microporous membranes of cell culture inserts.

The selected airway epithelial cell models were afterwards further used to study the effects of important signaling molecules that are associated with inflammation processes. The aim of these experiments was to investigate whether the *in vitro* models are capable of developing natural inflammation and remodeling features which is an important requirement for suitable cell models for toxicological studies.

In the last step, the CULTEX[®] LTC was used for the cultivation of the developed *in vitro* models, to evaluate the effects of such an automated system for cell growth and differentiation as well as for its suitability to deliver highly reproducible cell cultures.

43

4.1 Evaluation of cell lines and primary cells of the bronchial epithelium

Different permanent lung epithelial cell lines and primary cells were characterized for their ability to develop important differentiation patterns. The most important functional characteristics of the bronchial epithelium are the mucus secretion, cilia beating, barrier property (tight junctions) and self-renewal capacity (presence of basal cells).

All cultures were prepared on cell culture inserts in 12-well-format. They were cultivated under submerged conditions in their cell-specific growth medium within the first days of cultivation until the cells formed a confluent layer. Then the apical medium was removed and the cells were cultivated at the air-liquid interface with the respective differentiation medium. Most cultures were supplied with the basic differentiation medium with the following composition:

BEBM (Lonza)/ DMEM (incl. 1g/l glucose, HEPES), at a ratio of 1:1

+ BEBM *Single Quots*[®] (without retinoic acid)

- Hydrocortisone
- Insulin
- Transferrin
- Epinephrin
- Triiodothyronine
- Human epidermal growth factor (hEGF)
- Bovine pituitary extract (BPE)
- Gentamicin/amphotericin B (GA)
- + 1mM CaCl₂
- + 2% Ultroser®
- + 100 nM retinoic acid

The exact concentrations of the additives provided as SingleQuots[®] by Lonza are unknown, but they are described to be specifically adapted to the requirements of bronchial epithelial cells.

In order to analyze the cell cultures for the presence of the desired differentiation characteristics, several analytical methods were applied frequently. The growth status and the sterility of the cultures were checked by *in situ* microscopy (inverted microscope), several times a week. To learn more about the differentiation progression of the respective

cell line over the cultivation time, the different inserts representing one cell culture were taken for histology at different time points. For that purpose, the cells were fixed with 10 % formaldehyde and, after dehydration, embedded in Paraplast®. Histological sections $(5 - 7 \mu m)$ were routinely stained with hematoxylin-eosin (HE), a standard staining for the visualization of different cell components. For the detection of certain differentiation markers, immunohistochemical staining was applied. To get an overview of the distribution of special differentiation characteristics, e.g. cilia, throughout the entire culture, immunofluorescence analyses were performed sporadically on normal, non-embedded cell cultures after a short fixation with 4% formaldehyde. Table 4-1 gives an overview of the markers (antibodies) that were routinely analyzed for the detection of the most important differentiation features or pathological alterations of bronchial epithelial cells by immunohistochemical analyses occasionally by immunofluorescence. or For immunohistochemistry, the DAB (3,3'-Diaminobenzidine) chromogen was used for the detection of antibody binding and thus, positive staining signals were indicated by a brown colour (see 6.1.5.2).

Marker (antibody)	Detection of
MUC5AC	Mucin 5AC
MUC5B	Mucin 5B
Tumor protein p63	Basal cells
Acetylated alpha-tubulin	Cilia
Ki-67	Proliferating active cells
Cytokeratin 6 (CK6)	Basal cells and squamous cells
Cytokeratin 13 (CK13)	Squamous cells

Table 4-1: List of routinely used markers (antibodies) for the detection of important differentiation characteristics of bronchial epithelial cells by immunohistochemistry or immunofluorescence

The formation of functional tight junctions was determined by measuring the TEER [transepithelial electrical resistance (Ω / cm²)] of the cultures. The higher the TEER values the tighter were the intercellular connections through the development of tight junctions.

Protocols and further information on the analytical methods as well as the cultivation procedures for each cell line are given in chapter 6 (Appendix).

The results of the analyses were used for evaluating the suitability of the respective cell line to serve as a *in vitro* model of the bronchial epithelium.

4.1.1 Continuous epithelial lung cell lines

4.1.1.1 ChaGo-K1

The ChaGo-K1 is a cancer-derived cell line which is not well characterized. Since no data on the differentiation properties of these cells are available in the literature, the ChaGo-K1 was included for evaluation in this thesis.

The major benefits that recommend this cell line are its permanent availability (deposited in a public cell collection) and the extraordinarily high growth rate and cell viability over many passages (at least up to 20 passages after arrival from ATCC). Moreover, ChaGo-K1 cells grow on uncoated surfaces and do not demand any special medium additives. These characteristics are highly advantageous for routine and high-throughput testing in toxicology.

The cells were cultivated in cell culture flasks with a growth area of 75 cm² and with RPMI 1640 supplemented with 10 % FBS. They showed a high metabolic activity which was indicated by a rapid acidification of the culture medium visible through the color change of the phenol red indicator to yellow.

For further characterization, the ChaGo-K1 were seeded onto microporous insert membranes with 0.4 or 1.0 μ m pores that were either uncoated or coated with collagen type I or IV (see 6.1.1.4). The varying coatings or pore sizes of the membrane had no obvious effects on the growth and differentiation of the cells. The ChaGo-K1 were seeded at a density of 2 × 10⁵ cells/ cm² and reached confluency after two days under submerged conditions. The cultures were then cultivated at the air-liquid interface and supplied with the basic differentiation medium.

It soon became obvious that the ChaGo-K1 cell line does not provide a suitable *in vitro* model of bronchial epithelial cells. *In situ* microscopy initially suggested that the roundish ChaGo-K1 cells formed a dense monolayer and a remarkable amount of a mucus-like substance was detectable on top of the cultures already after eight days which increased in correlation with the culture age. However, the TEER values were around zero (after subtraction of the blank value), indicating a lack of functional tight junctions. This assumption was further verified by histology. The histological sections revealed that the cells formed a multilayered network of individual cells without intercellular connections. The basal layer was tightly adhered to the membrane, whereas the upper layers became detached during handling. This growth pattern was observed in all ChaGo-K1 cell cultures (Figure 4-1).



Figure 4-1: Growth pattern of ChaGo-K1 cells cultivated on collagen IV coated microporous membranes (0.4 μm pores) after a total cultivation time of 14 days at the air-liquid interface with basic differentiation medium (HE stained histological section; magnification: 200x).

The cells were small in size and had a high nuclear to cytoplasmic ratio (N/C >1; Figure 4-2). These characteristics are mostly found in cells of small cell carcinoma $^{(219)}$.

Immunohistochemical analyses with antibodies against the major airway mucins, MUC5AC and MUC5B, showed no positive signals and cilia-like structures were also not detected in ChaGo-K1 cultures.

Conclusions

In summary, the ChaGo-K1 cells were lacking most of the important properties of normal bronchial epithelial cells and thus do not provide a realistic *in vitro* model of the airway

epithelium. No further investigations on the metabolism of these cells were performed during this thesis. Due to the fact that they failed to form tight junctions, it is not likely that the cultures might reflect properties of a functional epithelial barrier.



Figure 4-2: Morphology of ChaGo-K1 cells cultivated on collagen IV coated microporous membranes (0.4 μ m pores) at the air-liquid interface with basic differentiation medium after a cultivation time of 20 days (HE stained histological section; magnification: 630x).

4.1.1.2 Calu-3

A second cancer cell line was subsequently tested - the Calu-3. In contrast to the ChaGo-K1, Calu-3 cells are well-established and characterized. Previous studies have proven their suitability to study aspects of ion transport, drug transport and metabolism and the functions of tight junctions in the airway epithelium ⁽²²⁰⁻²²²⁾. Unlike many other cancerderived cell lines, Calu-3 is described to be heterogeneous and therefore might represent different cell types of the airway epithelium when cultivated under appropriate conditions ⁽²²³⁾. Their permanent availability and high cell viability over many passages recommend the Calu-3 cells for routine toxicological testing.

The cells were cultivated in Eagle's Minimum Essential Medium (EMEM), supplemented with 10 % FBS and in uncoated culture flasks with a growth area of 75 cm². Their growth pattern reflects the cancerous origin of the cells. In contrast to primary cells (see 4.1.2.3), the Calu-3

show a patchy growth pattern. Single cell clusters fuse during the exponential growth phase and finally form a dense cell layer (Figure 4-3).



Figure 4-3: Growth pattern of Calu-3 cells cultivated in cell culture flasks (75 $\rm cm^2)$ with EMEM including 10 % FBS

Interestingly, although derived from tumors, these cells have a relatively low growth rate, especially when compared to the ChaGo-K1. Therefore, the subcultivation of the Calu-3 cells was only performed once a week. However, due to their dense growth, the cell yield obtained by passaging the cells of one culture flask (75 cm²) with a confluency of approximately 80% was extraordinarily high (up to 1.5×10^7 cells/ 75 cm²).

When the cells were seeded onto collagen type I coated microporous insert membranes (with 0.4 μ m pores) at a density of 2.5 × 10⁵ cells/ cm², a confluent cell layer was detectable after five to seven days under submerged conditions. After establishing ALI conditions and feeding the cells with basic differentiation medium, they soon formed monolayers with columnar-shaped cells and exhibiting high TEER values between 1330 and 3300 Ω /cm², depending on the culture age and amount of mucus secretion. These high TEER values clearly indicate the presence of functional tight junctions. The morphology of Calu-3 cells under these conditions was highly similar to normal bronchial epithelial cells but cilia were lacking (Figure 4-4).



Figure 4-4: Morphology and growth pattern of Calu-3 cells cultivated on collagen I coated microporous insert membranes (0.4 μ m) at the air-liquid interface with basic differentiation medium after a cultivation time of 31 days (25 days under ALI conditions; HE stained histological section; magnification: 400x).

Secretion of a mucus-like substance was detectable after three to four days under ALI conditions which increased with culture age. The secretion activity was higher when the Calu-3 were cultivated on membranes with 1.0 μ m pores but the life span of these cultures was reduced compared to those that were cultivated on membranes with 0.4 μ m pores. Accordingly, the inserts with 0.4 μ m pores were constantly used for the cultivation of Calu-3 cells. Under these conditions, the cultures had a life span of about 28 - 35 days and, in a few cases, of up to 50 days before drastic changes in their cell morphology (e.g. large vacuoles) were detectable by *in situ* microscopy and histology (Figure 4-5).



Figure 4-5: Morphological changes of Calu-3 cells grown on microporous membranes (collagen I coated; 0.4 μ m pores) after a total cultivation time of 63 days (56 days under ALI conditions) with basic differentiation medium (HE stained histological section; magnification: 630x).

The secretion of the main airway mucins MUC5AC and MUC5B was verified in Calu-3 cultures by immunohistochemical analyses, although positive signals were sparse (Figure 4-6). However, the secreted products are normally distributed on top of the cell layer

and are mostly washed away during the fixation and dehydration of the cultures prior to the paraffin embedding. Therefore, it is likely that the secretion of MUC5AC and MUC5B was much higher than the immunohistochemical staining suggested.



Figure 4-6: Immunohistochemical staining of the mucins 5AC (A) and 5B (B) in a Calu-3 culture cultivated on collagen I coated microporous insert membranes (0.4 μm) at the air-liquid interface with basic differentiation medium after a cultivation time of 27 days (21 days under ALI conditions). The cell nuclei were counterstained with hematoxylin (Magnification: 400x).

Basal cells were lacking in all cultures, which was verified by the negative immunohistochemical staining with the antibody against the tumor protein p63 which is a marker for basal cells in pseudostratified epithelia, like the bronchial epithelium (Figure 4-7) (224)



Figure 4-7: Immunohistochemical staining of the basal cell marker p63 in a Calu-3 culture cultivated on collagen I coated microporous insert membranes (0.4 μ m) at the air-liquid interface with basic differentiation medium after a cultivation time of 31 days (25 days under ALI conditions). The cell nuclei were counterstained with hematoxylin (Magnification: 1000x).

Cilia-like structures were detected neither *by in situ* microscopy nor in the histological sections. The acetylated α -tubulin which was used as cilia marker in this thesis was strongly expressed throughout the entire cells but not specifically at their apical side, which further indicates a lack of cilia (Figure 4-8). The acetylated α -tubulin is not exclusively expressed in cilia but it is also a component of the microtubules in mammalian cells. In Calu-3 cultures, the expression of this cytoskeletal protein was found to be considerable high.



Figure 4-8: Immunohistochemical staining of the cilia marker acetylated α -tubulin in a Calu-3 culture cultivated on collagen I coated microporous insert membranes (0.4 μ m) at the air-liquid interface with basic differentiation medium after a cultivation time of 50 days (43 days under ALI conditions). The cell nuclei were counterstained with hematoxylin (Magnification: 630x).

Most remarkably, the variations between different inserts belonging to one cell culture (cultivated under identical conditions), were very low in the case of Calu-3 cells. The growth
pattern, cell morphology, life span, amount of mucus secretion and even the TEER values were highly similar within Calu-3 cultures.

Conclusions

The preliminary characterization of the Calu-3 cell line revealed that these cells might provide a good *in vitro* model of the bronchial epithelium for toxicological studies. Besides their similar morphology, they also showed many properties of differentiated bronchial epithelial cells, like the secretion of the two predominant airway mucins, MUC5AC and MUC5B. Moreover, the Calu-3 cultures were characterized by high TEER values which indicated the formation of functional tight junctions. Their high viability over many passages and the superior cell yield as well as the consistency between the different inserts of one Calu-3 culture and thus, the high reproducibility, recommend these cells for routine testing of substances or high-throughput screening.

However, on average, the TEER values of the Calu-3 cultures exceeded those observed in NHBE cultures (see 4.1.2.4). Therefore, it is likely that Calu-3 cultures might be less sensitive to certain agents and show barrier properties distinct from the airway epithelium *in vivo*. The lack of cilia is another factor that has to be considered when extrapolating the *in vitro* results obtained from toxicological studies with Calu-3 cultures to the *in vivo* situation. Moreover, no basal cells are present in Calu-3 cultures. The implications of the lack of basal cells on the regeneration capacity of the cultures after injury were not analyzed during this thesis.

Nevertheless, due to the numerous beneficial characteristics, the Calu-3 cells were further used in this thesis for the development of 3D co-culture models.

4.1.1.3 16HBE14o-

The 16HBE14o- is a well-established permanent cell line. In contrast to the ChaGo-K1 and Calu-3, the 16HBE14o- is not derived from a tumor but was established from normal human bronchial epithelial cells by immortalization through transformation of the large T antigen of the SV40 virus ⁽⁴⁶⁾.

There is much information available on the optimal cultivation and the differentiation ability of these cells ⁽²²⁵⁾. However, the development of certain differentiation characteristics, like

cilia or the mucus secretion, are controversially described in the literature ^(46, 69, 157, 174). It is also not clear under which conditions the 16HBE14o- are able to form functional tight junctions ⁽²²⁶⁾.

The cells are not publicly available and their use is licensed. For this thesis, the 16HBE14ocell line was obtained from Professor Gruenert (Cardiovascular Research Institute, University of California, San Francisco) via a Material Transfer Agreement (MTA) for a half-year test period.

The 16HBE14o- cells are grown on surfaces that are coated with a mixture of collagen and fibronectin (see section 6.1.1.4). They were supplied with Minimum Essential Medium (MEM) supplemented with 10 % FBS and cultivated in 75 cm² culture flasks. The 16HBE14o-showed a patchy growth similar to Calu-3 cells but with lower density (Figure 4-9). The growth rate was relatively high compared to the Calu-3 cells and therefore the subcultivation could be performed twice a week, resulting in a high cell yield. The cell viability was constant over many passages (at least up to 30 passages after arrival from the University of California).



Figure 4-9: Growth pattern of 16HBE14o- cells cultivated in cell culture flasks (75 $\rm cm^2)$ with MEM including 10% FBS

For further characterization, the cells were seeded on collagen/fibronectin coated insert membranes (0.4 μ m pores) at a density of 2.5 × 10⁵ cells/ cm². A confluent cell layer was

observed after approximately three to four days under submerged conditions. The cultures were then cultivated at the air-liquid interface and supplied with basic differentiation medium. Histological sections revealed that, under these conditions, the 16HBE140- did not form polarized monolayers but grew in a multilayered manner. The cell morphology was distinct from the columnar-shape of normal human bronchial epithelial cells (Figure 4-10).



Figure 4-10: Morphology of 16HBE14o- cells cultivated on collagen/fibronectin coated microporous insert membranes (0.4 μ m) at the air-liquid interface with basic differentiation medium after a cultivation time of 21 days (14 days under ALI conditions; HE-stained histological section; magnification: 400x).

The 16HBE14o- cultures were further characterized by an extensive secretion of a highly viscous substance whose amount increased with culture age and which often led to a reduced stability of the cultures. The secreted material accumulated between the cells and thus the intercellular contacts were obviously disrupted. In many cases, the upper cell layers became detached from the basal layer while the latter remained attached to the membrane (Figure 4-11).

When the apical surface of the cultures was rinsed with PBS once or twice a week, the secretory material was reduced and the culture stability and life span could thus be increased (Figure 4-12).



Figure 4-11: Increase of secretory material of 16HBE14o- cells cultivated on collagen/fibronectin coated microporous insert membranes (0.4 μ m) at the air-liquid interface with basic differentiation medium after a cultivation time of 31 days (24 days under ALI conditions; HE-stained histological section; magnification: 400x).



Figure 4-12: Increased stability and life span of 16HBE14o- cultures after rinsing with PBS once or twice a week. The cultures were cultivated on collagen/fibronectin coated microporous insert membranes (0.4 μ m) at the air-liquid interface with basic differentiation medium after a cultivation time of 56 days (50 days under ALI conditions; HE stained histological section; magnification: 400x).

Interestingly, although a large amount of secreted substances was detected in all 16HBE14ocultures, immunohistochemical analyses revealed a sparse staining for MUC5AC and a lack of MUC5B, indicating that the main component of the secretory product of 16HBE14o- cells is not represented by these two prominent airway mucins (Figure 4-13).



Figure 4-13: Immunohistochemical staining of the major secreted airway mucins MUC5AC (A) and MUC5B (B) in a 16HBE14o- culture cultivated on collagen/fibronectin coated microporous insert membranes (0.4 μ m) at the air-liquid interface with basic differentiation medium after a cultivation time of 56 days (51 days under ALI conditions). The cell nuclei were counterstained with hematoxylin (Magnification: 400x).

The secreted material was also found to be barely soluble in PBS or water, which is not typical for secreted mucins of the airway mucus. *In vivo*, the mucus layer of the airways is viscoelastic and has a water content of about 95% in which the other components are dissolved (see section 3.4.2.1). The presence of large amounts of highly viscous and water-insoluble substances, like mucin 2 (MUC2), would greatly impair the mucociliary clearance ⁽²²⁷⁾. No further characterization of the secreted products of 16HBE14o- cells was performed in this work. However, it is likely that this secreted material severely limits the suitability of this cell line for toxicological studies since it might act as a protection shield and thus presumably reduces the sensitivity of the cells.

Although the 16HBE14o- cells were initially described to be able to develop cilia ⁽⁴⁶⁾, a recent study refuted this assumption ⁽¹⁷⁴⁾. During this thesis, cilia were detected neither by *in situ* microscopy nor by immunohistochemical analyses of the acetylated α -tubulin marker (Figure 4-14).



Figure 4-14: Immunohistochemical staining of the cilia marker acetylated α -tubulin in a 16HBE14O- culture cultivated on collagen/fibronectin coated microporous insert membranes (0.4 μ m) at the air-liquid interface with basic differentiation medium after a cultivation time of 56 days (51 days under ALI conditions). The cell nuclei were counterstained with hematoxylin (Magnification: 400x).

In the differentiated bronchial epithelium, the proliferative activity is restricted to the basal cells whereas the ciliated and secretory cells rest in a quiescent state. Immunofluorescence staining of the proliferation marker Ki-67 on histological sections of a 16HBE140- culture after a cultivation time of 37 days revealed that proliferating cells are spread over all layers of the multilayered culture, which further reflects the unordered growth and a lack of functional structuring within the cultures.



Figure 4-15: Immunofluorescence analyses of the proliferation marker Ki-67 on a histological section of a 16HBE14o- culture cultivated on collagen/fibronectin coated microporous insert membranes (0.4 μ m) at the air-liquid interface with basic differentiation medium after a cultivation time of 37 days (32 days under ALI conditions). The Ki-67 staining was visualized through a TRITC-labelled secondary antibody (red; A). The cell nuclei were stained with DAPI (blue; B). Signals of both stainings were merged to obtain an overview of the ratio of proliferating, Ki-67 positive cells (purple; C) to the total number of cells.

The ability of the 16HBE14o- cells to form functional tight junctions is inconsistently described in the literature. Erhardt et al. (2002) suggested that the development of TJs in these cells might depend on the cultivation conditions. According to this study, 16HBE14o- cells only develop TJs under submerged conditions, but not when cultivated at the air-liquid interface ⁽²²⁶⁾. To verify this assumption in this thesis, the TEER value of six 16HBE14o- insert cultures were frequently monitored over a time period of 18 days. This was enabled by the use of the Millicell ERS-2 Volt-Ohm-meter (Millipore) which provides a noninvasive technique for the measurement of TEER values (see section 6.1.5.4). For the cultivation of the test cultures, the regular cultivation cycle was applied, meaning that the cells were grown under submerged conditions with their cell-specific growth medium (MEM incl. 10 % FBS) until they formed a confluent layer, which was reached after eight days. Starting from

day nine, the cultures were cultivated at the air-liquid interface. To further study the influence of the medium composition on the formation of TJs, the growth medium was not replaced by the basic differentiation medium until the cells were cultivated for three days under ALI conditions. Accordingly, the cell cultures were supplied with the growth medium during the first three days under ALI conditions. Three of the six insert cultures were not adapted to the differentiation medium afterwards, but were treated with 200 μ l of a 0.5 % Triton X-100 solution for 10 min at room temperature. Triton X-100 is known to rapidly induce apoptosis. The TEER values of these cultures were determined 2 h after the addition of Triton X-100 and they were used as blank values.

The results revealed that the TEER values in all six insert cultures were highest between days seven and eight. At this time point, the cells nearly reached 100 % confluency and were still cultivated under submerged conditions. The highest TEER values were between 580 and 800 Ω / cm² which is similar to those observed in NHBE cells (see 4.1.2.4). However, as soon as the cultures were adapted to the ALI conditions, the TEER values dropped immediately to low values around 100 Ω / cm² which can be regarded as no transepithelial electrical resistance. The adaptation to the differentiation medium at day eleven had no effect on the TEER values (Figure 4-16).



Figure 4-16: Analysis of tight junction formation in six 16HBE14o- insert cultures by monitoring the TEER values over a culture period of 11 (A) and 18 days (B). The cells were cultivated under submerged conditions with cell-specific growth medium (MEM + 10 % FBS) until they formed a confluent layer after 8 days. Starting from day nine, the cultures were cultivated at the air-liquid interface but still supplied with the growth medium until day 11; A: At day 11, three 16HBE14o- insert cultures were treated with Triton X-100 to induce apoptosis. The TEER values were determined 2 h after the addition of Triton X-100 and these values were considered as blank values. B: The other three insert cultures were adapted to the basic differentiation medium (BEBM/ DMEM (1:1) including SingleQuots[®] (w/o retinoic acid), 1 mM CaCl₂, 2 % Ultroser[®] G and 100 nM retinoic acid) at day 11 and monitoring of the TEER was continued until day 18.

This experiment clearly demonstrated that the 16HBE14o- cells are able to form functional tight junctions but only when cultivated under submerged conditions. This type of cultivation, however, prevents the cell differentiation of airway epithelial cells.

Conclusions

Although the 16HBE14o- cell line has been widely used for studying aspects of drug absorption, ion transport and toxicological effects of substances in airway epithelial cells, the data presented does not support its suitability to serve as a realistic *in vitro* model of the airway epithelium. The stratified growth pattern as well as the lack of TJs under ALI conditions does not reflect the integrity of the airway epithelial barrier *in vivo*. Moreover, the highly viscous secreted material with its large amount of water-insoluble components does not resemble the properties of the airway mucus and might provide a protection shield for the 16HBE14o- which possibly reduces the sensitivity of these cells to certain agents and stimuli. Moreover, the restricted use and high licensing costs do not recommend this cell line as a universal *in vitro* model for toxicology.

4.1.1.4 Uncharacterized immortalized clonal cell lines

In the next step, a set of nine immortalized bronchial epithelial cell lines was analyzed with special regard to the cell viability over many passages and the ability to develop natural differentiation characteristics. These clonal cell lines have been immortalized by the German biotechnology company Sirion Biotech GmbH in Munich (Germany) in order to test the efficiency of the respective technique for the generation of permanent cell lines.

The company specializes in the genetic modification of mammalian cells, including the establishment of immortalized cell lines. Sirion Biotech recently transfected NHBE cells with a lentiviral vector expressing the Human papillomavirus oncogene E6/E7 (HPV E6/E7). Transformation with this oncogene was often successfully applied to immortalize human keratinocytes ⁽²²⁸⁾.

The immortalized cell lines derived from NHBE cells were established by Sirion as follows:

The original NHBE cells were obtained from Lonza as frozen stocks in passage 2. Two different lots were used for immortalization. After thawing, the cells underwent two additional subculture cycles before the transfection was performed. After the transfection, the lentivirus-transduced cells with effective integration of the transgene were selected for stable gene expression by antibiotic selection using G418 (Geniticin). This selection is possible through the presence of the geneticin resistance gene which was included in the

lentiviral vector. The expression of the integrated transgene was further confirmed by qRT-PCR analyses. From this stable cell pool, a single cell cloning was performed by means of the limiting solution method in 96-well plates. The clonality of each cell line was verified using the Clone Select Imager (Genetix). The clonal cell lines were then expanded and characterized for cell viability, morphology and the expression of the Clara cell protein 10 (CC10). The latter analysis was performed due to the fact that CC10 is a specific marker for Clara cells which are the progenitor cells in the small airways and are thus able to differentiate into ciliated as well as secretory cells. Accordingly, the expression of high amounts of CC10 might indicate a Clara cell-specific character of the respective cell line and thus the potential to differentiate into other cell types of the airway epithelium.

As a final step, frozen stocks of each clonal cell line were prepared and stored in liquid nitrogen until use. The ability of these cell lines to develop natural differentiation characteristics, like cilia formation or mucus secretion, was not tested by Sirion. Therefore, during this thesis the cells were characterized with regard to their cell viability and the development of certain differentiation characteristics.

After thawing, the cells were transferred to cell culture flasks with a growth area of 25 cm² and cultivated in BEGM (BEBM + SingleQuots[®]; Lonza) including 40 μ g/mL G418). The addition of G418 is essential to specifically expand cells with a stable expression of the transgene. The growth pattern and cell morphology of the immortalized cells was highly similar to normal (non-transformed) bronchial epithelial cells (Figure 4-17). The cells were subcultivated when they reached approximately 70 - 90 % confluency according to the same procedure as the NHBE cells. At least one vial of each cell line was prepared for storage in liquid nitrogen in the next passage after thawing.



Figure 4-17: Growth pattern of immortalized NHBE cells of clone 1 (7F4167), passage 7, cultivated in cell culture flasks (25 cm^2) with BEGM including G418

The cell clones showed a high variability in their cell viability and proliferation activity. Out of the nine cell lines tested, only five could successfully be cultivated on cell culture inserts. The others did not form a confluent monolayer. Two cell lines (of different donors) showed a high viability over many passages, while the viability of the others greatly decreased with increasing passage number. When grown on cell culture inserts (under identical conditions as NHBE cells, 4.1.2.4), the cells of all five cell clones were lacking many differentiation characteristics of normal bronchial epithelial cells, like the development of cilia or the pseudostratified growth. A similar stratified structure was observed in all cultures but the cell morphology differed between the cell lines (Figure 4-18).



Figure 4-18: Inverse microscopic images (A+C) and HE stained histological sections (B+D) of cultures of two clonal cell lines. The cell morphology of the two clonal cell lines differs while the stratified growth is similar. A+B: Monocultures of clone 1 (P07) after a cultivation time of 11 (A) or 8 days, respectively (B). C+D: Monocultures of clone 13 (P07) after a cultivation time of 12 days (Magnification B+D: 400x)

Secretion was detected in some cultures by inverse microscopy, but no mucin 5AC was verified in immunohistochemical analyses, indicating a lack of goblet cells. All cultures were characterized by a high number of p63-positive cells, but obviously these cells did not function as airway epithelial basal cells since they did not differentiate into ciliated and mucus-producing cells (Figure 4-19).



Figure 4-19: Immunohistochemical staining of the basal cell marker p63 in a culture of the clonal cell line 7 (99498) cultivated on collagen IV coated microporous insert membranes (0.4 μ m) under ALI conditions with basic differentiation medium after a cultivation time of 29 days (23 days under ALI conditions). The cell nuclei were counterstained with hematoxylin (Magnification: 630x).

The TEER measurement revealed values similar to those observed in NHBE cells, indicating the presence of a barrier function (Table 4-2). However, TEER values of stratified and pseudostratified cultures are hardly comparable.

Cell line (donor)	Passage	TEER (Ω/ cm²)	Culture age (days)
Clone 3 (99498)	7	1020	21
Clone 3 (99498)	9	1209	8
Clone 7 (99498)	7	790	12
Clone 7 (99498)	7	820	12
Clone 23 (7F4167)	7	1010	8
Clone 13 (7F4167)	7	1137	14
Clone 13 (7F4167)	7	1505	19

Table 4-2: TEER values of cultures prepared from different immortalized clonal cell lines dependent on passage number and culture age.

Conclusions

By testing these immortalized clonal cell lines, the suitability of the immortalization induced by the HPV E6/E7 transduction for human bronchial epithelial cells was evaluated.

The results demonstrated that the technique was successful for immortalizing NHBE cells, since at least two out of nine cell clones showed an increased life span with a high cell viability over at least up to 20 passages. In 2D cultures (cultivated in culture flasks), the cells showed a normal epithelial morphology and a proliferation rate comparable to NHBE cells. The fact that no mucociliary differentiation was achieved under ALI conditions is not necessarily related to the immortalization technique. There are some other possible reasons for the lack of normal differentiation characteristics. For example, as discussed below (4.1.2.4), cryopreservation can potentially cause adverse effects in non-transformed human bronchial epithelial cells. The cells used for the immortalization were obtained as cryopreserved vials from Lonza. Consequently, it is conceivable that these cells had already lost their differentiation ability before immortalization due to the cryopreservation and/or shipping procedure. Moreover, commercial primary cells are generally obtained in passage 2

and after arrival they are further cultivated in cell culture flasks until the next subcultivation. As stated below in section 4.1.2.4, NHBE cells of the first and second passage provide the best source with regard to the differentiation ability.

The data presented here suggest that the immortalization with the HPV E6/E7 oncogene might be a suitable technique for NHBE cells, but the use of freshly isolated NHBE cells of passage 1 is strongly recommended. Moreover, the cells should be tested prior to the immortalization for their ability to develop the most important differentiation characteristics.

4.1.2 Normal human bronchial epithelial cell lines

The characterization of several permanent cell lines, either cancer-derived or immortalized, revealed that they all have distinct advantages and disadvantages. However, all tested cell lines lack certain important characteristics of differentiated normal human bronchial epithelial cells.

Therefore, in the next step, normal (non-transformed) human bronchial epithelial cells were tested for their suitability to serve as appropriate *in vitro* models of the airway epithelium.

4.1.2.1 Commercial normal human bronchial epithelial cells (HBEpC)

The testing of normal human bronchial epithelial cells was initially performed with commercial cells which can be obtained from different providers. Depending on availability, cells from donors of different ages and with different medical backgrounds as well as from smokers or non-smokers can be selected. In this thesis, two different lots of these cells were purchased from Promocell GmbH:

1. Lot 9082701.10, passage 2 (HBEpC908); Donor: Male, 68 years, Caucasian, nonsmoker

2. Lot 8121902.18, passage 2 (HBEpC812); Donor: Male, 21 years, Caucasian, smoker

The cells were obtained in proliferating state (not cryopreserved) in culture flasks and the first subcultivation was performed 24 h after arrival. They were then cultivated in cell culture flasks with a growth area of 25 cm^2 and supplied with the recommended serum- and

antibiotic-free culture medium provided by Promocell (Airway Epithelial Cell Growth medium).

In order to guarantee the availability of these cells for future studies, a comprehensive cell pool of both lots of the passages three to five was prepared for storage in liquid nitrogen. Due to the limited number of cells, no experiments were performed with fresh cells before cryopreservation.

For evaluation of the cell viability after cryopreservation, one vial each of frozen cells of passage three was thawed and subcultivated over several passages in 25 cm² cell culture flasks until they became senescent. The cell seeding density of each passage was $3.5 \times 10^{5}/25$ cm². The number of viable cells was determined with the CASY Cell Counter and Analyzer System (see 6.1.1.2).

After two subcultivation cycles, it became obvious that the two lots differed in their growth rates when cultivated in cell culture flasks. The cells of the lot 9082701.10 (HBEpC908) showed a higher viability and reached confluency earlier than the HBEpC812. The HBEpC908 showed no decrease in cell viability until passage nine. But in passage ten, the growth rate and cell viability was considerably reduced and morphological alterations were observed (Figure 4-20). The majority of cells showed an increased size and some cells exhibited long extensions indicating the beginning of the senescent state (Figure 4-21).



Figure 4-20: Test of cell viability of HBEpC908 in different passages. The cells were cryopreserved in passage 5 and, after thawing, subcultivated until passage 10. The cells were cultivated in 25 cm² cell culture flasks at a seeding density of $3.5 \times 10^5/$ 25 cm². Subcultivation was performed every 7 days and the number of viable cells was determined using the CASY Cell Counter + Analyzer System.



Figure 4-21: Altered morphology of HBEpC908 cells of passage 10, cultivated in a 25cm² culture flask and supplied with Airway Epithelial Cell Growth Medium (Promocell). The cells were frozen in liquid nitrogen in passage three, thawed after several weeks and subcultivated for another 7 passages.

For further characterization, the cells of both lots were seeded on collagen type IV- coated insert membranes (0.4 μ m pores). For that purpose, cells of the passage three were used. The cell seeding density was found to be a critical factor for HBEp cells in order to form a confluent layer or a natural pseudostratified structure. The optimal seeding density was determined by trial and error and was found to be around 1.0 - 1.5 × 10⁵/ cm². It was important that the cells reached confluency of nearly 100 % before starting the ALI cultivation, since the life span of the cultures was otherwise dramatically reduced and no cell differentiation was achieved. Similar negative effects were observed when the growth phase under submerged conditions was too long (beyond 100 % confluency). In the latter case, the cells already started to detach and a confluent layer was thus not achieved.

Due to the fact that the recommended growth medium of these cells is serum-free, the omission of the serum substitute Ultroser[®] G, which was normally added to the basic differentiation medium, was initially tested. Under these conditions, the cells of both lots showed a stratified growth and an undifferentiated squamous morphology (Figure 4-22).



Figure 4-22:Growth pattern and cell morphology of HBEpC812 (A) and HBEpC901 (B) cultures cultivated on collagen IV coated microporous membranes (0.4 μ m pores) at the air-liquid interface with basic differentiation medium without Ultroser® G after a cultivation time of 13 (A) or 28 (B) days. (HE stained histological sections; magnification: 400x).

Immunohistochemical analyses with an antibody against cytokeratin 6 (CK 6) revealed an extremely strong signal in every cell throughout the cultures (Figure 4-23). CK6 is a marker for squamous cells and also strongly expressed in basal cells but not expressed in differentiated columnar cells of the pseudostratified airway epithelium ⁽²²⁹⁾.



Figure 4-23: Immunohistochemical staining of the squamous cell marker cytokeratin 6 (CK6) in a HBEpC812 culture cultivated on collagen IV coated microporous insert membranes (0.4 μ m) at the air-liquid interface with basic differentiation medium without Ultroser® G after a cultivation time of 13 days (9 days under ALI conditions). The cell nuclei were counterstained with hematoxylin (Magnification: 400x).

Since none of the cultures showed natural differentiation characteristics under these initial conditions, in the next step Ultroser[®] G was added to the differentiation medium with a final concentration of 2 %. The addition of this serum substitution agent was shown to greatly enhance the differentiation capacity of the HBEpC908 cells. Histological sections revealed a pseudostratified structure with columnar-shaped cells in some of the HBEpC908 cultures (Figure 4-24). Moreover, by means of immunohistochemical analyses the presence of basal cells was detected (Figure 4-25).



Figure 4-24: Pseudostratified structure of a HBEpC901 culture cultivated on collagen IV coated microporous membranes (0.4 μ m pores) at the air-liquid interface with basic differentiation medium including 2 % Ultroser® G after a cultivation time of 54 days. (49 days under ALI conditions; HE stained histological sections; magnification: 1000x).



Figure 4-25: Immunohistochemical staining of the basal cell marker p63 in a HBEpC908 culture cultivated on collagen IV coated microporous insert membranes (0.4 μ m) at the air-liquid interface with basic differentiation medium including 2 % Ultroser® G after a cultivation time of 54 days (49 days under ALI conditions). The cell nuclei were counterstained with hematoxylin (Magnification: 400x).

By *in situ* microscopy, strong cilia beating and secretion activity were furthermore observed in these HBEpC908 cultures after a cultivation time of approximately 28 days. The presence of cilia was verified with the acetylated α -tubulin marker in immunohistochemical analyses (Figure 4-27). Moreover, the secretion of MUC5AC was also verified and the immunohistochemical staining of this marker revealed that the MUC5AC positive cells shared the characteristic morphology of the secretory goblet cells of the bronchial epithelium (Figure 4-27).



Figure 4-26: Immunohistochemical staining of acetylated α -tubulin in a HBEpC908 culture cultivated on collagen IV coated microporous insert membranes (0.4 µm) at the air-liquid interface with basic differentiation medium including 2 % Ultroser® G after a cultivation time of 54 days (49 days under ALI conditions). The cell nuclei were counterstained with hematoxylin (Magnification: 400x).



Figure 4-27: Immunohistochemical staining of the mucin 5AC in a HBEpC908 culture cultivated on collagen IV coated microporous insert membranes (0.4 μ m) at the airliquid interface with basic differentiation medium including 2 % Ultroser® G after a cultivation time of 54 days (49 days under ALI conditions). The cell nuclei were counterstained with hematoxylin (Magnification: 630x).

However, above mentioned differentiation properties were only observed in a few inserts of the HBEpC908 cultures while others showed a similar flattened morphology and stratified growth as the cultures that were cultivated without Ultroser[®] G. A high variability was even observed between different inserts of the same cultures (cultivated under identical conditions in the same receiver plate) ranging from well-differentiated to completely undifferentiated cultures. The differences in the phenotypic characteristics correlated with the TEER values. While the well-differentiated cultures had TEER values between 600 and 1000 Ω / cm², the values determined for the stratified and undifferentiated cultures were

considerably lower (300 – 400 Ω / cm²). The TEER values of the differentiated cultures were comparable to those that were found in other studies in NHBE cultures (700 – 1200 Ω / cm²) in which functional barrier properties were verified ⁽²³⁰⁻²³¹⁾.

Interestingly, the differentiation was not achieved in any culture when using HBEpC908 beyond the passage four, although no decrease in the cell viability was detectable until passage nine.

The HBEpC812 cells did not show natural differentiation characteristics of bronchial epithelial cells, under none of the tested cultivation conditions.

Conclusions

In general, it was shown that isolated normal human bronchial epithelial cells may retain their ability to develop a phenotype similar to their *in vivo* counterparts and thus provide the most realistic *in vitro* model. The differentiated cell population showed a pseudostratified structure and included ciliated, goblet (secretion of MUC5AC and MUC5B) and basal cells. Moreover, the cell layer formed functional tight junctions with a normal TEER. However, the formation of these properties was strongly depended on appropriate culture conditions and the HBEp cells were shown to be highly sensitive to small deviations from these adjusted conditions.

One out of two tested lots was even shown to be completely unsuited as *in vitro* model due to a lack of the capability to differentiate into a pseudostratified cell layer with mucus producing and ciliated cells, indicating that lot- or donor-specific variations in the differentiation patterns are likely to occur.

In summary, the data suggest that the differentiation of non-transformed bronchial epithelial cells presumably depends on a variety of factors and conditions. These decisive factors might include the handling of the cells after isolation but also the medical and personal background of the respective donor.

Later in this thesis, it is shown that cryopreservation is another critical factor for the differentiation ability of normal human bronchial epithelial cells. Moreover, only cells of the passages one and two were found to be highly suitable for representing realistic *in vitro* models, in general, since in higher passages the ability to develop differentiation characteristics drastically decreased and was greatly varying between different lots.

73

Considering these crucial factors, the use of commercial normal human airway epithelial cells is problematic in many respects:

- First, the cells are generally provided in passage two by the supplier. Thus, experiments can only be started at least with cells of passage three. The ability of these commercial cells to develop realistic features after their arrival in the laboratory is therefore questionable and presumably highly lot-specific.
- 2. Second, commercial cells are mostly shipped as frozen cell stocks. Freezing and thawing of cells generally bears the risk of damage. Moreover, temperature shifts are critical for frozen cells and might lead to a severe decrease in the cell viability after thawing. Therefore the shipping of frozen cells is another risk factor.
- 3. Third, to ensure the availability of cells of the same lot for later experiments, frozen cell stocks have to be prepared. Here again, only cells of passage three or higher can be used. After thawing, they have to undergo another subcultivation cycle before they can finally be employed for the respective studies. Thus, they will probably not be able to develop realistic differentiation characteristics at this stage.

4.1.2.2 Isolation of bronchial epithelial cells from normal human lung tissue

In order to overcome the limitations of commercial normal bronchial epithelial cells and to ensure their availability in sufficient quantity, a protocol for the isolation of cells from human lung tissue was developed.

The lung tissue was frequently provided by the Thoracic Surgery Department (KRH Klinikum Oststadt) from cancer patients that underwent a lobectomy. The study was notified to the Ethics Commission by the hospital and tissue samples were only taken from patients that have signed an agreement form. Extraction of the bronchus pieces from the surgical tissue was usually performed within the first hour after the surgery. The samples were taken from areas distant from the carcinoma. The age of the donors varied between 50 and 84. Two to five samples were taken from each donor, comprising bronchi of different generations. The bronchi had thus varying diameters between approximately 3 - 11 mm. The length of the isolated pieces was between 5 and 10 mm. The samples were stored in ice cold PBS and placed on ice for transportation.

After arrival at the laboratory, the samples were washed twice with PBS and, in some cases, one of the three pieces was subsequently stored in 4 % formaldehyde for fixation at 4 °C for 12 - 14 days. This reference sample was used for the generation of histological sections while the other pieces were prepared for the immediate cell isolation or for storage in liquid nitrogen for later cell isolation. The isolation procedure was basically adopted from Bals and colleagues with some modifications ⁽²³²⁾. First, the bronchi were isolated from the residual surrounding tissue and opened longitudinally by means of a scalpel. The samples that were intended for storage in liquid nitrogen were cut into smaller pieces of about 5 mm x 5 mm. All samples were subsequently placed into plastic tubes each containing 25 - 30 mL of the incubation medium (approximately 3 - 4 pieces per tube). The incubation was performed at 4 °C on a rocking platform for 24 h. The incubation medium contained several antibiotics that are effective against the most common bacteria in the human lung and an antifungal agent. It was additionally supplemented with DNAse I to digest the nucleic acid that leaks from damaged cells and DTT (dithiothreitol) that helps to solubilize proteins. Both agents prevent an increase of viscosity in the medium during the incubation.

After this first incubation step, the sample pieces that were intended for the storage in liquid nitrogen were transferred to cryogenic vials containing 1.8 mL of the cold Recovery[™] Cell Culture Freezing Medium (Life Technologies) or homemade freezing medium (see section 6.1.1.2) and incubated in the "Mr. Frosty" Freezing container (Nalgene) at -80 °C for approximately 24 h before they were transferred to the liquid nitrogen tank. The cryogenic vials and the freezing container were pre-cooled at 4 °C for 1 h before use.

For thawing of the tissue, the cryogenic vials were dipped into a 37 °C water bath until the majority of the Freezing Medium was unfrozen. The medium and the frozen tissue piece were transferred to a Petri dish containing PBS (pre-warmed at room temperature) and washed twice with fresh PBS.

The next steps were either performed with fresh tissue or with tissue pieces that were recovered from liquid nitrogen storage. The specimen was incubated in dissociation medium for another 24 h at 4 °C on a rocket platform. The dissociation medium had the same composition as the incubation medium but was further supplemented with protease XIV, an extremely non-specific proteolytic enzyme that can be used for the dissociation of various tissues. The exact compositions of the used media are described in detail in section 6.1.1.1.

Following the incubation steps, the tissue samples and the dissociation medium were transferred to Petri dishes with a diameter of 9 cm. The bronchus pieces were fixed with forceps at one end while their luminal surface was scraped with a scalpel to remove the epithelial cells from the tissue underneath. The scalpel was rinsed afterwards in the dissociation medium to release the cells into the Petri dish. The cell suspension was subsequently centrifuged for 10 min at 170 x g and the supernatant was discarded. The cell pellet was resuspended in 5 – 10 mL BEGM (Lonza), depending on the size of the tissue sample and the expected cell yield.

After the isolation of epithelial cells, the tissue pieces were further used for the isolation of fibroblasts (see 4.1.3.2).

The suspension containing the isolated epithelial cells was centrifuged again for a few seconds at 310 x g (centrifugation stopped when the speed of 310 x g was reached) to separate the cells from larger tissue fibers. The supernatant (a small residual volume was left in the tube) was then transferred to the collagen/fibronectin coated wells of a 6-well plate. The total volume of the cell suspension per well was between 3 and 4 mL (depending on the approximate number of cells). The final yield of NHBE cells was strongly dependent on the viability of the cells in the tissue sample and was highly variable between the different specimens. The amount of cells showing a strong cilia beating was found to be a reliable measure for the cell viability and predictive for the cell yield. In the case of a high viability, numerous ciliated cells, which were detected through their fast movement, were detectable with an inverted microscope directly after the isolation. The cilia beating of isolated cells usually decreased within a few min and the cells lost their columnar shape and became roundish (Figure 4-28). Cells in cell aggregates retained a strong cilia beating and their columnar shape generally longer- for up to 48 h.



Figure 4-28: Normal human bronchial epithelial cells, resuspended in BEGM, approx. 15 min after isolation from fresh lung tissue. The cells underwent three morphological stages before they adhered to the culture vessel. Cells with strong beating cilia (1) were only detectable within the first 30 min. Afterwards the beating slowed down and the cells became more roundish (2). In the last stage, the cilia were lost and the cells had a round shape (3).

The isolated cells in the wells were supplied with nutrition by partially renewing the medium every other day. Within the first three to four days after isolation, the removed medium was transferred to new collagen/fibronectin-coated wells and all wells were filled up to 3 - 4 mL again with fresh BEGM. With this procedure, the cells were supplied with fresh medium without losing any unattached cells.

Remarkable differences in cell viability, cell yield and time until the cells adhered to the culture vessels were observed between fresh and cryopreserved tissue pieces. When using fresh tissue, the cells usually still formed cell aggregates or were trapped in fibres of the connective tissue after scraping them off the bronchial surface, while only few isolated cells were detectable. The cell viability was mostly very high, meaning that many cells exhibited strong cilia beating. Moreover, the number of epithelial cells was generally also high. However, the most crucial factor for the final yield of NHBE cells was their adherence to the culture vessels. Generally, only isolated cells or cells of small cell aggregates adhered to the culture surface. Larger aggregates, on the other hand, were floating in the medium for many days and did not adhere. The time until the adherence occurred was found to be critical since the early adhering cells usually showed the highest viability and the strongest differentiation ability in later stages while the cells that adhered only after more than eight

to ten days often resulted in NHBE cells with a lower viability or with unusual cell morphology. The adherence of the isolated epithelial cells was further strongly influenced by the level of contamination with erythrocytes (Figure 4-29). The presence of these blood cells generally negatively influenced the attachment of cells due to a lack of space, since the erythrocytes accumulated near the bottom of the wells. The number of isolated erythrocytes highly differed between the tissue samples. However, their number was constantly reduced by transferring parts of the cell suspension from well to well, since these non-nucleated cells are not able to divide.



Figure 4-29: Normal human bronchial epithelial cell culture in BEGM, approx. 2 h after isolation from fresh lung tissue. The culture is highly contaminated with erythrocytes (round biconcave disc-shaped cells) accumulating at the bottom of the culture vessel.

When using cryopreserved tissue, the number of bronchial epithelial cells after the isolation procedure was considerably lower, as well as the cell viability, indicated by a highly reduced number of beating cilia. However, in contrast to fresh tissue, the cells were mostly separated from each other and seldomly trapped within tissue fibres or cell aggregates. Moreover, the cell suspension was free from erythrocytes.

An additional dissociation step was mostly performed when fresh tissue samples were used, in order to "rescue" single cells from cell clumps which were floating in the medium. This procedure was adapted from Fulcher *et al.* ⁽²⁰¹⁾. For that purpose, within 48 h after the cell

isolation, a large part of the cell culture medium containing most of the cell aggregates was removed from some of the wells and transferred to a centrifugation tube. After centrifugation at 220 x g for 7 min, the BEGM was discarded and the cell pellet was resuspended in 10 mL of the "Rescue Medium". This medium contained collagenase, a protease that specifically dissociates collagen fibrils. Residual collagen fibrils of the connective tissue and basal lamina might be the cause for the strong cohesions of the isolated cells within the cell clumps. The cell suspension was transferred to a 25 \mbox{cm}^2 cell culture flask and the dissociation was performed at 37 °C and monitored carefully with an inverse micrsocope. The incubation of the cells in the "Rescue Medium" should not exceed 1 h and should be stopped when the dissociation was completed ⁽²⁰¹⁾. The dissociation was generally stopped after 10 - 20 min by adding 10 % FBS (1 mL). The cell suspension was afterwards centrifuged for 7 min at 220 x g and the cell pellet was resuspended in BEGM. The cells were transferred to an appropriate number of collagen/ fibronectin-coated wells and cultivated as mentioned above. When the cell aggregates were incubated in this special dissociation medium for no longer than 20 min (an incubation of 10 min was mostly sufficient), this procedure was highly efficient to increase the NHBE yield. The cells generally started to adhere shortly after the treatment and within the following 24 h, the majority of cells was attached to the bottom. In contrast, in the wells with "non-rescued" cells, only a minor number of cells adhered and divided. Consequently, the "rescued" cells reached a confluent layer earlier than the "normal" cells and were generally the first to be subcultivated (Figure 4-30). Although dissociation enzymes, such as collagenases, might cause damage to cells, the viability of the NHBE cells which underwent the additional dissociation procedure, (determined by the CASY Cell Analyser System; see 6.1.1.2) was generally comparable to the non-treated cells.



Figure 4-30: Normal human bronchial epithelial cells adhered to collagen/fibronectin coated culture vessels 16 days after isolation. Cells that underwent an additional dissociation step with "rescue medium" (B) formed a confluent layer earlier than the "non-rescued" cells (A) in some cases.

The additional dissociation step was occasionally also performed with cells obtained from frozen tissue, although large cell aggregates were rare. Here it was shown that for some samples, this dissociation step was even essential for obtaining any adhered NHBE cells, since in these few cases, the "normal" procedure did not result in adherent and dividing cells. This observation suggests that the collagenase modifies the cell surface and thus enhances the cell adherence. However, in other cases, the treatment with the rescue medium led to an altered morphology of the cells and a reduced viability.

Besides the "rescue" procedure, a further centrifugation of floating cell aggregates and single cells supported the adherence and increased the cell yield in some samples. Here, the medium of the wells (containing "non-rescued" and non-adhered cells) was centrifuged for 5 min at 500 x g, resuspended in BEGM (the volume was adjusted to the estimated cell density) and transferred to new collagen/fibronectin coated wells. Soon after the centrifugation, single cells started to adhere. After 48 h, even some cell clumps were attached to the bottom and a few cells started to grow out from the aggregates. These outgrowing cells generally differed in their appearance and seemed to be less viable since they carried large vacuoles (Figure 4-31). However, cells with a low viability were mostly lost during the subcultivation procedure and thus did not contribute to the final cell pool. Therefore, the viability of the cells that have undergone the additional centrifugation step was similar to the "normal" (non-centrifuged) cells.

The isolated cells were termed NHBE cells (Normal Human Bronchial Epithelial) and the respective sample number was added to the name of each NHBE lot.



Figure 4-31: An additional centrifugation of the cell suspension of normal human bronchial epithelial cells within approx. 5 days after isolation often resulted in the adherence of large cell aggregates. The cells growing out from these adhered cell aggregates often had an altered morphology

Although the above-mentioned approach was effective to increase the number of adherent cells, the cell yield and viability in general varied strongly between the tissue samples. There are several possible reasons for these differences:

- A. Donor-specific variations
 - \rightarrow depending on age, personal and medical background
- B. Tissue-specific variations
 - ightarrow depending on handling and sampling time after surgical removal
- C. Medium-specific variations (enzyme/antibiotic activity)
 - \rightarrow depending on date of medium preparation

The latter point could be excluded by only using freshly prepared media, while the other factors could not be influenced and thus, tissue specific differences in the yield and viability of NHBE cells existed.

In summary, the use of cryopreserved tissue generally led to a reduced cell yield compared to fresh tissue. However, it was shown that some fresh samples also resulted in a low cell yield which was due to tissue specific variations. On the other hand, the NHBE isolation procedure was more straightforward with cryopreserved specimens because interfering erythrocytes and large cell aggregates were lacking. And, the cell yield could be increased by using more than one tissue piece for the cell isolation. Moreover, the cryopreservation of the tissue enabled the delay of the NHBE isolation to a time point when fresh cells were required.

4.1.2.3 Establishment of normal human bronchial epithelial (NHBE) cell lines

The isolated NHBE cells were subcultivated as soon as they formed a dense layer in the wells (Figure 4-32).



Figure 4-32: NHBE cells forming a confluent cell layer in wells of a 6-well plates approximately 17 days after isolation from tissue.

In some cases, they had already reached approx. 100 % confluency in the middle of the wells, but not at the outer edges. The cells were then subcultivated anyway, to prevent a growth arrest of the confluent cells. The detachment of the cells was achieved with 1 mL trypsin/EDTA (0.05 %/ 0.02 %) per well after rinsing the cells twice with PBS (w/o Mg²⁺ and Ca²⁺). The trypsinization was performed at room temperature for several min until the major part of the cells became detached. The enzyme activity was stopped with 1 mL trypsin inhibitor from soy bean (1 mg/mL in PBS) mixed with 2 mL BEGM. If a large amount of the cells was still attached to the culture vessel, the trypsinization was repeated. After

detachment, the number of viable cells was determined with the CASY Cell Counter and Analyzer System (see 6.1.1.2). The cell suspension was then centrifuged for 5 min at 220 x g to remove the trypsin. Dependent on the subsequent approach, the cell pellet was either resuspended in BEGM or in freezing medium. The medium volume was adjusted to the desired cell number per milliliter. The cells were used for subcultivation in culture flasks, for the establishment of mono- and co-cultures on microporous insert membranes or for storage in liquid nitrogen. For the preparation of cell stocks for storage in liquid nitrogen, only cells of the first two passages were used. Here, the cells were resuspended in a defined amount of Recovery[™] Cell Culture Freezing Medium (Gibco[®], Life Technologies Corp.) or homemade freezing medium (see section 6.1.1.2).

For the general cultivation of the NHBE cells, culture flasks with a growth area of 25 cm² or occasionally with 75 cm² were used (Figure 4-33). In the case of cells of the passage 1, the culture surface was previously coated with collagen/fibronectin to support their adherence and thus to increase the cell yield. The cells were seeded at a density of $3 - 3.5 \times 10^5/25$ cm² and fed with BEGM (BEBM including all enclosed SingleQuotsTM, Lonza). Subcultivation was performed once a week.



Figure 4-33: Normal human bronchial epithelial cells of passage 1 cultivated in a collagen/fibronectin coated culture flasks (25 cm²) and supplied with BEGM.

Comparable to the commercial HBEp cells, no remarkable differences in the proliferation rate and cell viability were observed in NHBE cells when cultivated in cell culture flasks over at least eight passages. But, after this time point, the growth rate started to decrease until it came to a complete arrest, approximately in passage 10.

4.1.2.4 Characterization of normal human bronchial epithelial (NHBE) cell lines

For characterization, freshly isolated NHBE cells between the passages 1 and 5 were grown on inserts with collagen type IV or type I coated microporous membranes (0.4 μ m pores). No obvious differences between the two coating types were observed. However, the collagen type IV coating was mostly used due to its higher physiological relevance through mimicking the biochemical properties of the basement membrane *in vivo*.

Similar to the HBEpC, the seeding density was found to be a crucial factor also for NHBE cells, in order to obtain a confluent cell layer. The optimal density was between $1 - 1.5 \times 10^5$ / cm². The turnover from the submerged cultivation to the air-liquid interface was optimally performed when the cells showed a confluency of 100 % and latest 24 h after reaching this state of complete confluency. Any deviations from this cultivation mode or the appropriate seeding density resulted in a lack of differentiation properties and structural alterations accompanied by a decrease of the culture life span.

The work on commercial human bronchial epithelial cells already revealed that these nontransformed cells are able to form cell populations with ciliated, mucus-producing (MUC5AC and MUC5B) and basal cells and to develop tight junctions with realistic TEER values (600 – 2200 Ω / cm²) when cultivated at the air-liquid interface and supplied with the basic differentiation medium, including Ultroser[®] G. However, these previous studies suggested that the differentiation strongly depends on the passage number and is highly variable between different lots.

In freshly isolated NHBE cells, natural differentiation patterns with goblet, ciliated and basal cells were achieved in nearly all cultures under identical culture conditions when using cells of the passages one and two, independent of donor-specific characteristics (Figure 4-34 - Figure 4-36). Once differentiated, the cultures were also characterized by a long shelf life of at least 12 weeks.

84

However, the ability to develop *in vivo*-like characteristics decreased dramatically in higher passages. Moreover, the differentiation capacity of "older" cells (beyond passage two) showed high donor-specific variations. Here, some lots retained the differentiation ability up to passage 4, while others showed no natural differentiation characteristics and resulted in cultures with a decreased life span.



Figure 4-34: Differentiated culture of NHBE cells of passage 2 cultivated on collagen IV coated microporous membranes (0.4 μ m pores) at the air-liquid interface with basic differentiation medium after a cultivation time of 42 days (38 days under ALI conditions). The cell population is characterized by a pseudostratified structure with basal cells, ciliated cells and non-ciliated columnar cells (potentially secretory cells; HE stained histological sections; magnification: 1000x).



Figure 4-35: Immunohistochemical staining of the mucins 5AC (A) and 5B in a NHBE culture (passage 2) cultivated on collagen IV coated microporous insert membranes (0.4 μ m) at the air-liquid interface with basic differentiation medium after a cultivation time of 42 days (38 days under ALI conditions). The cell nuclei were counterstained with hematoxylin (Magnification: 400x).



Figure 4-36: Immunohistochemical staining of the basal cell marker p63 in a NHBE culture (passage 2) cultivated on collagen IV coated microporous insert membranes (0.4 μ m) at the air-liquid interface with basic differentiation medium after a cultivation time of 42 days (38 days under ALI conditions). The cell nuclei were counterstained with hematoxylin (Magnification: 400x).

The fact that no decrease in the cell viability was observed until passage eight when cultivated in cell culture flasks (see 4.1.2.3) clearly demonstrated that the decrease in the differentiation ability cannot be associated with the general cell viability of NHBE cells.

Although, NHBE cells of the passages one and two were mostly able to develop the desired natural differentiation properties, independent on the original sample, lot-specific morphological variations in the differentiated cell populations were nevertheless abundant. These variations comprised the distribution of the different cell types within the cell layers as well as the shape of the cells and structure of the cultures. In cultures of some NHBE lots, a higher secretion activity was observed, pointing to an increased number of goblet cells, while in other cultures a higher number of ciliated cells were detected. The columnar-shape of the ciliated and goblet cells as well as the pseudostratified structure was also found to be more pronounced in some distinct cultures. Since these variations were reproducibly obtained, the occurrence of donor- or tissue-specific differences in NHBE cells was clearly proven in this thesis.

The availability of NHBE cells of passage one and two was the limiting factor for the performance of experiments with cultures of these cells. The time from the isolation to the final use of NHBE cells was between three and five weeks and the number of cells was highly varying between the samples. Therefore, the suitability of cryopreserved NHBE cells was also tested. For that purpose, different lots of NHBE cells that were frozen in liquid nitrogen in the passages one or two were cultivated under the same conditions as fresh cells that never underwent cryopreservation. In most cases, the cryopreserved cells did not retain the ability to show the natural differentiation characteristics, like the formation of cilia or a pseudostratified structure, as they did before freezing (Figure 4-37). The cultures mostly had

a stratified growth and the cells showed a squamous morphology. Moreover, the shelf life of these cultures was reduced to approximately four weeks until the cells partially started to detach and holes within the cell layer became visible.



Figure 4-37: Two cultures of the same NHBE lot before (A) and after cryopreservation (B). Both cultures were cultivated under identical conditions (on collagen IV coated insert membranes, at the air-liquid interface and with basic differentiation medium) for 26 days (21 days under ALI conditions; HE stained histological sections; magnification: 400x).

Conclusions

Freshly isolated NHBE cells in passage one and two provide the most reliable and realistic *in vitro* model of the airway epithelium since they are able to differentiate into all common cell types of the bronchial epithelium, including ciliated, goblet and basal cells. In addition, the formation of functional tight junctions in these cultures was verified by TEER values which ranged between 600 and 2000 Ω / cm². Moreover, in contrast to the commercial HBEp cells, the development of the natural differentiation characteristics was highly reproducible in NHBE cultures when using cells of the passages one and two.

However, the restricted availability of cells in passage one and two as well as the loss of the differentiation ability after cryopreservation and the donor-specific variations limit the suitability of NHBE cells as universal *in vitro* model for toxicological studies, especially in the case of large scale assessments. Nevertheless, whenever possible, freshly isolated normal human bronchial epithelial cells should be the first choice for *in vitro* studies on the bronchial epithelium or they should be employed for verification of the results obtained with permanent cell lines.

Cryopreservation of lung tissue was found to be a valuable method to guarantee the availability of NHBE cells at a defined time point.

4.1.3 Human lung fibroblasts

In order to realize 3D co-culture systems mimicking the EMTU of the bronchi, a suitable human lung fibroblast line was additionally required. For that purpose, the commercial fetal fibroblast line IMR90 was initially used. Later adult fibroblasts were also included which were isolated from human bronchial tissue.

4.1.3.1 IMR90- fetal human lung fibroblasts

The IMR90 human fibroblast cell line is derived from a 16-week-old female fetus with a normal karyotype. The IMR90 is a commercial cell line (ATTC CCL-186) and thus permanently available. Moreover, it is described to have a long life span, of up to 50 passages ⁽²³³⁾. These characteristics recommend the IMR90 as a cell line for the routine work in order to provide reproducible conditions. Accordingly, IMR90 fibroblasts have already been used in various co-culture studies ^(178, 234-235).

The cells were obtained in passage 10 and showed a stable and high viability up to 15 more passages. A pool of frozen cell stocks has been established and the IMR90 cells were used between the passages 13 and 25.

The IMR90 cells were cultivated in 75 cm² culture flasks with a seeding density of 2×10^6 cells/ 75 cm² and supplied with EMEM (MEM Eagle, Lonza 125F) supplemented with 2mM L-glutamine, 10 % FBS and 1 mM sodium pyruvates. Subcultivation was performed twice a week.

4.1.3.2 Establishment of normal human lung fibroblast cell lines (NHLF)

Due to the fact that the IMR90 fibroblasts are derived from a fetus and might therefore show properties that are distinct from adult cells, the inclusion of adult fibroblasts for verifying the results obtained with IMR90 is advantageous. For that purpose, a protocol for the isolation of fibroblasts from the human bronchial tissue samples was developed.
In the first approaches, only fresh lung tissue was used. Later, the fibroblast isolation from cryopreserved samples was also tested. Here, the same tissue pieces that were previously used for scraping off the epithelial cells were "recycled". The isolation of normal human lung fibroblasts (NHLF) was performed according to the outgrowth procedure where the tissue was attached to a culture surface and, under appropriate culture conditions, the fibroblasts grew out from the samples. In general, the first fibroblasts were visible around the tissue piece earliest after seven days (Figure 4-38).



Figure 4-38: Fibroblasts growing out from human bronchial tissue cultivated in Petri dishes (3 cm diameter) for 20 days.

The tissue was then further cultivated for another seven to ten days before it was removed and the cells were cultivated in the absence of the tissue until reaching the desired cell density. During the first attempt of fibroblast isolation, the importance of the following parameters was evaluated:

- Size of tissue samples
- Diameter of Petri dishes
- Medium composition

These investigations revealed that the optimal tissue size for the isolation of fibroblasts was around 3×3 mm. Sharp cutting edges were shown to promote the fibroblast outgrowth

which were obtained by using a scalpel. For unknown reasons, the cell yield was significantly higher when using Petri dishes with a diameter of 3 cm compared to 9 cm Petri dishes. The medium level played a crucial role for the adherence of the tissue pieces on the culture surface. Best results were obtained when the medium level covered approximately half of the tissue piece. If the medium volume was too high, the tissue detached from the surface.

In the first approaches with fresh lung tissue samples, the specimens and isolated fibroblasts were supplied with RPMI 1640 supplemented with 10 % FBS, L-glutamine (2 mM), sodium pyruvate (1 mM) and penicillin/streptomycin (10 U/ 10 μ g/mL).

In later approaches the isolation of pure fibroblast cultures under above mentioned culture conditions sometimes failed. Here, another cell type was predominantly isolated. Besides a few epithelial cells and fibroblasts, a third cell type was outgrowing from the tissue in high numbers. Most likely, these cells were macrophages, since this cell type is described to be the first to grow out from tissue ⁽²³⁶⁾. The cell morphology also supports the presumption of macrophage isolation. The supposed macrophages accumulated directly beneath the tissue pieces and were thus only detectable after removing the samples. In contrast, the fibroblasts migrated to the outer edge of the tissue and were visible alongside the sample (Figure 4-39).



Figure 4-39: Two different cell types growing out from human bronchial tissue cultivated in Petri dishes (3 cm diameter) for 14 days. After removal of the tissue sample, a large number of presumed macrophages (1) were detected, while only a few fibroblasts (2) grew out from the tissue at the outer edges.

Isolated macrophages are described to have a low proliferation rate and to contribute little to the final culture after trypsinization ⁽²³⁶⁾. In fact, the freshly isolated macrophages started to die at the outer edges of the outgrowth population a few days (2 - 3) after the removal of the tissue piece. However, due to their large number, they were still predominant and the few isolated fibroblasts were often suppressed. If a mixed cell population composed of assumed macrophages and fibroblasts was subcultivated and transferred to cell culture flasks, none of the cell types could be expanded.

In later approaches, the utility of cryopreserved tissue samples for the fibroblast isolation was tested. Under the above mentioned culture conditions, the outgrowth of fibroblasts was not achieved. But when using culture medium with a higher concentration of D-glucose [4.5 g/L; DMEM (Biochrom) incl. L-glutamine, sodium pyruvate, supplemented with 10 % FBS and penicillin/streptomycin (10 U/ 10 μ g/mL)], the isolation of fibroblast was successful with some of the tested tissue samples. Compared to fresh lung tissue, the time until the first cells were detectable at the edges of the samples took at least 14 days. No macrophage contamination was observed when using cryopreserved tissue.

Interestingly, once the fibroblast outgrowth was achieved, the particular tissue piece yielded a large number of fibroblast over several weeks when transferred to new Petri dishes from time to time. After removal of the tissue, the fibroblasts were cultivated for another five to seven days until subcultivation.

For maintenance and expansion, the cells were cultivated in 75 cm² culture flasks with a seeding density of $1 - 2 \times 10^6$ cells/ 75 cm² and supplied with RPMI (incl. 10 % FBS, L-glutamine, sodium pyruvates and penicillin/ streptomycin). The subcultivation was performed twice a week and the cells showed a stable viability at least until passage ten. A pool of frozen cell stock was prepared from each lot between the passages two and four.



Figure 4-40: Normal human lung fibroblasts of passage 2 after isolation from lung tissue cultivated in 75 cm^2 culture flasks and supplied with RPMI incl. 10 % FBS, L-glutamine, sodium pyruvates and penicillin/streptomycin

Conclusions

The results revealed that the success of isolating fibroblasts from normal human bronchial tissue varied significantly between different samples. When using fresh lung tissue, the NHLF yield was strongly dependent on the extent of macrophage contamination which was found to differ between tissue samples of different donors. On the other hand, the outgrowth of NHLF from cryopreserved tissue was only achieved in some of the tested samples.

However, if the NHLF isolation was successful, the cell yield was generally very high and the cell viability was stable for at least ten passages.

4.2 Development of 3-dimensional in vitro models of the airway epithelium

In order to simulate the natural interactions between the bronchial epithelial layer and the underlying connective tissue, including the ECM and fibroblasts, as they take place *in vivo*, two different 3D co-culture models were developed in this thesis.

4.2.1 Sandwich Cell Model (SCM)

The Sandwich Cell Model (SCM) is based on a protocol that has originally been developed for the construction of 3D skin equivalents ⁽²³⁷⁾. It provides a realistic *in vitro* replication of the bronchial epithelium. Similar to the in vivo structure, the SCM is composed of an epithelial cell layer and the underlying lamina propria - a layer of loose connective tissue whose main components are fibers of ECM proteins in which cells are embedded. Fibroblasts represent the predominant cell type and collagen type I the most abundant protein in this type of connective tissue. The SCM is accordingly composed of epithelial cells, fibroblast and a collagen type I matrix. The practical implementation of the natural airway epithelial structure was realized by the following basic protocol: Fibroblasts were suspended in a collagen solution which was previously brought to a neutral pH by adding a NaOH-containing buffer solution (see 6.1.2). An appropriate volume of this collagen-fibroblast solution was applied onto microporous membranes of cell culture inserts and incubated at 37 °C for polymerization of the collagen. Afterwards, the epithelial cells were applied onto the hardened collagen matrix and the cultures were cultivated under submerged conditions until the epithelial cells formed a confluent layer. To promote the differentiation of the epithelial cells, the cultures were subsequently supplied with the basic differentiation medium and cultivated at the air-liquid interface (Figure 4-41).



Figure 4-41: Basic protocol for the Sandwich Cell Model (SCM). A: Lung fibroblasts are suspended in a pH neutral collagen solution and applied onto the membrane of a cell culture insert. B: After the complete polymerization of the collagen at 37°C, lung epithelial cells are applied on the collagen matrix and the co-cultures are cultivated under submerged conditions. C: As soon as the epithelial cell layer reached 100% confluency, the cultures are cultivated at the air-liquid interface to promote the cell differentiation (1= Collagen matrix; 2= Fibroblasts; Undifferentiated epithelial cells; 4= Ciliated cell; 5= Basal cell; 6= Goblet cell; 7= Mucus).

As described in section 3.3.3.1.1, there are many commercial collagen type I products with different properties. In the first approaches, acid soluble collagen I isolated from rat tails (Serva Electrophoresis GmbH) was used for the SCM cultures. This type of collagen has also been used for the generation of bronchial epithelial *in vitro* models in other studies ^(177, 238).

Initial experiments revealed that the most critical factor for the successful construction of SCM cultures is the achievement of a stable collagen matrix. A frequently occurring problem was the contraction of the polymerized collagen matrix a few days after preparation, leading to instability of the cultures.

In vivo, contraction of the connective tissue is an important event during wound healing. Fibroblasts are described to mediate collagen contraction both *in vivo* and *in vitro* ⁽²³⁸⁾. *In vitro* studies furthermore revealed that epithelial cells enhance the fibroblast-mediated collagen contraction ^(177, 238).

In the SCM cultures, the contraction was either starting shortly after changing the cultivation conditions from submerged to ALI or within the first ten days under ALI conditions, depending on other applied parameters (see below). During contraction, the collagen matrix detached from the housing of the cell culture insert and occasionally also from the membrane. The detachment mostly started at one side of the culture and then expanded stepwise. The extent of contraction was differing between cultures and also between inserts of the same culture (cultivated under identical conditions; Figure 4-42).



Figure 4-42: 12-well receiver plate with ten cell culture inserts carrying SCM cultures prepared from different epithelial cells (HBEpC, 16HBE14o-) and two HBEpC monocultures (marked). All SCM cultures clearly show contraction of the collagen matrix.

In most cases, the shrinkage highly reduced the surface of the culture and made it impracticable for the generation of intact histological sections or immunofluorescence analyses. Moreover, the development of a natural structuring of the epithelial layer was hindered by the collagen contraction. A wave-like structure with an uneven multi-layered growth was formed as a consequence of the non-uniform contraction and by compression of the epithelial cells due to the sudden lack of space.

Besides the collagen contraction, another crucial point of the SCM cultures was found to be the viability of the fibroblasts which were embedded in the collagen matrix. Observations by *in situ* microscopy and histological sections revealed that the fibroblasts often stopped proliferation or even died. Both, the collagen contraction and the viability of cells embedded in collagen were potentially strongly related to the physical properties of the matrix ⁽²³⁹⁾. For example, the pH of the matrix and the thickness and arrangement of collagen fibers seem to influence the cell viability and morphology of fibroblasts. In addition, the fiber characteristics are dependent on the incubation temperature and time ⁽²⁴⁰⁻²⁴¹⁾.

In order to obtain a stable collagen matrix with viable fibroblasts, different modifications of the original protocol with regard to the pH, incubation temperature and duration as well as the thickness of the collagen matrix were tested. The influence of the pH was tested by adding different amounts of the NaOH buffer solution. This approach also led to variations in the collagen concentration. It was found, that an increase of the pH from originally 7 - 7.5 to approximately 8 - 8.5 by adding a higher volume of NaOH buffer solution improved the stability of the collagen matrix and delayed the contraction. Nevertheless, after 7 - 14 days the collagen matrix started to contract also in these cultures.

The time point when the contraction started, as well as its extent, was depending on the respective epithelial cell line. The SCM cultures prepared with the two permanent cell lines 16HBE14o- and Calu-3 showed a less shrinkage of the collagen matrix and the contraction started later than in cultures with HBEpC.

In further approaches, the effects of the incubation temperature and duration as well as the thickness of the collagen matrix were tested. For that purpose, the collagen solution was applied onto the membranes in different amounts (150 - 300 μ l) and incubated either at 4 °C, room temperature or 37 °C for varying time intervals (2 – 5 h or overnight). However, none of the tested variations delivered reproducible SCM cultures without collagen

95

contraction. But, modifications of the temperature resulted in differences in the cell viability of the fibroblasts. The highest viability of this cell type was observed when the collagen matrix was immediately incubated at 37 °C.

SCM cultures that were stable for up to four weeks (showing no collagen contraction) were only obtained sporadically and in these cases histological sections revealed that either the epithelial cells did not form a confluent layer or the fibroblasts were lost by cell death or both. These observations clearly suggest that both cell types play a pivotal role in the collagen contraction *in vitro*.

In order to investigate the role of the collagen source for producing stable collagen matrices, collagen type I solutions from different suppliers and isolated by different techniques and/or from other animals and organs were tested in the next steps.

Two collagen solutions prepared from bovine skin by acid-based extraction methods (3 mg/mL, BD Biosciences and Sigma-Aldrich) delivered similar results as the previously used rat tail collagen. None of the tested approaches resulted in stable and reproducible SCM cultures. Moreover, contraction of the collagen matrix was even more pronounced than in previous cultures that were prepared with the rat tail collagen solution. Thus, the two bovine skin collagen solutions were also precluded from further approaches.

Next, two collagen solutions isolated from porcine tendons were tested (Cellmatrix I-A and I-P, Nitta Gelatin Inc.). Both products contain 3 mg/mL of collagen type I and have a pH of 3.0. The difference is that the Cellmatrix I-A solution contains acid-soluble collagen while the Cellmatrix I-P consists of pepsine-solublized collagen. As stated by the manufacturer, the two collagen solutions are specially designed to culture cells on top or embedded within the collagen matrix. A comprehensive instruction for the preparation of the collagen matrices was furthermore provided. The basic protocol was as follows: The collagen solutions were each mixed with NaOH neutralizing buffer (see 6.1.2) and 10X MEM medium at a ratio of 8:1:1. When cultivating cells embedded in the polymerized collagen, the manufacturer recommends the preparation of a basal collagen layer as shown in Figure 4-43.



Figure 4-43: Schematic representation of a SCM culture composed of a basal collagen layer (A: 100 μ l collagen solution); and a main collagen layer in which the fibroblast are embedded (B: 170 – 200 μ l collagen solution; 1= Ciliated cell; 2= Basal cell; 3= Goblet cell; 4= Mucus).

Different combinations of the two collagen solutions for the matrix preparation were tested. For the base layer, the Cellmatrix I-A or I-P were used alternatively. For the main collagen layer (with embedded fibroblasts) a 1:1 mixture of both collagen solutions was additionally tested. Best results with regard to matrix stability and fibroblast viability were obtained when using the combination of Cellmatrix I-A and I-P solutions. The use of Cellmatrix I-A alone resulted in a decreased stability and a contraction of these matrices was observed more often. The Cellmatrix I-P seemed to produce matrices with a higher stiffness. A higher ECM matrix stiffness was suggested to be beneficial for the proliferation and gene regulation of fibroblasts ⁽²⁴²⁾. The use of a 1:1 mixture of I-A and I-P, resulted in a collagen/fibroblast layer with a high stability for up to six weeks without a visible contraction, at least when cultivated in the absence of epithelial cells. Moreover, these results were highly reproducible.

The preparation of a base collagen layer between the membrane and the main collagen matrix was found to be not beneficial. On the contrary, these SCM cultures showed a decreased stability and thus this additional layer was omitted in later cultures. The purpose of this layer was to prevent the fibroblasts from attaching to the membrane and growing there. However, since histological section revealed that the fibroblasts were uniformly distributed in the main collagen matrix, the base layer was not required.

For the main collagen matrix, a volume of $200 - 250 \mu$ l of the collagen/fibroblast solution was found to be optimal. The number of fibroblasts should rather be low $(4 - 5 \times 10^5)$ culture), since they generally showed a steady proliferation during the cultivation time. Although the fibroblasts were only sporadically detected in the histological

section of SCM cultures (5 – 7 μ m thickness), their growth and morphology could be well monitored by *in situ* microscopy.

In order to find optimal conditions for the polymerization of the collagen matrix, different incubation times and temperatures were applied. However, these parameters had no obvious impact on the matrix stiffness and stability but only on the duration of the complete polymerization. The final protocol includes an incubation step at 37 °C for 3 - 5 h before the epithelial cells are applied onto the polymerized collagen matrix (see section 6.1.2).

In initial experiments, Calu-3 cells were used for the generation of SCM cultures with the Cellmatrix collagen solutions since previous approaches with rat tail collagen revealed that these cells induced a less intense contraction of the collagen than the HBEpC. The SCM cocultures composed of Calu-3 and IMR90 cells and a collagen matrix prepared from Cellmatrix I-A and I-P (1:1) were stable for at least four weeks. No contraction of the collagen matrix was observed during this time but a slight degradation of the collagen starting after approximately 20 days. The Calu-3 cells showed their typical monolayered growth and columnar shape. Moreover, mucus secretion was observed by inverse microscopy and verified by immunohistochemical analyses of MUC5AC in these cultures (Figure 4-44).

The limiting factor of these SCM cultures was the life span of the Calu-3 cells. As described in section 3.5.2.2, the Calu-3 cell line is characterized by an early differentiation of the cells after approximately three days under ALI conditions but also by a limited life span. In SCM cultures, the cells showed an altered morphology after four to five weeks. At the same time, a partial degradation of the collagen was detected. The cultures had a milky to cloudy appearance and histological sections revealed small crumbles within the collagen matrix.

As discussed earlier in this thesis, normal, non-transformed human bronchial epithelial cells (passage 1 and 2) provide the best *in vitro* model of the bronchial epithelium since they retained most of the natural characteristics of their *in vivo* counterparts. Moreover, once differentiated, the cultures were characterized by a long life span of several months. Thus, they are well suited for long term toxicological studies. Co-culture systems composed of NHBE or HBEp cells and fibroblasts embedded in a matrix of ECM proteins provide an even better reflection of the natural structuring and might furthermore enable the simulation of interactions taking place within the EMTU (see section 3.4.2). Accordingly, these models are one step closer to a realistic *in vitro* model.

98



Figure 4-44: SCM cultures composed of Calu-3 and IMR90 cells (embedded in collagen I) after a cultivation time of 26 days (20 days under ALI conditions). The monolayered Calu-3 cells are characterized by a columnar-shape. A: slight degradation of the collagen is indicated by small crumbles within the matrix (HE stained histological sections). B: The presence of mucus producing cells was verified by immunohistochemical analyses of MUC5AC (see arrows; the cell nuclei were counterstained with hematoxylin; magnification: 400x)

But, first SCM cultures of HBEpC prepared with rat tail collagen I revealed a low stability and reproducibility, due to a strong contraction of the collagen matrix. The contraction was more pronounced than in cultures of Calu-3 or 16HBE14o- indicating the presence of factors that promote matrix remodeling. However, since normal human bronchial epithelial cells (HBEpC or NHBE) in general showed extensive passage- and donor-specific variations in their differentiation patterns, their suitability for the generation of stable and reproducible SCM cultures might potentially also differ. Therefore, for further evaluation and optimization of the SCM co-culture system with the Cellmatrix I-A and I-P, NHBE cells from different donors and of different passages were used in addition to the HBEpC.

Although the use of Cellmatrix I-A and I-P (1:1) for the preparation of the collagen matrix highly enhanced the stability of the SCM cultures with NHBE or HBEp cells, a contraction was nevertheless observed in some cultures.

Overall, NHBE cells of passage one and two resulted in the most stable SCM cultures. Here, mostly no contraction of the collagen matrix was observed for at least up to 25 days and the cultures were well reproducible when using cells of the same donor and passage. However, donor-specific variations in the SCM cultures became evident, since a few cultures prepared with NHBE cells of passage one still showed a contraction already after 14 days under ALI conditions. When using NHBE cells in higher passages (non-cryopreserved) for the preparation of SCM cultures, contraction of the collagen was observed more frequently.

Degradation of the collagen was observed in all cultures, starting after 3-4 weeks, and its degree increased with increasing culture time (Figure 4-45).

The differentiation ability of NHBE cells was considerably reduced in SCM cultures. The epithelial cells mostly formed a stratified layer with two distinct regions: One to three layers of cells with a "normal" morphology which stained positive for the basal cell marker p63 and an upper layer composed of cells with a flattened morphology. Cytokeratin 13 (CK13), which is a marker for squamous cells, was only detected sporadically in few cells and to a low amount, suggesting that no squamous metaplasia-like phenotype, as shown in section 4.3, was developed in these cultures (Figure 4-46).



Figure 4-45: SCM cultures prepared with NHBE cells of passage 1 after a cultivation time of 29 days. The cultures did not show a contraction of the collagen matrix but a high degradation or drying out of the collagen.



Figure 4-46: Immunohistochemical analysis of the basal cell marker p63 (A) and the squamous cell marker CK13 (B) in SCM cultures prepared from NHBE cells of passage 2 and IMR90 cells embedded in collagen type I, after a cultivation time of 18 days (14 days under ALI conditions). A: One to three layers of basal cells were detected by immunohistochemical analyses of the p63 marker. B: CK13 was detected only sporadically in few cells, suggesting that no squamous metaplasia-like phenotype was developed (the cell nuclei were counterstained with hematoxylin; magnification: 400x)

The contraction of the collagen matrix enhanced the differentiation of the NHBE cells. Although, a multilayered structure was also found in these cultures, the upper layer exhibited cells with a normal columnar shape (Figure 4-47). However, this structure was only found in some areas of the cultures and the reproducibility of these contracted SCM cultures was highly limited.

The use of different fibroblast lines - either NHLF of different donors or IMR90 - had no influence on the matrix remodeling or on the growth and differentiation patterns of the epithelial cells. Moreover, the seeding density of the epithelial cells was also found to neither effect the induction and extent of the collagen contraction nor their differentiation ability (Figure 4-48).



Figure 4-47: SCM culture prepared with NHBE (PO3) and IMR90 showing a strong contraction of the collagen matrix after a cultivation time of 18 days (14 days under ALI conditions). The NHBE cells form two layers. The upper layer shows cells with a natural columnar shape. (HE stained histological sections; magnification: 400x)



Figure 4-48: SCM cultures prepared with HBEpC (P05) and IMR90 after a cultivation time of 20 days. The HBEpC were applied with different seeding densities, 1.5×10^5 (A) and 3.0×10^5 (B) which had no obvious effect on the stability of the collagen matrix or on the differentiation pattern. (HE stained histological sections; magnification: 400x).

In order to enhance the differentiation capacity of NHBE cells in SCM cultures, the collagen matrix was complemented with other ECM proteins which are known to promote the differentiation of bronchial epithelial cells *in vivo*.

The ECM proteins laminin, fibronectin and collagen IV were added alternatively to the collagen solution in physiologically relevant concentrations:

Laminin: 4.4 µg/mL (approx. 1 µg/ insert culture)

Fibronectin: 9.2 µg/mL (approx. 2 µg/ insert culture)

Collagen type IV: 27.3 µg/mL (approx. 6 mg/ insert culture)

In the cultures prepared with laminin, a strong contraction of the collagen matrix was observed after 14 days. The generation of histological sections from these cultures was difficult and thus no concluding results about the effect of laminin on the differentiation of NHBE cells in SCM cultures are available.

The addition of fibronectin also reduced the stability of the SCM cultures. While the control cultures (without fibronectin) were stable up to at least 22 days, a contraction of the collagen matrix was detectable after 18 days in cultures with fibronectin. Moreover, no obvious improvement of the cell differentiation was observed when adding fibronectin (Figure 4-49).



Figure 4-49: SCM cultures prepared with NHBE cells of P02 and IMR90 after a cultivation time of 22 days (18 days under ALI conditions). The collagen matrix was either supplemented with 2 μ g (total concentration) fibronectin (A) or not (B). No remarkable differences in the structure and cell morphology were detected between these cultures. (HE stained histological sections; magnification: 400x).

The SCM cultures containing collagen type IV as additional ECM component did not show a contracted collagen matrix for up to four weeks. But here a strong degradation of the collagen was observed starting after approximately 20 - 23 days. The differentiation pattern of the NHBE cells did not show any obvious differences to the control cultures (without collagen IV addition). They again formed two distinct layers composed of a basal layer with "normal cells" covered by a second layer with flattened cells as observed in previous studies. Ciliated cells were not detected in any SCM cultures but mucus secretion (MUC5AC) was verified in the cultures prepared with and without collagen type IV after 14 days (Figure 4-50).



Figure 4-50: Immunohistochemical analysis of MUC5AC in SCM cultures composed of NHBE cells (P01) and IMR90 with (A) and without (B) collagen IV added to the collagen I matrix (after a cultivation time of 22 days; 18 days under ALI conditions). The epithelial cells of both cultures formed two distinct layers with normal shaped basal cells covered by flattened cells. A few MUC5AC producing cells were detected in both cultures (arrows; Magnification: 400x)

In summary, the use of the Cellmatrix I-A and I-P (1:1) collagen solutions enhanced the practicability of the SCM as an *in vitro* model of the bronchial epithelium. These SCM cultures were characterized by a high stability of the collagen matrix and accordingly by an improved reproducibility when prepared with Calu-3 cells as epithelial component. On the other hand, when using NHBE cells, a contraction of the collagen was sometimes still observed. Moreover, degradation of the collagen was found in all SCM cultures, but it was highest in those with NHBE or HBEp cells as epithelial component. No degradation and contraction of the collagen was observed in cultures that were solely composed of fibroblasts embedded in collagen and that were lacking the epithelial layer. Therefore, these matrix remodeling events can clearly be attributed to the bronchial epithelial cells.

As described in section 3.4.2.1, matrix metalloproteinases (MMP) are known to play a central role in matrix remodeling and degradation. While fibroblasts express a variety of proteins of the MMP family only some of them are also produced by epithelial cells, like MMP-9 and MMP-2. MMP-9 is elevated in the airways of COPD and asthma patients ⁽¹²²⁾ and since these airway diseases are associated with remodeling of the connective tissue in the

airways, investigations on the role of MMP-9 in collagen contraction and degradation in the SCM cultures were performed by immunohistochemical staining.

MMP-9 expression was verified in both Calu-3 and NHBE cells. Moreover, positive staining signals were also found throughout the collagen matrix. In NHBE cells, the MMP-9 expression was restricted to the basal cell layer. Since the Calu-3 cells were characterized by a monolayered growth without basal cells, the staining pattern of MMP-9 differed in these cultures. However, interestingly the Calu-3 cells showed a clear polarization and MMP-9 expression was here also restricted to the basal side of the cells. (Figure 4-51).

There were no obvious differences in the staining intensities and accordingly the levels of MMP-9 expression seemed to be similar in SCM cultures of NHBE and Calu-3 cells. This suggests that MMP-9 is not responsible for the observed differences in collagen contraction and degradation and it can be concluded that there are other main inducers of the collagen remodeling events in SCM cultures that need further investigation.



Figure 4-51: Immunohistochemical analyses of MMP-9 expression in SCM cultures prepared with Calu-3 (A) or NHBE cells (B). A: The expression is strongest on the basal side of the polarized Calu-3 cells and throughout the collagen matrix. B: MMP-9 expression is restricted to the basal cell layers and the collagen matrix, while no MMP-9 expression was detected in cells of the upper layer (Maginification: 400x).

Conclusions

Although the Sandwich Cell Model provide a good *in vitro* simulation of the bronchial epithelium regarding the structure and potential interactions between epithelial cells, fibroblasts and the extracellular matrix, its practicability and suitability was shown to be highly limited, due to the following reasons:

1. Contraction is a frequently occurring event when working with collagen matrices ^(177, 238). Although matrices prepared from a 1:1 mixture of the Cellmatrix I-A and I-P collagen solutions highly increased the stability of the matrices, a contraction was nevertheless observed in many SCM cultures when using normal (non-transformed) human bronchial epithelial cells. While the differentiation of the epithelial cells was partially enhanced in SCM cultures with contracted matrices, the reproducibility was considerably reduced. Moreover, since the collagen shrinkage greatly reduced the culture surface (up to approximately 60 %), their suitability for exposure experiments is thus limited and the paraffin embedding for histological analyses was complicated. The induction of the collagen contraction was potentially donor- and passage-specific but it was also sometimes varying between inserts of the same culture (prepared from the same cells and cultivated under identical conditions) and therefore not predictable.

2. Degradation of the collagen was another common occurrence in SCM cultures. Histological sections revealed that this degradation was starting at the upper end of the collagen matrix, directly beneath the epithelial cells. Moreover it was not detected in cultures that were only composed of fibroblasts embedded in the collagen matrix. This clearly suggests that degradation is also mainly induced by the epithelial cells. The collagen degradation was another factor leading to a reduced reproducibility and stability of the SCM cultures. Moreover, the life span of the cultures was thus considerably decreased.

3. The differentiation ability of NHBE and HBEp cells was highly diminished in SCM cultures. A realistic structure and morphology comparable to the bronchial epithelium was not achieved under none of the tested culture conditions. Mucus secretion was only detected sporadically in a few cells and cilia were completely lacking in these cultures. Moreover, the cells showed an abnormal stratified growth. Therefore the suitability of these cultures to serve as a reliable *in vitro* model is restricted. Only the Calu-3 cells showed a normal growth

107

and differentiation in this co-culture model. But, Calu-3 cultures, in general, are not suited for long-term studies.

4. The application of several analytical methods is complicated or not feasible in the SCM coculture system. For example, the two cell types can barely be monitored separately by *in situ* microscopy. Contraction and degradation additionally hinder the analysis by *in situ* microscopy and complicate the paraffin embedding procedure for histology. The determination of TEER values was found to be impracticable in SCM cultures with the Millicell ERS-2 Volt-Ohm Meter (Millipore). In addition, individual analyses of epithelial cells and fibroblast, like the measurement of their cell viability or gene expression analyses by RT-PCR are not feasible since they cannot be separated from each other. In summary, many commonly analyzed endpoints are not applicable and consequently the use of SCM cultures for toxicological testing is rather impracticable.

4.2.2 Compartment Cell Model (CCM)

In order to overcome the limitations of the SCM, another co-culture system composed of fibroblasts and epithelial cells was developed during this thesis. The principle of this 3D culture method was published by Pohl et al. in 2009. In contrast to the SCM, the two cell types are spatially separated in this co-culture model and can only interact via soluble factors that are released into the cell culture medium ⁽¹⁷⁴⁾.

In the original protocol according to Pohl and colleagues, the two cell types are cultivated on opposite sides of the microporous membranes of cell culture inserts. This protocol was initially adopted in this thesis. Briefly: The insert was turned by 180 °C and the fibroblast suspension was applied on the basolateral side of the membrane. After an incubation time of approximately 3 hours to allow the fibroblast to adhere to the membrane, the inserts were repositioned in the wells of the receiver plate and the epithelial cell solution was added onto the apical side of the membrane which was previously coated with collagen type I or IV. The cultures were cultivated under submerged conditions until the epithelial cells formed a confluent cell layer. Afterwards, the epithelial cells were cultivated at the air-liquid interface while the fibroblast remained submerged in the cell culture medium (basic differentiation medium; Figure 4-52).



Figure 4-52: Schematic representation of the original protocol of the CCM (according to Pohl *et al.* 2009). **A**: The fibroblasts are seeded on the basolateral side of the microporous membrane of cell culture inserts. **B**: The epithelial cells are applied on the apical side of the membrane and thus, the two cell types are spatially separated but can interact through secreted molecules that are released into the cell culture medium. The cultures are cultivated under submerged conditions until the epithelial cells formed a confluent layer. **C**: The epithelial cells are afterwards cultivated at the air-liquid interface to promote their differentiation while the fibroblasts remain submerged in the culture medium (1= Fibroblasts; 2= Undifferentiated epithelial cells; 3= Ciliated cell; 4= Basal cell; 5= Goblet cell; 6= Mucus).

It became soon evident that this co-culture model is not suitable for long-term cultivation. Histological sections revealed a gradual detachment of the fibroblasts from the membrane in all cultures starting already after a cultivation time of eight to twelve days. In some cases, the detached fibroblasts were removed during the medium renewal, but mostly they adhered to the bottom of the multiwell plate and grew there. Thus, they were still able to interact biochemically via the cell culture medium.

The latter observation gave rise to the CCM *variant I*, a modified protocol of the CCM, which enhanced the co-culture system with regard to an easier preparation and more convenient handling. Here, the IMR90 fibroblasts were seeded on the bottom of 12-well receiver plates at a density of approx. 1 x 10⁴ cells/well and supplied with their specific growth medium (EMEM including 10 % FBS). The epithelial cells were applied onto the membranes of cell culture inserts (coated with collagen I or IV) and cultivated under conventional, cell line-specific conditions in a separate receiver plate. After the epithelial cells nearly reached 100 % confluency, the inserts were transferred to the fibroblast-containing wells and the growth medium was replaced by differentiation medium. After another two days, the air-liquid interface was established to promote the differentiation of the epithelial cells (Figure 4-53).



Figure 4-53: Schematic representation of the CCM variant *I*. **A**: The epithelial cells are seeded on microporous membranes of cell culture inserts and cultivated under conventional, cell-specific conditions until reaching nearly 100 % confluency. B: The fibroblasts are seeded on the bottom of insert receiver plates and cultivated separately under cell-specific conditions for several days. C: The inserts with the epithelial cells are placed into the fibroblast-containing receiver plate and the growth medium is replaced by the differentiation medium, but the cultures are still cultivated under submerged conditions for another 2 days. D: The epithelial cells are afterwards cultivated at the air-liquid interface to promote their differentiation while the fibroblasts remain submerged in the culture medium (1= Undifferentiated epithelial cells; 2= Fibroblasts; 3= Ciliated cell; 4= Basal cell; 5= Goblet cell; 6= Mucus).

For comparison, monocultures of both cell types were cultivated under identical conditions. The protocol was tested with NHBE cells and IMR90 fibroblasts or the Calu-3 cell line with IMR90.

Histological sections of NHBE cells cultivated in CCM *variant I* cultures revealed that this coculture system promoted their differentiation. Ciliogenesis was observed earlier and was more pronounced in co-cultures than in the corresponding monocultures of the NHBE cells (Figure 4-54).



Figure 4-54: Differentiated NHBE cells that were either cultivated in CCM *variant I* co-cultures with IMR90 fibroblasts (A) or as monoculture (B) for 24 days (19 days under ALI conditions). The ciliogenesis was enhanced in co-cultures (HE stained histological sections; magnification: 1000x).

Moreover, the TEER values were considerable higher when the NHBE were co-cultivated with fibroblasts in CCM *variant I* co-cultures, indicating the presence of an enhanced barrier function in these cultures (Table 4-3). The inserts were placed in new receiver plates for TEER measurement, to exclude influences of the fibroblasts on the TEER values.

Table 4-3: Comparison of TEER values of each three inserts of NHBE cells either cultivated in CCM *variant I* co-cultures with IMR90 fibroblasts or as monoculture after a cultivation time of 19 days.

TEER value [Ω/cm2]	NHBE in co-culture with IMR90	NHBE monoculture	Culture age (days)
	>900	<600	19

In Calu-3 cells no considerable differences in cell morphology or their differentiation ability were observed between CCM *variant I* co-cultures and monocultures. However, when regarding the TEER values, it was shown that the development of tight junction was here also enhanced in the co-cultures.

Table 4-4: Comparison of TEER values of Calu-3 cells either cultivated in CCM *variant I* cocultures with IMR90 fibroblasts or as monoculture after a cultivation time of 15 and 29 days. One insert each was analyzed.

	Calu-3 in co- culture with IMR90	Calu-3 monoculture	Culture age (days)
TEER value $[\Omega/cm^2]$	1733	763	15
	3047	2192	29

Overall, interactions between fibroblasts and bronchial epithelial cells were clearly observed in the CCM *variant I*. The fibroblasts seemed to promote and enhance the differentiation of the epithelial cells in both cancer-derived and normal cells. However, a highly critical point for this co-culture system was the frequent replacement of the fibroblast before they reached 100 % confluency. Confluent fibroblasts express genes that are involved in cellular senescence, like the transcription factor p16 ⁽²⁴³⁾. Under these conditions, the secretion products of the fibroblasts might be altered and consequently their effect on the epithelial

cells possibly differs. Indeed, alterations in cell morphology or structure were observed in NHBE cells in CCM *variant I* cultures when the fibroblast formed a confluent cell layer. To overcome this difficulty, the fibroblasts were frequently replaced by a new receiver plate with sub-confluent fibroblasts before complete confluency was reached. The inserts with the epithelial cells were transferred to this new plate one to three days after seeding the fibroblasts.

However, this procedure hindered the continuous monitoring and analysis of the fibroblasts over the complete cultivation time. It is conceivable that the interactions between fibroblasts and epithelial cells are reciprocal and might therefore also have an effect on the fibroblasts. Moreover, a frequent replacement of the "old" with "new", unaffected fibroblasts might disturb the ongoing process of interactions between the two cell types.

These considerations, led to the development of a second modified protocol of the CCM, the CCM *variant II*.

Here, the fibroblasts were embedded in a collagen matrix, similar to the SCM, allowing a three-dimensional growth of the cells. The collagen matrix was prepared according to the same protocol as for SCM cultures (with Cellmatrix I-A and I-P). The collagen/fibroblast solution was poured into the wells of the receiver plate and incubated for approx. 3 h at 37 °C until the polymerization of the collagen was completed. Afterwards these fibroblast cultures were cultivated with the specific growth medium (EMEM (IMR90) or RPMI (NHLF) incl. 10 % FBS) until starting the co-cultivation. The cell culture inserts were placed into the receiver plate when the epithelial cells nearly formed a confluent cell layer and the co-cultures were supplied with differentiation medium under submerged conditions for one day. The inserts were then raised to the air-liquid interface (Figure 4-55).

The studies on the CCM *variant II* cultures were accompanied by monocultures of the respective epithelial cell line and of collagen-embedded fibroblasts, as a reference.

112



Figure 4-55: Schematic representation of the CCM *variant II*. A: The epithelial cells are seeded on microporous membranes of cell culture inserts and cultivated under conventional, cell-specific conditions until reaching nearly 100 % confluency. B: The fibroblasts are solved in a pH neutral collagen solution and applied on the bottom of insert receiver plates. After polymerization of the collagen (37 °C, approx. 3 h), they are supplied with their cell-specific growth medium and cultivated separately for two to seven days. C: The inserts with the epithelial cells are placed into the fibroblast-containing receiver plate and the growth medium is replaced by the differentiation medium, but the cultures are still cultivated under submerged conditions for another day. D: The epithelial cells are afterwards cultivated at the air-liquid interface to promote their differentiation while the fibroblasts remain submerged in the culture medium (1= Undifferentiated epithelial cells; 2= Fibroblasts; 3= Collagen matrix; 4= Ciliated cell; 5= Basal cell; 6= Goblet cell; 7= Mucus).

In initial approaches, the volume of the collagen/fibroblast solution was optimized, in order to form a uniform matrix within the wells while leaving an appropriate space between the matrix and the insert membrane for medium supply. The optimal volume was 270 µL.

The nutrition supply was found to be a critical factor in the CCM *variant II*. The capacity of the wells was reduced due to the collagen matrix, which was especially problematic during the ALI cultivation. A lack of nutrition indicated by a rapid acidification of the medium was often observed in initial approaches, leading to a reduced differentiation ability of the epithelial cells. However, if the medium level was too high, it penetrated the epithelial layer and hindered the ALI cultivation and accordingly also decreased the differentiation ability of these cultures. Therefore, the medium volume was optimized in a stepwise process and was found to be appropriate in a range of 550 – 580 μ L. The medium was renewed every 48 h and care was taken to avoid damage of the collagen matrix.

The collagen matrix was stable over several months and no contraction was observed. The fibroblasts, either NHLF or IMR90, showed a continuous growth and normal spindle-shape morphology (Figure 4-56). Due to the 3-dimensional growth, they nevertheless did not form a confluent layer and accordingly contact inhibition was prevented.



Figure 4-56: NHLF embedded in a collagen matrix and grown on the bottom of 12-well receiver plates. A: NHLF in collagen two days after seeding. B: NHLF in collagen after a cultivation time of 29 days in a CCM *variant II culture* with NHBE cells.

Similar to the SCM, degradation of the collagen was also observed in these cultures, starting after approx. 3 weeks. While this degradation had no obvious adverse effects on the fibroblasts or epithelial cells, it hindered the observation of the fibroblasts by *in situ* microscopy. Interestingly, the degradation was partially prevented when no additional calcium chloride was added to the differentiation medium. Due to this observation, the basic differentiation medium was modified for all CCM *variant II* cultures by omitting the additional calcium chloride. This modification did not negatively influence the differentiation of the epithelial cells but was highly advantageous for the stability of the collagen matrix.

The CCM *variant II* co-culture system was first evaluated with different lots of NHBE cells and either IMR90 or NHLF cells. A normal differentiation pattern of the NHBE cells was observed in these cultures with a pseudostratified structure and columnar-shaped cells. Moreover, the cell population included ciliated, mucus-producing and basal cells (Figure 4-57).

The cultures were stable for at least three months. Their exact life span has not been investigated; however, since no differences in cell viability and morphology were observed after 12 weeks, it is likely that they can be used for long-term experiments beyond three months.



Figure 4-57: NHBE cells (P01) cultivated in CCM variant II co-cultures with NHLF after a cultivation time of 84 days (79 days under ALI conditions). The cell population exhibited MUC5AC secreting cells (A; arrows), p63 positive basal cells (B) and ciliated cells (C; cilia stained with acetylated α -tubulin; cell nuclei counterstained with hematoxylin; magnification: 400x)

The cultures were characterized by an extensive secretion and an enhanced development of cilia when compared to the respective monocultures. The epithelial cell layer was rinsed with PBS frequently (once a week) to partially remove the mucus that otherwise accumulated on top of the cultures. This procedure revealed that the viscosity of the secretion products varied between the mono- and co-cultures. The mucus of the co-cultures was characterized by a very high viscosity indicating an enhanced expression of gel-forming mucins, like MUC5AC and 5B. In some cultures the mucus layer could be retained after fixation and paraffin embedding and was thus visible in the histological sections. Immunohistochemical analysis showed a strong staining for MUC5AC in this mucus layer (Figure 4-58).



Figure 4-58: NHBE cells cultivated in CCM *variant II* co-culture with NHLF after a cultivation time of 84 days (79 days under ALI conditions). The culture showed an extensive mucus secretion and the mucus accumulated on top of the epithelial cells. This mucus layer could be preserved during fixation and paraffin embedding and was thus still visible in the histological section. Immunohistochemical analyses revealed a strong staining of MUC5AC in this mucus layer (the cell nuclei were counterstained with hematoxylin; magnification: 400x).

As stated in 4.1.2.4, monocultures of NHBE cells showed donor-specific differences in their growth pattern and differentiation properties. In some cultures, a pseudostratified structure with columnar-shaped cells was barely achieved. Moreover, the development of ciliated cells was rare in some lots of NHBE cells. In these cases it was shown, that the differentiation could be highly enhanced by co-culturing these cells with NHLF or IMR90 fibroblasts in CCM *variant II* cultures (Figure 4-59).

The ciliogenesis generally started earlier and was more pronounced in the CCM *variant II* cocultures than in the corresponding monocultures. Immunofluorescence analysis on these cocultures revealed the presence of a high number of ciliated cells which were uniformly distributed across the epithelial cell layer (Figure 4-60).



Figure 4-59: NHBE cells cultivated in monoculture (A; 29 days; 25 days under ALI conditions) and CCM *variant II* co-culture with IMR90 fibroblasts (B; 38 days; 33 days under ALI conditions). NHBE cells of the same donor and passage were used and the cultures were cultivated under identical conditions (HE stained histological sections; magnification: 400x).



Figure 4-60: Immunofluorescence analysis of NHBE cells (P01) in CCM variant II co-cultures with IMR90 after a cultivation time of 55 days. Cilia are stained with acetyl- α -tubulin (red) and cell nuclei with DAPI (blue; magnification: 400x).

The barrier function of the NHBE cell layer through the development of tight junctions was furthermore enhanced in the CCM *variant II* co-culture system. The TEER values were generally significantly higher in the co-cultures than in NHBE monocultures. Examples are given in Table 4-5.

NHBE NHBE in co-Culture age culture with monoculture (days) NHLF (n=3) (n=3) 2349 (± 191) 896 (± 39) 27 **TEER value** $[\Omega/cm^2]$ 1058 (± 182) 2016 (± 57) 39

Table 4-5: Comparison of TEER values of NHBE cells either cultivated in CCM variant II co-cultures with NHLF fibroblasts or as monoculture (average of three inserts each \pm SD).

After the comprehensive evaluation of the CCM *variant II* co-culture system for the cultivation of NHBE cells, its suitability was further tested with Calu-3 cells. Here, generally no remarkable differences in cell morphology and differentiation were detected in histological sections of mono- and co-cultures.

The CCM variant II prepared with Calu-3 resulted in cultures that were stable for at least four weeks, a life span which is comparable to Calu-3 monocultures. However, when grown as monocultures, Calu-3 cells often showed a flattened morphology when reaching the end of their culture life span. On the contrary, in the CCM variant II, the Calu-3 cells still exhibited a normal, columnar morphology after 29 days (Figure 4-61). Although the exact life span of Calu-3 cells in CCM variant II cultures was not yet investigated, the results suggest that this co-culture system might prolong the cell viability of these cells.



Figure 4-61: Calu-3 cells cultivated in CCM *variant II* cultures with IMR90 (A) or as monoculture (B) after a cultivation time of 29 days (23 days under ALI conditions; HE stained histological sections; magnification: 400x).

A positive effect of the cultivation of Calu-3 cells in CCM *variant II* cultures was furthermore suggested when regarding the TEER values. Higher values were consistently observed in Calu-3 co-cultures than in Calu-3 monocultures, indicating an enhancement of the formation of barrier properties by the fibroblasts. Examples are given in Table 4-6.

	Calu-3 in co- culture with IMR90	Calu-3 monoculture	Culture age (days)
TEER value [Ω/cm ²]	1392	2065	29
	790	1590	18
	916	1257	26

Table 4-6: Comparison of TEER values of Calu-3 cells either cultivated in CCM *variant II* co-cultures with IMR90 fibroblasts or as monoculture. One insert each was analyzed.

Conclusions

The results highlight the suitability and usefulness of the CCM *variant II* co-culture system to serve as a relevant *in vitro* model of the airway epithelium, due to the following reasons:

1. It was clearly demonstrated that interactions between fibroblasts and bronchial epithelial cells - normal cells as well as Calu-3 - were enabled in the CCM *variant II*, leading to an enhanced differentiation of the epithelial cell layer. These results suggest that the presented co-culture model provides a more realistic *in vitro* replica than monocultures by taking into consideration the reciprocal interactions taking place in the EMTU. *In vivo* these complex interactions between fibroblasts, ECM and epithelial cells play a pivotal role in inflammation response and wound healing ⁽²⁴⁴⁾. Accordingly, the CCM *variant II* might provide a valuable model for toxicological testing *in vitro*.

2. The surrounding collagen matrix allows a 3-dimensional growth of the fibroblasts. Thus, a continuous proliferation is enabled without reaching complete confluency. Accordingly, the fibroblasts can also be cultivated for several months and monitored throughout the complete cultivation time of the CCM *variant II* cultures. The use of collagen as scaffold for the 3D cultivation furthermore provides a more realistic environment for the fibroblasts and might support their natural physiological properties.

3. In contrast to the SCM, the CCM *variant II* allows the separate analyses of both cell types - epithelial cells and fibroblasts - with all commonly used analytical methods, such as ELISA, Western Blot, immunofluorescence and cell viability assays. The production of histological sections of the epithelial layer was also straightforward and the determination of TEER values was feasible in these cultures. This point further supports the usefulness of the CCM *variant II* cultures for toxicological assessments.

4.3 Stimulation of NHBE cells by special additives in the culture medium

The most important requirement of suitable *in vitro* models of the airway epithelium for toxicological testing is a realistic response, including natural signaling pathways, after the exposure to selected molecules or stimuli. Moreover, pathologic features, as goblet cell hyperplasia or squamous metaplasia, should ideally be inducible in these cultures in order to provide *in vitro* models of diseased airways which most likely behave differently in exposure studies.

In this thesis, a squamous metaplasia-like phenotype was occasionally observed in mono- or SCM co-cultures prepared with NHBE cells or HBEpC. These cultures showed a stratified

growth with one or more layers of p63 positive basal cells covered by multiple layers of flat, metaplastic epithelial cells which showed strong expression of the squamous cell markers CK6 and CK13 (Figure 4-62). A similar structure was found in human and rat lungs exhibiting squamous metaplasia ⁽²⁴⁵⁻²⁴⁶⁾.



Figure 4-62: SCM culture of HBEpC 908 cells (P03) with IMR90 fibroblasts after a cultivation time of 19 days (14 days under ALI conditions). The epithelial cells show a stratified growth (A; HE stained histological section) with multiple layers of p63 positive basal cells (B). The upper layers are characterized by cells with a squamous morphology which show a high expression of the squamous cell markers CK13 (C) and CK6 (D; the cell nuclei were counterstained with hematoxylin; magnifications: A-C: 630x; D: 400x)

This structural alteration was consistently observed in all cultures prepared with the commercial HBEpC 812 and could thus clearly be attributed to the cell line in this case (see also section 4.1.2.1).

However, in other cases, the altered phenotype was only achieved sporadically while other cultures prepared with the same cells (same donor and passage) that were cultivated under identical conditions showed a normal differentiation with ciliated, mucus-producing and basal cells (Figure 4-63). In these cultures, the factors leading to the development of the stratified growth and squamous morphology of the cells could not be determined and this altered phenotype was thus not specifically inducible.



Figure 4-63: Monocultures of NHBE cells, lot 005 (P02) cultivated under identical conditions for 23 days (19 days under ALI conditions). A: In some insert cultures, a stratified growth, squamous morphology and keratinization of the NHBE cells was detected. B: In other insert cultures, the NHBE cells showed a normal growth and differentiation pattern (HE stained histological sections; magnification: 400x)

In order to find conditions that reproducibly lead to the development of the pathological phenotype of squamous metaplasia or goblet/ basal cell hyperplasia, different modifications of the cell culture medium were tested by changing concentrations of single components or by adding special signaling molecules that are associated with airway diseases.

4.3.1 Effects of retinoic acid deficiency in NHBE cultures

Retinoic acid (RA) deficiency is described to induce squamous metaplasia in airway epithelial cells both *in vitro* and *in vivo* (see 3.5.4) ⁽¹⁹⁵⁾. In the present study, the lack of RA in the cell culture medium was shown to induce diverse responses in the NHBE cell cultures. While in some cultures a squamous metaplasia-like growth and cell morphology was observed, others did not reveal any differences compared to the control cultures that were supplied with 100 nM RA. These cultures showed a rather normal differentiation into ciliated- and mucus producing cells and did not show a flattened morphology also under RA deficiency (Figure 4-64). These observations suggest that the induction of squamous metaplasia by RA deficiency is strongly dependant on the respective cell line and thus it was rejected as universal factor for the induction of pathologic features of airway diseases in *in vitro* models.



Figure 4-64: NHBE monocultures after a cultivation time of 30 days (26 days under ALI conditions). The cultures were either cultivated under the conventional culture conditions (with basic differentiation medium (A) or without retinoic acid in the differentiation medium (B). Ciliated cells were detected in both cultures (HE stained histological sections; magnification: 630x)

4.3.2 Effects of different EGF concentrations in NHBE cultures

The Epidermal Growth Factor (EGF) is important for the growth of epithelial cells *in vitro* and is thus an essential component of commercial cell culture complete media for airway epithelial cells, like the BEGM (Lonza). A concentration of 0.5 ng/mL is generally used and higher concentrations, like 10 or 25 ng/mL were shown to impair the normal differentiation of airway epithelial cells ^(9, 158, 200). In this thesis, the effect of different EGF concentrations was tested in different lots of NHBE cells. As a starting point, a concentration of 250 ng/mL was evaluated, which is a 100-fold increase of the concentration used in some formulations of cell culture media for airway epithelial cells ^(9, 158). In subsequent approaches, EGF concentrations of 50 and 40 ng/mL were tested. The control cultures were supplied with the basic differentiation medium which contained 0.5 ng/mL EGF. All analyses were performed using NHBE monocultures. The effect of 50 ng/mL EGF was additionally tested in NHBE cells that were co-cultivated with IMR90 fibroblasts in CCM *variant II* co-cultures. The stimulation of the cultures by increasing the EGF content in the cell culture medium was started at different time points of the cultivation. Until then, the cells were cultivated under

conventional culture conditions. The duration of the stimulation was also varied in the different approaches. Table 4-7 gives an overview of all experiments conducted in order to analyze the effect of elevated EGF concentrations.

Table 4-7: List of different approaches for the stimulation of NHBE cells with increased EGF concentrations. Two different NHBE cell lots (011 and 015) were stimulated with varying concentrations of EGF. The starting point (culture day) as well as the duration of stimulation (days) was varied. The stimulation of the NHBE015 cells with 50 ng/mL was performed with mono- and CCM *variant II* co-cultures (with IMR90 cells).

NHBE lot no.	EGF concentration [ng/mL]	Starting point (culture day)	Days of stimulation	Co-culture (CCM variant II)
011	250	9	20	Ν
015	50	13	22	Y
011	40	20	8	Ν

The results clearly demonstrated that the EGF concentration was a critical factor for the normal differentiation of NHBE cells and structural alterations are inducible by modifying the EGF content in the cell culture medium.

In the first approach, a 500-fold increased EGF concentration was applied. The starting point was set within the early differentiation phase of the NHBE cultures, only five days after the ALI cultivation was initiated. After 20 days of stimulation with 250 ng/mL, the NHBE cells showed a severe impairment of their normal differentiation. Histological sections revealed a stratified growth with flattened cells that started to detach in the upper layers (Figure 4-65). The cell morphology further suggested a strong cornification. In addition, the cultures had no measurable TEER.


Figure 4-65: Differentiated NHBE cultures after a cultivation time of 29 days (25 days under ALI conditions). The cultures were either cultivated under conventional culture conditions (with basic differentiation medium; **A**) or with an increased EGF concentration of 250 ng/mL leading to a severe impairment of the normal differentiation (**B**; HE stained histological sections; magnification: 400x)

In the subsequent approach, the effect of 50 ng/mL EGF, an 100-fold increase of the standard concentration, was tested. Here, another lot of NHBE cells was used and CCM *variant II* co-cultures were additionally included. As a control, monocultures of fibroblasts embedded in a collagen matrix were also provided with the increased EGF concentration. The stimulation started at day 13 after seeding the epithelial cells, which was the ninth day of ALI cultivation. After 22 days of stimulation with the elevated EGF concentration, all NHBE monocultures showed an irregular growth pattern with several areas showing a multilayered (stratified) structure interspersed by areas with only one or two cell layers. The cells in the middle layers of the stratified areas showed a squamous morphology and stained positive for the squamous cell markers CK6 and CK13 (Figure 4-66).



Figure 4-66: Differentiated NHBE cultures after a cultivation time of 35 days. The cultures were either cultivated under conventional culture conditions (with basic differentiation medium; A) or with an increased EGF concentration of 50 ng/mL (B-D). The stimulation with 50 ng/mL EGF resulted in the formation of stratified areas which were irregularly distributed over the culture surface [(B); A-B: HE stained histological sections; magnification: 200x]. The middle layers of these stratified areas were composed of squamous cells which showed a positive staining in immunohistochemical analyses of CK13 and CK6 [(C-D); the cell nuclei were counterstained with hematoxylin; magnification: 400x].

Interestingly, the increased EGF concentration had no obvious effect on the cell growth and morphology when the NHBE cells were co-cultivated with IMR90 in the CCM variant *II* co-culture system. This might be explainable by the fact that fibroblasts also possess EGF receptors (EGFR) that can bind EGF molecules and thus, less EGF is available that can be bound by the EGFR of the epithelial cells. *In vivo*, injured epithelial cells release a variety of factors, including EGF, to which fibroblasts respond by migration, contraction and proliferation. This activation of fibroblasts, in turn, is essential for the wound healing of the epithelium which is associated with structural alterations like hyperplasia or squamous metaplasia. Therefore, it is likely that these alterations will also be detectable in CCM variant *II* cultures, when increasing the time of EGF stimulation.

An impairment of the barrier function of the NHBE cells was already observable in both mono- and co-cultures, indicated by a decrease in the TEER values (Table 4-8).

Moreover, ELISA analysis revealed a slight increase of IL-8 secretion after stimulation with 50 ng/mL EGF in NHBE cells in both mono- and co-culture, as well as in IMR90 monocultures, compared to the respective non-stimulated control cultures (Figure 4-67). These results suggest an inflammatory response in both cell types after the stimulation with 50 ng/mL EGF.



Figure 4-67: Results of ELISA analysis of IL-8 secretion (pg/mL) in NHBE015 mono- or CCM variant II co-cultures and IMR90 monocultures (embedded in collagen) after stimulation with 50 ng/mL compared to control cultures.

In a third approach, the EGF concentration was raised to 40 ng/mL, which is an 80-fold increase of the normal concentration. In contrast to the previous approaches, the cells were almost fully differentiated before the stimulation started, indicated by cilia beating and mucus secretion as observed by *in situ* microscopy. Moreover, the duration of the stimulation was reduced to eight days. Histological sections of the stimulated cultures revealed an altered phenotype with a stratified structure, which was uniformly distributed over the entire cultures. These alterations were found in all four inserts of the test cultures, suggesting a high reproducibility of the results (Figure 4-68).



Figure 4-68: NHBE monocultures after a cultivation time of 28 days (24 days under ALI conditions). The cultures were either cultivated under conventional culture conditions (with basic differentiation medium; A) or with an increased EGF concentration of 40 ng/mL leading to basal cell hyperplasia (B; HE stained histological sections; magnification: 400x)

Immunohistochemical analyses with several markers consistently confirmed the existence of basal cell hyperplasia in the EGF-stimulated cultures (Figure 4-69). The basal cell marker p63 revealed the presence of multiple layers of basal cells (Figure 4-69 A-B). The superficial layer showed no positive staining of p63 but here a normal differentiation pattern with mucus-producing and ciliated cells was verified (Figure 4-69 C-F). This structure is typical for basal cell hyperplasia *in vivo*. Interestingly, immunohistochemical analysis of the proliferation marker Ki-67 showed positive signals only in a few basal cells, although the basal cell hyperplasia is generally induced by an increased proliferation of the basal cells (Figure 4-69 G-H). The fact that the Ki-67 marker was expressed only in a low number of basal cells might indicate that the development of the basal cell hyperplasia was completed in these cultures. *In vivo*, basal cell hyperplasia is believed to be an irreversible alteration but also a transitory state for the development of squamous metaplasia. Therefore there were three possible explanations for the low proliferative activity in the culture: (1) a steady state of basal cell hyperplasia was entered; (2) reversion of the basal cell hyperplasia was in progress. The second

hypothesis is not likely, due to the fact that the EGF stimulation was still proceeding at the time point when the analyses were conducted. The induction of squamous metaplasia was also not indicated, since the squamous cell marker CK13 was only sporadically expressed in a few cells, which was similar to the CK13 staining pattern in the control cultures (Figure 4-69 I-J). However, considering the structural alterations found in the previous approach with 50 ng/mL, the later transition to squamous metaplasia is conceivable, when increasing the stimulation time.



Figure 4-69: Immunohistochemical analyses of various markers in NHBE monocultures stimulated with 40 ng/mL and control cultures. Further information are provided in the main text (the cell nuclei were counterstained with hematoxylin; magnification: 400x).

The TEER values revealed a severe impairment of the barrier function properties of the NHBE cells after stimulation with 40 ng/mL EGF (Table 4-8).

Table 4-8: TEER values of NHBE cultures (cultivated as mono- or CCM *variant II* co-cultures) after stimulation with increased EGF concentrations compared to control cultures. The TEER values of the test cultures are listed for each insert individually, while those of the respective control cultures are shown as mean values with standard deviation (±).

Culture Type	EGF conc. [ng/mL]	Days of EGF stimulation	Culture age	TEER [Ω/cm2]	Control (Mean TEER; [Ω/cm2])
Mono (NHRE015)	50	22	25	1585	- 2001 (+ 268)
	50	J 22 3	22	1317	2001 (± 208)
CCM variant II	50	22	25	1155	4025
(IMR90, NHBE015)	50	22	35	1395	1835
				487	-
Mono (NHBE011)	40	$28 \qquad \frac{528}{518}$	1201 (+ 16)		
	-10		20	518	1201 (± 10)
				496	

ELISA analyses showed again an increased secretion of IL-8 in the EGF-stimulated NHBE cultures, which was in congruence with the results of the previous experiment. However, both ELISA and TEER analyses also clearly demonstrated the existence of extensive donor-specific differences. The average TEER value of the non-stimulated monocultures of NHBE015 was 2001 Ω / cm², while that of NHBE011 monocultures was 1201 Ω / cm² (Table 4-8). The average IL-8 secretion of non-stimulated NHBE015 monocultures was 7597 pg/mL at a total cultivation time of 35 days and it was elevated to 24995 pg/mL after stimulation with 50 ng/mL EGF for 21 days (Figure 4-70). In comparison, non-stimulated NHBE011 monocultures revealed a low IL-8 secretion, in average of 214 pg/mL, which was increased to 1125 pg/mL after eight days of stimulation with 40 ng/mL EGF. These donor-specific variations might lead to varying effects of certain stimuli in different lots of NHBE cells and it is thus recommendable to test the reproducibility of the results by using different NHBE lots for each experiment in the future.



Figure 4-70: Results of ELISA analysis of IL-8 secretion (pg/mL) in NHBE011 monocultures after stimulation with 40 ng/mL compared to control cultures.

In summary, dependent on the concentration and/ or the donor, EGF either induced basal cell hyperplasia or metaplasia-like changes in NHBE cultures. The latter was only found in some distinct areas of the culture surface and here squamous cells were observed. The secretion of the proinflammatory cytokine IL-8 was shown to be increased and the barrier function property was impaired in both cases, suggesting a general effect of EGF on these two parameters.

4.3.3 Effects of TGF-ß1 stimulation in NHBE cultures

The Transforming Growth Factor-ß1 (TGF-ß1) has repeatedly been reported to cause squamous metaplasia-like alterations in several lung cell lines ^(117, 131). In order to develop *in vitro* models showing such pathologic features, NHBE cultures were stimulated with different concentrations of TGF-ß1 and for varying time periods in this thesis. Based on the findings of Doerner and Zuraw (2009), which described morphological changes in NHBE cells after the exposure to 2 ng/mL for 3 days, this concentration was initial adopted in this study and the stimulation was performed for four days ⁽¹²⁸⁾. However, in these initial studies, no obvious morphological alterations could be observed between the cultures that were exposed to TGF-ß1 and the control cultures by histology and *in situ* microscopy. Moreover, the TEER values were also similar in both cultures.

Therefore, the concentration as well as the duration of exposure was increased in later approaches. Table 4-9 gives an overview of all stimulation experiments on NHBE cells with TGF-ß1.

Table 4-9: List of different approaches for the stimulation of NHBE cells with TGF-ß1. Four different NHBE lots (008, 011, 015 and 023) were stimulated with varying concentrations of TGF-ß1. The starting point (culture day) as well as the duration of stimulation (days) was varied. The stimulation of the NHBE008 and NHBE015 cells with 3 ng/mL was performed with mono- and CCM variant II co-cultures (with IMR90 cells).

NHBE lot no.	TGF-ß1 concentration [ng/mL]	Starting point (culture day)	Days of stimulation	Co-culture (CCM variant II)
008	2	22	4	Ν
011	4	20	8	Ν
023	4	22	7	Ν
008	3	13	13	Y
015	3	20	22	Y

It was shown that the TGF-ß1 stimulation of NHBE cells with all the tested variations of concentration and duration did not induce a squamous metaplasia-like change. Although alterations of the cell morphology from columnar to squamous were observed after stimulation with 40 ng/mL for seven or eight days in cultures of two different NHBE lots, no stratified structure resembling the squamous metaplasia *in vivo* was developed (Figure 4-71).



Figure 4-71: NHBE monocultures of two different lots (023 and 011) after a cultivation time of 28 (NHBE011) and 29 (NHBE023) days, respectively. The cultures were either cultivated under conventional culture conditions (with basic differentiation medium; A-B) or with the addition of 40 ng/mL TGF-ß1 for 7 (NHBE023; C) or 8 (NHBE011, D) days leading to alterations of the cell morphology from columnar to squamous (HE stained histological sections; magnification: 400x)

The stimulation with 3 ng/mL TGF-ß1 for 13 and 22 days in two different lots of NHBE cells resulted in the formation of stratified areas which were irregularly distributed over the culture surface. These areas also did not show the typical structure of squamous metaplasia, since they were composed of two to three layers of squamous cells and an upper layer with columnar cells (Figure 4-72). *In vivo*, the squamous metaplasia is characterized by an upper layer showing squamous cells and multiple underlying layers of proliferating cells which appear less flattened.



Figure 4-72: NHBE monocultures of two different lots (015 and 008) after a cultivation time of 26 (NHBE008) and 42 (NHBE015) days, respectively. The cultures were either cultivated under the conventional culture conditions (with basic differentiation medium; A-B) or with the addition of 3 ng/mL TGF-ß1 for 22 (NHBE015; C) or 13 (NHBE008, D) days leading to the formation of stratified areas with an upper layer composed of columnar cells and 2 - 3 underlying layers with squamous cells (HE stained histological sections; magnification: 400x)

The TEER values as well as the IL-6 and IL-8 secretion showed high donor-specific variations but did not deliver any concluding results and were thus not considered further.

The TGF-ß1 stimulation of NHBE cells co-cultivated with fibroblasts in CCM variant II cultures did not induce obvious morphological changes under the tested conditions. Comparable to the EGF stimulation, a potential explanation is that both cell types, NHBE and fibroblasts, have TGF beta receptors 1 which bind competitively the free TGF-ß1 molecules in the cell culture medium.

In conclusion, the stimulation of NHBE cells with TGF-ß1 resulted in morphological alterations in all different approaches. However, the cultures did not show characteristics of squamous metaplasia, as it was expected. The potential of TGF-ß1 to induce such metaplastic changes when modifying the conditions, with regard to the concentration and duration of the stimulation, needs further investigation.

4.3.4 Effects of IL-13 stimulation in NHBE cultures

Interleukin-13 (IL-13) is described to induce goblet cell hyperplasia *in vitro* ^(132-133, 247). In the literature, there is different information on the IL-13 concentration leading to an increase of the number of goblet cells. In the present study, a concentration of 10 ng/mL IL-13 was tested for the stimulation of NHBE cells of two different lots (Table 4-10). In both experiments, the cultures were in the late differentiation stage (day 22), showing already some ciliated cells and mucus secretion, when starting the IL-13 stimulation.

Table 4-10: List of different approaches for the stimulation of NHBE cells with IL-13. Two different NHBE lots (008 and 023) were stimulated with 10 ng/mL IL-13. The starting point (culture day) as well as the duration of stimulation (days) was varied.

NHBE lot no.	IL-13 concentration [ng/mL]	Starting point (culture day)	Days of stimulation	Co-culture (CCM variant II)
008	10	22	15	Ν
023	10	22	7	Ν

In the first approach the cultures were prepared for histology after seven days of stimulation and the histological sections revealed that an increase of mucus-producing cells was already detectable at that time (Figure 4-73).



Figure 4-73: NHBE monocultures after a cultivation time of 29 days (24 days under ALI conditions). The cultures were either cultivated under conventional culture conditions (with basic differentiation medium; A) or with the addition of 10 ng/mL IL-13 for 7 days leading to goblet cell hyperplasia (B; HE stained histological sections; magnification: 400x)

In the second approach, the duration of the stimulation was prolonged to 15 days and here it was shown, that an extensive goblet cell hyperplasia was induced and a strong expression of mucin 5AC was observed by immunohistochemistry (Figure 4-74).



Figure 4-74: NHBE monocultures after a cultivation time of 37 days (32 days under ALI conditions). The cultures were either cultivated under conventional culture conditions (with basic differentiation medium; A) or with the addition of 10 ng/mL IL-13 for 15 days leading to goblet cell hyperplasia (B; A-B: HE stained histological sections; magnification: 400x). A strong staining of the mucin 5AC was verified in the IL-13 stimulated cultures by immunohistochemistry (C; the cell nuclei were counterstained with hematoxylin; magnification: 400x)

ELISA analyses revealed that there was no significant change in IL-6 and IL-8 secretion in the IL-13 stimulated cultures compared to the control cultures. The TEER values again showed donor-specific variations and did not deliver concluding results.

In summary, it was shown that the stimulation with 10 ng/mL IL-13 induces goblet cell hyperplasia in NHBE cells and that the extent can potentially be influenced by the duration of the stimulation. Since both tested lots of NHBE cells developed the hyperplastic change using the same concentration of IL-13, the reproducibility of the goblet cell hyperplasia in other lots under the same conditions is suspected.

Conclusions

Due to the fact that the prevalence of chronic airway diseases has increased in recent decades, studies on the function of inflamed airways has become more important. Inflammation of the airway epithelium impairs the barrier properties, including the mucociliary clearance and the formation of tight junctions, and thus alters the effects of inhaled substances. Accordingly, *in vitro* models simulating certain aspects of the airway epithelium under pathologic conditions could provide an important contribution for toxicological risk assessment.

Basal and goblet cell hyperplasia as well as squamous metaplasia are common pathologic features in chronic airway diseases and these lesions are described to play a critical role in the bronchial carcinogenesis ⁽²⁴⁸⁾.

In this thesis, the development of protocols for the induction of basal cell hyperplasia as well as goblet cell hyperplasia in NHBE cultures was achieved by applying special culture conditions. Thus, *in vitro* models simulating these pathologic conditions can be easily generated from "normal" NHBE cultures and used as a complement for toxicological evaluations of substances. Moreover, it was shown that squamous metaplasia might be inducible by the stimulation of the cells with appropriate concentrations of TGF-β1.

4.4 The CULTEX[®] Long Term Cultivation (LTC) System

The CULTEX[®] LTC system was recently developed by Cultex Laboratories GmbH in order to automate the cultivation of cell cultures grown on insert membranes at the air-liquid interface. The automated cultivation is a time-saving method and furthermore enables the standardization of cell culture conditions and accordingly the reproducibility of experiments. During this thesis, the suitability and usefulness of the CULTEX[®] LTC was tested with different cell lines of the human airways.

4.4.1 Material testing

In preliminary experiments, 16HBE14o- and A549 cells were cultivated in the prototype of the CULTEX[®] LTC. The cultivation system allows the control of the three insert positions individually, which means that the medium levels as well as the time intervals for the medium renewal and mixing can be adjusted in each chamber separately. For initial evaluation of the system and for determination of appropriate cell-specific conditions, the three parameters were varied arbitrarily for each insert position. For comparison, all runs were accompanied by control cultures of the respective cell line cultivated under conventional culture conditions (in receiver plates).

These preliminary approaches showed that none of the tested conditions produced viable cell cultures that were comparable to the control cultures. In situ microscopic analysis revealed that all A549 and 16HBE140- cultures that were cultivated in the CULTEX[®] LTC prototype for 7 - 20 days exhibited an uneven cell layer with large gaps between single cell clusters (Figure 4-75).



Figure 4-75: Inverse microscopical images of 16HBE14o- monocultures after a total cultivation time of 15 days. The cultures were either cultivated under conventional conditions with manual medium renewal (A) or in the CULTEX® LTC for 7 days with a computer-controlled medium renewal (B) every 48 h. The cultures that were cultivated in the LTC showed large gaps in the cell layer (B) while the control cultures had a normal, uniform cell layer (A).

Histological sections verified these observations. Here, it was shown that only few cells remained attached to the membranes while the control cultures showed a normal growth pattern and cell morphology. These results were frequently obtained with both A549 and

16HBE14o- cultures, even when the conditions in the LTC chambers were adjusted to those of the control cultures, with a medium renewal frequency of 48 h, a similar medium level and a suppression of the teflon membrane movement (see section 3.6.1). This led to the assumption that single material components might be the cause for the failure of the system rather than the cultivation process itself. To examine this hypothesis, all materials of the CULTEX® LTC that came into contact with the cell cultures or the medium during the cultivation were tested for cytotoxicity. For that purpose, an easy and efficient method for material testing was developed. In principal, this method is based on the "test by direct contact" which is described in the International Standard (ISO) 10993-5 for testing in vitro cytotoxicity of medical devices. Here, the cells are cultivated in direct contact to the test material. The main drawback of the original "test by direct contact" is that the test materials might have a physical impact on the cell cultures through their movement in the culture vessel. To overcome this, the samples were physically separated from the cell cultures in the newly developed method. This was enabled by placing the material samples onto the microporous membrane of a standard cell culture insert (6-well format) while the cells were grown in the wells of the receiver plate. In order to guarantee an unrestricted diffusion of all substances leaching from the material, the insert membranes were further perforated with a dissecting needle. The cell cultures were thus in contact with the test materials via the medium and their cytotoxic potential could be analyzed both qualitatively and quantitatively.

The qualitative analysis was performed by *in situ* microscopy and the quantitative by a commercial cell viability assay - either the CellTiter Blue Cell Viability Assay (Promega GmbH) or the WST-1 (Roche Diagnostics). The protocol for the material testing procedure is described in detail in section 6.1.4 and was published recently ⁽²⁴⁹⁾.



Figure 4-76: Procedure of biomaterial testing. A. The cells are seeded onto the bottom of 6-well receiver plates and cultivated for 24 h in their cell specific growth medium prior to the experiment. B. The membranes of commercial cell culture inserts (6-well format) are dissected with a needle. C. The test specimens are placed onto the perforated membrane and the cell culture inserts are applied into the wells of the receiver plate. D. The specimen is covered with cell culture medium and the cell cultures are thus cultivated in presence of the test sample for several days.

All experiments were accompanied by positive-, negative-, and blank controls. As positive control, a brass alloy (CuZn39Pb3) was used since brass, in general, is described to be cytotoxic ⁽²¹⁸⁾. Its cytotoxic potential was further verified in a preliminary test. The negative control consisted of silicone O-rings (peroxidically cross-linked; dimension: 3 mm x 1 mm) which are FDA (US Food and Drug Administration) approved and which have been proven to be non-cytotoxic in previous experiments. The wells of the blank controls were also equipped with a cell culture insert but without a test sample.

For test purposes, different cell lines were initially used (A549, IMR90 and 16HBE14o-). In case of a strong cytoxic potential of a test material, morphological alterations or a decrease in the number of cells were easily detectable in all tested cell lines. However, a moderate cytotoxic potential was only observable in IMR90 cells by a reduction of the cell size but not by a visible decrease of the cell number. Fibroblasts in general are characterized by a large cell size, a spindle-shaped morphology with long cell elongations and an orientated growth pattern. Small alterations of these typical characteristics are easily detectable by *in situ*

microscopy. Therefore, they are suitable cell models for the material testing and accordingly, the IMR90 cells were later further used for evaluating the materials for the CULTEX[®] LTC.

The *in situ* microscopic observations, in complement with the results obtained from the commercial cell viability assays, clearly demonstrated that some of the materials used in the CULTEX[®] LTC prototype had a cytotoxic potential, which was presumably the cause for the failure of the cultivation system. In order to select or preclude materials for the renewed construction of the CULTEX[®] LTC, all potential material candidates were thus tested for cytotoxicity according to the here described method before they were used.

4.4.2 Application of the CULTEX[®] Long Term Cultivation (LTC) System

The newly constructed CULTEX[®] LTC cultivation system was afterwards initially tested with Calu-3 and A549 cells. Here again, different variations of the medium level and time intervals for the medium renewal and mixing were tested. All runs were accompanied by control cultures of the respective cell line that were cultivated under the conventional conditions with a manual medium renewal every 48 h.

After seeding, all cultures were first cultivated in receiver plates under submerged conditions until they formed a confluent cell layer. Subsequently, the test cultures were placed into the CULTEX[®] LTC where they were cultivated at the air-liquid interface and supplied with basic differentiation medium according to the adjusted medium renewal frequencies. The basic experimental set-up is shown in Figure 4-77: The CULTEX[®] LTC modules were placed into an incubator with standard cultivation conditions (37 °C, 5 % CO₂, 95 % rH). The modules were connected to the medium supply flask and medium waste flask via silicone hoses and to the power supply via cables. The power supply was connected to the computer as well as to the pumps for the medium circulation. In order to guarantee the stability of the light and temperature sensitive ingredients of the culture medium, the medium bottle was placed into a lightproof thermo cover.



Figure 4-77: Experimental setup for the cultivation of cell cultures grown on insert membranes at the air-liquid interface in the computer-controlled CULTEX[®] Long Term Cultivation (LTC) system

The initial experiments with A549 and Calu-3 cells aimed on the evaluation of the CULTEX[®] LTC with regard to the practicability for the contamination-free cultivation of cell cultures under stable culture conditions. For that purpose the cells were cultivated for 7 to 18 days in the CULTEX[®] LTC.

It was shown that the cultivation under sterile conditions was enabled in the system. Moreover, the adjusted culture conditions could be maintained by the computer software during the entire cultivation time and small deviations from the original medium level were balanced immediately by pumping medium from or to the respective chamber.

Histological analysis of the cell cultures further demonstrated that the cultivation in the CULTEX[®] LTC in general enhanced the stability of the cell cultures. Furthermore, the growth pattern was greatly influenced by the adjusted conditions in the system. For example, the frequency of the teflon membrane movement, which enabled the mixing of the cell culture medium in the chambers and provided mechanical stress, was found to affect the

proliferation activity in both, A549 and Calu-3 cells. In A549, an increase in the number of cell layers was observed in correlation with an increase of the frequency of the teflon membrane movement (Figure 4-78).



Figure 4-78: A549 monocultures after a total cultivation time of 14 days. The cultures were either cultivated under conventional conditions with manual medium renewal every 48 h (A) or in the CULTEX[®] LTC with varying computer-controlled conditions (B-C). An increase of the mixing frequency from 60 min (B) to 24 min (C), while maintaining the medium renewal frequency at 36 h and the medium renewal volume at 1 mL, resulted in an enhanced proliferation and the formation of a stratified structure in the A549 cultures [HE stained histological sections; magnification: 400x (A-B) or 1000x (C)].

In Calu-3, the combination of both, mixing frequency and medium exchange was found to greatly influence the growth pattern in the cultures. Here, a normal phenotype with a monolayered growth was observed when the medium was exchanged every 24 h. The

columnar shape of the Calu-3 cells was more pronounced when the frequency of the teflon membrane movement was set to every 60 min compared to every 24 min or when no movement was applied. An increase of the time period between the medium exchange cycles from 24 to 48 h in combination with either omission of the movement or a frequency of 24 min resulted in an unordered "tumorous" cell growth with a multi-layered structure. Interestingly, an almost normal growth pattern was observed when the movement frequency was set to 60 min while maintaining the medium exchange rate at 48 h (Figure 4-79).



Figure 4-79: Calu-3 cells cultivated in two CULTEX[®] LTC systems. In each system, the medium renewal frequency was constant in all three insert positions: A - C: every 24 h; D - F: every 48 h. The stirring frequency was varied in all chambers: A + D: no stirring; B + E: every 30 min; C - F: every 60 min. The increase of the medium renewal frequency from 24 h to 48 h resulted in an enhanced proliferation. A stirring frequency of 60 min was found to promote a normal monolayered growth and the columnar shape of the Calu-3 cells (C + F).

Conclusions

The practicability of cultivating cell cultures on insert membranes at the air-liquid interface under sterile and stable conditions using the computer-controlled CULTEX[®] LTC was proven. Moreover, it was shown that the dynamic medium flow, in general, had a positive effect on the cells and that the phenotype of the cultures could be influenced by modifying the frequencies of the medium exchange and teflon membrane movement. The desired phenotype can thus be selected after testing different modifications and reproduced by applying the same conditions. However, it was shown, that there are cell line-specific responses to different alterations and therefore, the optimal conditions have to be adjusted to the respective cell line individually.

The presented computer-controlled system might be of great value especially for the cultivation of NHBE cells, since high variances between inserts of the same culture were observed with the conventional handling procedure. These variations are potentially related to small deviations in the culture conditions, like the medium level during the ALI cultivation. Since the reliability and reproducibility of data is a major requirement for toxicological studies, the observed variances are highly disadvantages.

The developed techniques enabling the regulation of the adjusted culture conditions in the CULTEX® LTC, like the ultrasonic sensor and the special software, could later also be used for the development of more sophisticated systems that allow the cultivation of a higher number of cultures under identical and stable conditions. Such systems would be highly useful in order to increase the throughput of toxicological studies while ensuring the reproducibility of the data.

5 Summary and Final Conclusions

The overall aim of this thesis was the development of suitable *in vitro* models of the bronchial epithelium that can universally be used for toxicological studies. Critical factors for appropriate cell culture models are the presence of important differentiation characteristics of bronchial epithelial cells, including the secretion of the main airway mucins 5AC and 5B, the development of cilia and the formation of functional tight junctions.

Long-term cultivation is often required for toxicological studies, since certain questions require the long-term or repeated exposure to the test substances in order to provide "normal" exposure conditions. For example, to mimic the cigarette smoke exposure of the airways in smokers in a near-realistic manner, a repeated exposure of the *in vitro* models over a certain time is required. Therefore, the stability and viability of the cultures during the study period are additional requirements for suitable cell models.

Normal bronchial epithelial cells isolated from human lung tissue (NHBE cells) were shown to provide the most reliable cell model of the airway epithelium since they generally retained the ability to develop all important differentiation characteristics of the bronchial epithelium *in vivo*. In addition, the long shelf life of the differentiated cultures also recommends them for long-term studies. A protocol for the isolation of epithelial cells from cryopreserved bronchial tissue was developed in this thesis which enables the flexible on-demand provision of NHBE cells. However, due to the fact that only cells of the passages one and two are generally able to show the desired differentiation patterns and because their storage in liquid nitrogen was found to be unfavorable, the availability of these cells is highly limited. The occurrence of donor-specific variations further restricts their use for large-scale and repeated experiments.

The shortage of available cells might be overcome by using continuous cell lines, since they can be cultivated, in general, over many passages without impairment of their cell viability and differentiation ability. However, the studies presented here revealed that these cell lines mostly lack natural differentiation patterns and are thus not comparable to their normal *in vivo*-counterparts. Only the Calu-3 cells shared many important features with differentiated normal bronchial epithelial cells, but here limitations in the shelf life of the cultures were observed, restricting their suitability for the long-term cultivation.

145

Cancer cell lines generally bear the risk of showing tumor-specific characteristics but they are nevertheless mostly used in current toxicological studies since those cell lines derived from normal cells are not widely available. To overcome this situation, a few cell lines have been developed by genetic manipulation (immortalization) of normal human epithelial cells (e.g. 16HBE14o-). Unfortunately, these cell lines are often licensed and can thus only be used with certain restrictions. Moreover, comparable to cancer-derived cell lines, they mostly lack some important features of the normal bronchial epithelium. The development of new immortalized cell lines exhibiting at least the most important characteristics for targetorientated toxicological assays is therefore highly desirable. In this thesis it was shown that the immortalization of normal bronchial epithelial cells can be achieved by transformation with the viral gene HPV E6/E7. However, the differentiation ability of these cells was limited. Besides the selected gene and vector for the immortalization, there are other possible reasons for the failure of the cells to develop natural differentiation characteristics, like the use of an inappropriate cell source. For example, cryopreservation was often shown to cause damage to cultured cells and thus, in general, it is recommended to only use freshly isolated, non-cryopreserved cells of the first two passages for the immortalization procedure. In summary, no final conclusions on the suitability of the transformation of normal human bronchial epithelial cells with the HPV E6/E7 gene for the generation of stable, continuous cell lines with natural differentiation patterns could be drawn in this thesis. But, it is worth making more efforts to develop suitable strategies for the immortalization of normal human airway epithelial cells, in order to be able to overcome the shortage of appropriate continuous cell lines and available primary cells for routine toxicological assessments in vitro.

In vivo, bronchial epithelial cells and fibroblasts form a unit - the *Epithelial Mesenchymal Trophic Unit* - which is characterized by complex interactions between the two cell types during growth, differentiation, repair and remodeling of the epithelium. To better reflect these conditions, *in vitro* models of the lung epithelium should ideally include both cell types. However, the establishment of such models is still a major challenge. In this work two different 3D *in vitro* models of the bronchial epithelium, including epithelial cells and fibroblasts were established - the Sandwich Cell Model (SCM) and Compartment Cell Model (CCM). It was shown that both co-culture models have certain advantages and disadvantages. The SCM provides a good reflection of the structure of the lining epithelium in the bronchi and bronchioles and it might be valuable for studying remodeling aspects of the epithelium after inflammation processes. However, contraction and degradation of the collagen matrix as well as limitations in the differentiation ability of NHBE cells in this co-culture model restrict its suitability for toxicological assessments.

In the CCM, the epithelial cells and fibroblasts are spatially separated but interactions between the two cell types are enabled through secreted factors that are distributed via the cell culture medium. The original CCM, based on the protocol developed by Pohl and colleagues (174), was shown to be unsuited for the long-term cultivation and was thus excluded from further studies. As a substitute, two different alternatives, the CCM variant I and variant II, were developed. In these models, the fibroblasts are grown in wells of receiver plates which are equipped with inserts carrying the epithelial cells. In contrast to the CCM variant I, the CCM variant II additionally includes an ECM component (collagen type I matrix) and therefore provides a more natural environment for the cells. The practicability of this co-culture model was proven in this study and moreover, clear interactions of the two cell types were verified in these cultures. In summary, the development of a realistic reproduction of the bronchial epithelium, which further includes natural interactions of epithelial cells and fibroblasts, was achieved in this thesis. The epithelial component can be represented by various cell lines, depending on the respective requirements. The easy generation and handling recommends the CCM variant II as universally applicable in vitro model of the bronchial epithelium for a variety of studies.

The increasing prevalence of chronic lung diseases worldwide demands the inclusions of *in vitro* models showing common types of pathologic conditions for toxicological risk assessment. In this thesis, it was shown that certain pathologic features that are often found in inflamed airways can be induced in cultures of NHBE cells by stimulation with certain growth factors or cytokines. For example, the development of protocols for the induction of goblet cell and basal cell hyperplasia was achieved. These cultures could provide valuable *in vitro* models for studying the effects of various substances in the airways of patients with chronic airway diseases, like asthma or COPD.

Besides the selection of suitable cell models, another important requirement for toxicological risk assessment is the reliability of the data. For that purpose an appropriate

147

number of samples and repetitions of the experiments is necessary. However, it was shown that especially in cultures of NHBE cells, high variations between individual inserts were observed with regard to cell morphology, TEER values and IL-6 or IL-8 secretion under conventional culture conditions. One possible reason for this variability might be the occurrence of slight differences in the medium level caused by inaccurate pipetting. In order to provide stable and reproducible culture conditions, automated cultivation systems are valuable. The CULTEX[®] LTC was specifically designed for the computer-controlled cultivation of cell cultures grown on inserts at the air-liquid interface. The here presented data revealed that the cultivation in the CULTEX® LTC under sterile and stable conditions was successful and, moreover, that the frequency of medium renewal and mixing greatly influenced the cell culture phenotype. The special technical features of the CULTEX® LTC which enable the computer-controlled regulation of the adjusted culture conditions and the detection of slight deviations of the medium level, like the ultrasonic sensor or the special software, could be adopted for the design of future generations of such automated systems that allow the cultivation of high numbers of cell culture inserts at the same time. Thus, toxicological studies could be greatly enhanced with regard to throughput and reproducibility of the data.

In conclusion, the results presented in this thesis demonstrated that there is currently no ideal cell model of the bronchial epithelium available. Thus, the necessity of developing new continuous lung epithelial cell lines was highlighted, in order to overcome the disadvantages of the existing ones. At present, cultures of isolated normal human bronchial epithelial cells of the first two passages provide the most realistic models with regard to natural differentiation patterns and the ability to develop common types of pathologic conditions *in vitro*. For studies that require high throughput, the Calu-3 cells might be a good alternative, since they share many features of normal bronchial epithelial cells and, moreover, they can be subcultivated over many passages and are thus permanently available. Unfortunately, they are not suitable for long-term exposure studies and cancer-specific characteristics cannot be precluded. Investigations on repair and remodeling processes in the airway epithelium *in vitro* ideally require the inclusion of natural interactions of the EMTU. For that purpose, the here developed co-culture model - CCM *variant II* – is highly suitable.

The here provided information and protocols are useful for the design and implementation of *in vitro* studies on lung toxicity and can therefore make a valuable contribution for the reduction of animal experiments in this field.

6 Appendix

6.1 Methods

6.1.1 Cultivation and handling of cell cultures

In general, the different cell lines were handled separately in the laminar flow hood to avoid cross-contamination. All cell culture works were performed according to the principles of sterile working procedures until fixation of the cultures with formaldehyde.

6.1.1.1 Cell culture media

Table 6-1 provides an overview of the cell line specific growth media and special additional ingredients or supplements. The compositions of the basic differentiation medium and the media used for the isolation of NHBE cells from lung tissue are listed below the table.

For the stimulation experiments with NHBE cultures, the basic differentiation medium was modified by adding the respective growth factors or cytokines in the mentioned concentrations (see section 4.3).

Cell line	Basal cell culture medium	Order information	Additional ingredients included	Supplements
A549	DMEM	Biochrom AG, FG 0415	stable L-glutamine, 3,7 g/L NaHCO ₃	10 % FBS
NHLF	RPMI 1640	Life Techologies Inc, 31870025	2 g/L NaHCO ₃	10 % FBS 2mM L-glutamine 1mM sodium pyruvate
IMR90, Calu-3	MEM Eagle with Earle´s salts (EMEM)	Lonza, BE12-125F		10 % FBS 2mM L-glutamine 1mM sodium pyruvate
16HBE14o-	MEM Earle's (MEM)	Biochrom AG, FG 0325	stable L-glutamine, 3,7 g/L NaHCO ₃	10 % FBS
NHBE	BEGM (BEBM + SingleQuots®)	Lonza, CC-4175	SingleQuots , including: Hydrocortisone Insulin Transferrin Epinephrin Triiodothyronin Human epidermal growth factor (hEGF) Bovine pituitary extract (BPE) Gentamicin/amphotericin B (GA) Retinoic acid (RA)	
НВЕрС	Airway Epithelial Cell Growth Medium Kit	Promocell GmbH, C-21160	Insulin Transferrin Epinephrin Triiodothyronin Human epidermal growth factor (hEGF) Bovine pituitary extract (BPE) Retinoic acid (RA)	

Table 6-1: Overview of cell line-specific growth media, including special ingredients or additional supplements

Order information for supplements:

- Fetal bovine serum (Lot 0310L; Biochrom AG, S0210)
- L-glutamine (200 mM, Gibco; Life Technologies Inc., 25030-024)
- Sodium pyruvate (100 mM, Gibco; Life Technologies Inc., 11360-039)

Basic differentiation medium:

Ingredients:

- BEGM[™] Bulletkit[™] (Lonza, order no.: CC-3170)
- DMEM (Gibco[®], Life Technologies Corp., order no.: 22320)
- CaCl₂ (Merck KGaA, order no.: 1.02382)
- Retinyl acetate-water soluble (Sigma-Aldrich; order no.: R0635)

• Ultroser[®] G (Cytogen GmbH, order no.: 15950-017)

500 mL of the basic differentiation medium were prepared as follows:

BEBM/DMEM (Gibco) 1:1	
BEBM	240 mL
DMEM	240 mL
+ BEBM Single Quots [®] (w/o RA)	
+ 2 % Ultroser® G	10 mL
+ 1 mM CaCl ₂ (10 % solution)	0.735 mL

The retinoic acid (RA) "SingleQuot" was not added but replaced by a homemade retinol stock solution with a concentration 100 nM, prepared from water soluble retinyl acetate. For CCM *variant II* co-cultures, the addition of CaCl₂ was omitted, since thus the turbidity of the collagen matrix could be prevented.

Incubation Medium for Isolation of Bronchial Epithelial Cells from human tissue

Ingredients:

- MEM Eagle (Joklik Modification; Sigma-Aldrich, M8028)
- Amphotericin B solution (250 µg/ml; Sigma-Aldrich, A2942)
- Ceftazidim (Sigma-Aldrich, C3809)
- DL-Dithiothreitol (DTT; Sigma-Aldrich, D0632)
- DNase I (from bovine pancreas; Sigma-Aldrich, DN25)
- Gentamycin (10 mg/ml; Biochrom AG, A2712)
- Penicillin/streptomycin (10000 U/10000 μg/ml; Biochrom AG, A2213)
- Tobramycin (Sigma-Aldrich, T4014)
- Vancomycin (Sigma-Aldrich, V2002)

100 mL were prepared as follows:

Tobramycin	4 mg
Ceftazidim	5 mg
Vancomycin	5 mg
DNase	1 mg
DTT	50 mg

Each powder was solved in 5 - 10 mL of MEM Eagle (Joklik Modification). The solutions were added to 50 - 75 mL of MEM Eagle (Joklik Modification; final volume: 100 ml) and sterilized by filtration through a 0.2 µm filter. The medium was stored at 4 °C up to six weeks.

The following antibiotics were added shortly before use:

Penicillin/streptomycin solution	0.5 mL
Amphotericin B solution	1 mL

Digestion Medium for Isolation of Bronchial Epithelial Cells from human tissue

Ingredients:

- Same as for incubation medium
- Pronase XIV (Sigma-Aldrich, P5147)

50 mL were prepared as follows:

Incubation medium (see above)	50 mL
Pronase XIV	50 mg

The solution was vortexed thoroughly and sterilized by filtration through a 0.2 μ m filter. The medium was prepared shortly before use.

Dissociation Medium for rescuing isolated Bronchial Epithelial Cells from clumps

Ingredients:

- DL-Dithiothreitol (DTT; Sigma-Aldrich, D0632)
- DNase I (from bovine pancreas; Sigma-Aldrich, DN25)
- Collagenase (Life Technologies Inc., 17018-029)
- Ethylenediaminetetraacetic acid (EDTA; Sigma-Aldrich, ED4SS)

20 mL were prepared as follows:

Collagenase	5 mg
DNase	0.2 mg
DTT	10 mg
EDTA	11.7 mg

Each powder was solved in 5 ml of MEM Eagle (Joklik Modification). The solutions were mixed and the final solution was sterilized by filtration through a 0.2 μ m filter. The medium was prepared shortly before use.

6.1.1.2 Subcultivation (passaging) of cells

Adherent growing cells, like epithelial cells and fibroblasts, are normally in close contact with their substrate. For subcultivation or for the preparation of cells for experiments (e.g. seeding on cell culture inserts), the cells have to be detached from their substrate. This can be realized by means of a trypsin solution, complemented with EDTA. The enzyme trypsin breaks connections of adhesion proteins between cells, and between cells and the contact substrate. Bivalent ions inhibit trypsin. For this reason, the trypsin solution also contains EDTA, a chelator for bivalent ions (especially Ca²⁺).

Long trypsin incubation times can cause irreversible damage to the cells. To avoid this, the cells should be monitored carefully during the incubation with trypsin and the inhibition of the enzyme activity should be performed immediately after the majority of the cells was detached from the culture vessel. The enzyme can be inhibited by serum-containing

medium, since sera (like FBS) contain trypsin inhibitors. For serum-free cultivation, a trypsin inhibitor isolated from soybean can be used for stopping the activity of trypsin.

Before starting, all solutions and media in appropriate volumes as well as cultureware products (cell culture flasks or inserts) that were required for the subcultivation or for the preparation of experiments were prewarmed at 37 °C in a waterbath or incubator.

The passaging was performed according to the following procedure:

After removal of the cell culture medium from the culture flask, the cell layer was rinsed thoroughly with sterile PBS (w/o Mg2+ and Ca2+; 8 mL for 75 cm2 and 4 mL for 25 cm2 cell culture flasks) and the PBS was afterwards removed. Subsequently, the trypsin/ EDTA solution was applied onto the cells and either incubated at 37 °C (epithelial cells) or at room temperature (fibroblasts). The trypsin concentration as well as the duration of the incubation was varied according to the requirements of the respective cell line. The detachment of the cells was monitored carefully by means of an inverted microscope. The trypsin activity was stopped when the majority of cells was detached from the culture vessel by adding an appropriate volume of either serum-containing or trypsin inhibitor-containing medium. Table 6-2 gives an overview of the concentrations and volumes of the trypsin activity for each cell line.

The cell suspension was transferred to a 50 mL centrifugation tube and 100 µL were used for the determination of the number of viable cells per mL with the CASY Cell Counter and Analyzer System. The CASY Model TT Cell Counter and Analyzer system (Roche Diagnostics GmbH) uses the "Electrical Current Exclusion" method for measuring cell viability of mammalian cell lines. This method depends on the membrane integrity of the cells. The electrical barrier of the intact cell membrane is higher than in dead cells and therefore, viable cells cause an increased level of resistance in the measuring pore of the system. Thus, the number of viable cells in a defined volume of the cell suspension can be determined.

Depending on the number of viable cells per mL, the cell suspension was either directly used for inoculation of new culture flasks or cell culture inserts or resuspended in a smaller medium volume after centrifugation, in order to increase the cell concentration per mL. In general, it is recommended to use only small volumes of the cell suspension which still contains residual trypsin and to dilute it in a ratio of at least 1:10 with cell culture medium. Centrifugation was performed for 7 min at 220 x g and after removing the supernatant, the cell pellet was resuspended in an appropriate volume of cell culture medium to obtain the calculated cell concentration per mL. The number of viable cells was determined by the CASY system. The prewarmed cell culture flasks containing an appropriate volume of cell culture medium were inoculated with the desired number of cells by pipetting the respective volume of the cell suspension into the flasks. Before seeding the cells on cell culture inserts, a volume of 1200 μ L of cell culture medium was pipetted into the wells of the receiver plate. The appropriate volume of the cell suspension containing the desired number of cells was subsequently applied onto the membrane and diluted with cell culture medium to a final volume of 500 - 600 μ L.

Cell line	Cell culture flask [cm ²]	Trysin/ EDTA (concentration, volume)	Trypsin reaction stopped with
NHBE/ HBEpC	75	0.05 %/ 0.02 % (w/v), 2 mL	2 mL trypsin inhibitor from soy bean (1x) with 4 mL BEGM
NHBE/ HBEpC	25	0.05 %/ 0.02 % (w/v), 1 mL	1 mL trypsin inhibitor from soy bean (1x) with 4 mL BEGM
Calu-3	75	0.25 %/ 0.02 % (w/v), 4mL	8 mL EMEM incl. 10 % FBS
16HBE14o-	75	0.05 %/ 0.02 % (w/v), 3 mL	8 mL MEM incl. 10 % FBS
ChaGo-K1	75	0.05 %/ 0.02 % (w/v), 2 mL	8 mL RPMI incl. 10 % FBS
IMR90	75	0.05 %/ 0.02 % (w/v), 2 mL	8 mL EMEM incl. 10 % FBS
NHLF	75	0.05 %/ 0.02 % (w/v), 2 mL	8 mL RPMI incl. 10 % FBS

Table 6-2: Information on the trypsin/ EDTA concentrations and volumes as well as the tryspin stopping solutions used for the subcultivation of different cell lines

6.1.1.3 Cryopreservation of cells and tissue

For the storage of cells in the gaseous phase of liquid nitrogen, for example to guarantee the availability of viable cells for future experiments, they have to be transferred to a specialized medium which contains a cryoprotectant agent. Mostly, DMSO is used for this purpose. DMSO is cytotoxic at room temperature and therefore all steps have to be performed on ice

and the freezing should be started as soon as possible after the addition of Dimethyl sulphoxide (DMSO). Mostly the ready-to-use "Recovery Cell Culture Freezing Medium" (Gibco) obtained from Life Technologies Corp. (12648-010) was used. Occasionally, the freezing medium was self-prepared as follows: 80 % DMEM (including L-glutamine, sodium pyruvate and HEPES) was mixed with 10 % DMSO (cell culture tested; Sigma-Aldrich) and 10 % FBS.

To avoid lethal intracellular freezing, the freezing down of cells was performed stepwise, in defined rates, by using a freezing container ("Mr. Frosty", Nalgene), which provides a repeatable cooling rate of 1 °C/ min. As soon as a temperature of -80 °C was achieved, the frozen cells were transferred to a cryogenic tank.

Before starting, the required number of cryovials was labelled (description of cells, passage number, date). The freezing container (Mr Frosty) was filled up with isopropanol to the filling mark. The cryovials were placed into the "Mr Frosty" and stored at 4 °C for 1 h before use.

The adherent cells were trypsinized according to the cell-specific protocol (see 6.1.1.2) and the number of viable cells was determined (by the CASY[™] Cell Counter + Analyzer). If required, a new culture flask was inoculated with the appropriate cell number before continuing with the next step.

The cell suspension was centrifuged for 7 min at 220 x g and 18 °C and meanwhile, the cooled and labeled cryovials were place on the clean bench. After centrifugation, the culture medium was discarded and the cell pellet was resuspended in cell culture freezing medium. The volume of freezing medium was adjusted to obtain the desired cell density (optimally between 2 - 3×10^6 cells/mL).

Each cryovial was filled with 0.5 mL of the cell suspension (if a large number of vials was prepared, the cell suspension was occasionally swirled to guarantee a uniform distribution of the cells within the suspension). The vials were quickly placed into the Freezing Container and stored at -80 °C overnight, before they were transferred to the cryogenic tank.

6.1.1.4 Coating of cell culture insert membranes or culture vessels

In order to provide a realistic environment for cells *in vitro*, the surfaces on which they are grown are often coated with proteins of the ECM. The nature of the surface coating plays an important role in their ability to attach, proliferate and differentiate ⁽²⁵⁰⁾. Commonly used

ECM components for surface treatment are collagen type IV, collagen type I and human fibronectin.

Type IV collagen is a ubiquitous component in basement membranes and provides the major structural support for this matrix. Especially epithelial cells show an enhanced attachment and growth when cultivated on surfaces coated with this ECM component.

Collagen type I is found in most tissues and organs. The surfaces coating with collagen I is most commonly used and it was shown that it promotes the growth and differentiation of many different cell types.

Human fibronectin is a glycoprotein that binds to integrins or ECM proteins, like collagens. It plays a major role in the attachment, growth and differentiation of cells *in vivo* and is therefore often used as a substrate for cell cultures.

In this thesis, varying surface coatings were used for the different epithelial cell lines. For the Calu-3, ChaGo-K1 and occasionally NHBE cells, the cell culture insert membranes were coated with 6.5 μ g/ cm² collagen type I. For that purpose, collagen type I from bovine skin with a concentration of 3 mg/mL (aqueous solution in 0.01 M HCl) was diluted in a ratio of 1:60 with 0.01 HCl and 130 μ L of this solution were applied on each insert membrane (12-well format; membrane diameter: approx. 1 cm²). The inserts were incubated at 37 °C for at least 1 h (preferably overnight). Afterwards, the membranes were rinsed twice with 200 μ L of sterile PBS and treated with UV for 2 h. If not immediately used, the inserts were stored at 2 - 8 °C for up to six weeks.

The collagen type IV coating (10 μ g/ cm²) was used for the cultivation of NHBE cells and HBEpC on insert membranes. A stock solution of 1 mg/mL was prepared with 0.5 M HCl. For each insert, 10 μ l of this stock solution were mixed with 140 μ L of ultrapure water (ddH₂O) and applied on the membrane. The incubation and storage was performed according to the same protocol as for the collagen type I coating.

A coating made up of a mixture of collagen type I and fibronectin was used for the following purposes: (1) Coating of culture flasks (75 cm²) and insert membranes for the cultivation of 16HBE14o- cells, (2) coating of 6-well plates for the cultivation of NHBE cells directly after the isolation from tissue and (3) coating of cell culture flasks (25 cm²) for the cultivation of primary NHBE cells after the first passaging.

The coating solution was prepared as follows:

For 100 mL:

10 mL of a BSA solution with a concentration of 1 mg/mL in PBS was prepared, filtered through a sterile filter (0.2 μ m pores) and added to 88 mL of cell culture medium [optimally LHC Basal Medium is used (Life Technologies Inc., order no. 12677-019)]. Each 1 mL of collagen type I (3 mg/mL) and fibronectin (1 mg/mL) were added. The solution was gently mixed and applied in the following volumes:

- 150 μL on insert membranes (12-well format; approx. 1 cm²)
- 3 mL per 75 cm² culture flask
- 2 mL per 25 cm² culture flask
- 1 mL per well (6-well plate)

The incubation and storage is performed as for collagen I and IV coated culture vessels.

6.1.2 Preparation of co- cultures according to the Sandwich Cell Model (SCM)

In the SCM, airway epithelial cells are co-cultivated with lung fibroblasts to enable natural interactions between the two cell types comparable to the *in vivo* conditions. The SCM cultures are basically prepared as follows: the epithelial cells (e.g. NHBE or Calu-3) are seeded on top of a collagen type I matrix in which fibroblasts are embedded.

For the polymerization of the collagen type I matrix, the collagen solution had to be brought to a neutral pH. This was achieved by adding a NaOH-containing buffer solution which was prepared as follows:

100 mL of 0.05N NaOH (e.g. 5 mL of 1N NaOH diluted to 100 mL with dH_2O)

- + 2.2 g NaHCO₃
- + 200mM HEPES

The solution was autoclaved and stored at 4 °C for up to 6 months.

All solutions and media for the trypsinization and – if necessary – passaging of the fibroblasts as well as the required materials for cell seeding, including the appropriate number of cell culture inserts and culture flasks, were prewarmed at 37 °C. The trypsinization was

performed as described in section 6.1.1.2. After estimating the number of viable cells per mL by means of the CASY system, the fibroblast solution was centrifuged for 7 min at 220 x g. The culture medium was afterwards removed and the cell pellet was resuspended in a calculated volume of cell culture medium, to obtain a concentration of approximately 1.0 or 2.0×10^6 cells/mL. The new culture flask was inoculated with the desired cell density.

The collagen matrix in which the fibroblasts are embedded was prepared shortly before use and stored on ice. Only small volumes of the collagen solution (up to 2.5 mL) were prepared. If higher volumes were required, the procedure was repeated.

The collagen/fibroblast layer was prepared as follows:

	4 parts Cellmatrix I-A	(e.g. 400 μL)
+	4 parts Cellmatrix I-P	(e.g. 400 μL)
+	1 part 10x MEM (or 10x DMEM)	(e.g. 100 μL)

The solution was mixed thouroughly.

+	1 part of the NaOH buffer solution	(e.g. 100 μL)
---	------------------------------------	---------------

The solution was mixed thouroughly and bubbling was avoided. A shift of the pH towards neutral was indicated by a change of the phenol red indicator in the 10x MEM from orange to pink colour.

Subsequently $100 - 150 \mu$ L of the fibroblast suspension were added to the collagen solution. For 1 mL of the collagen solution, 100μ L of the fibroblasts suspension containing 1.0×10^6 cells/mL were used. For 2 mL, the cell concentration was increased to approximately 2.0×10^6 by using a reduced amount of the culture medium for resuspension of the cell pellet after centrifugation. Of this cell suspension, 100μ L were added to 2 mL collagen solution

A final volume of 200 – 250 μ L of this collagen/fibroblast solution was quickly applied onto the insert membrane (12-well; 0.4 μ m pores) and incubated at 37 °C for 3 - 5 h.

All solutions and media for the trypsinization of the epithelial cells and - if required – new cell culture flasks were prewarmed at 37 °C. The trypsinization was performed as described in section 6.1.1.2. The number of viable cells was determined with the CASY system and the appropriate volume of the cell suspension containing the desired cell number was applied
onto the polymerized collagen matrix. In case of low cell densities in the suspension, the number of cells per mL was increased by centrifugation, removal of the supernatant and resuspension in a reduced volume of cell culture medium. To reach a final volume of 550 μ L on the apical side of the membrane, an appropriate volume of cell culture medium was added. The final medium volume in the wells was 1200 μ L. The cultures were provided with the specific growth medium of the respective epithelial cells line and cultivated under submerged conditions. As soon as the epithelial cells formed a confluent layer, the cultures were provided with differentiation medium and cultivated at the air-liquid interface.

6.1.3 Preparation of co-cultures according to the Compartment Cell Model (CCM) *variant II*

In the co-culture model CCM variant II, the fibroblasts are embedded into a matrix composed of collagen type I and grown in the wells of insert receiver plates. The epithelial cells are cultivated on conventional cell culture inserts (12-well; 0.4 μ m pores) whose membranes were previously coated with collagen type I or IV (see 6.1.1.4). After a short growth phase, the insert were placed into the fibroblast-containing wells and thus a direct contact between the two cell types was provided through the distribution of secretion products via the culture medium.

The passaging of the epithelial cells and fibroblasts was performed according to the procedure described in section 6.1.1.2. The epithelial cells were seeded on the insert membranes with the desired seeding density. The inserts were prewarmed before use. The final medium volume for the submerged cultivation was 550 μ L on the apical side and 1200 μ L on the basal side (in the wells).

The collagen type I matrix with embedded fibroblasts was prepared according to the same protocol as for SCM cultures (section 6.1.2) and 270 μ L of the solution was applied in wells of a new 12-well receiver plate. After an incubation of 3 h, 500 μ L of the fibroblast-specific growth medium was applied onto the hardened collagen matrix.

The epithelial cells and fibroblasts were cultivated separately until the epithelial cells nearly formed a confluent cell layer. The inserts were subsequently placed into the fibroblastcontaining wells and the co-cultures were supplied with differentiation medium under submerged conditions for one day. The inserts were then raised to the air-liquid interface and supplied with 580 μ L of differentiation medium that was applied onto the collagen/fibroblast layer in the wells. Care was taken during the renewal of the medium to avoid damage of the collagen matrix.

Histological section were only prepared from the epithelial cells. The determination of the TEER values in the epithelial cell layers was performed in the absence of the fibroblasts by placing the inserts in new receiver plates.

6.1.4 Material testing

The test materials were provided by Halter Engineering GmbH (Herisau, Switzerland), who was the constructor of the CULTEX[®] LTC. The size of the test materials was adjusted to fit the 6-well cell culture inserts. The samples were rinsed in 70 % ethanol, placed into glass Petri dishes and autoclaved (121 °C, 20 min). Afterwards they were placed on the membranes of cell culture inserts which had previously been perforated with a dissecting needle (approx. 1 mm in diameter), in order to enable the uniform mixing of the cell culture medium on the apical and basal side of the membrane. The cell culture inserts were then set in the wells of the companion plate carrying the cells that had been seeded the day before. The seeding density was 1 x 10^5 (A549) or 1.5×10^5 (IMR90, 16HBE140-), respectively, per well (6-well format). The cultures were supplied with 3 mL of their cell line- specific growth medium which was renewed before the inserts with the test samples were placed into the wells. The samples were covered with 1 mL medium and thus, the total volume of medium per well was 4 mL. All experiments were accompanied by control cultures which were prepared in the same way as the test cultures but without a test sample on the insert membrane.

The cells were cultivated in presence of the test materials for 3 days or until the control cultures reached approx. 80 % confluency. After removing the inserts with the samples, the *in situ* microscopic analysis was performed immediately. Then, the qualitative analysis of cytotoxicity was conducted by either using the resazurin-based CellTiter-Blue[®]Cell Viability Assay (Promega Corp.) or the formazan-based WST-1 (Roche Diagnostics).

The fluorometric CellTiter-Blue Cell Viability Assay uses the indicator dye resazurin to measure the metabolic capacity of the cells. Viable cells retain the ability to reduce the non-fluorescent resazurin into the highly fluorescent resorufin. The cell culture medium was replaced by 240 μ L CellTiter-Blue solution diluted in 1.28 mL fresh serum-free medium per 162

well. After an incubation time of 2 h, the optical density was read on a multiwell microtiter plate photospectrometer (Appliskan, Thermo Scientific) at a wavelength of 570 nm.

The colorimetric WST-1 assay is used to determine the ratio of proliferating cells through reduction of a water-soluble tetrazolium salt (WST) to formazan by electron transport across the plasma membrane of dividing cells. Thus, the absorbance of the formazan dye is proportional to the number of metabolically active cells. The cell culture medium in the wells was replaced by 125 μ L of the WST-1 solution diluted in 1.375 mL fresh serum-free medium. After an incubation time of 1 h at 37 °C, the absorbance was determined by means of a microplate spectrometer (Spectra Max 340 PC, Molecular Devices) at 450 nm.

6.1.5 Analytical methods

6.1.5.1 Histology

The paraffin embedding of cell cultures that were grown on microporous insert membranes enables the generation of histological sections. Thus, the growth pattern and differentiation status of the cells can be analyzed by staining the sections, for example with haematoxylineosin (HE) staining, or by immunohistochemistry.

The cells were first fixed with 10 % formaldehyde (each 1 mL on the basal and apical side) for 1 h at room temperature. The samples were then rinsed in tap water for 30 min (each 1 mL on the apical and basal side).

The inserts were subsequently dehydrated by applying increasing alcohol concentrations (each 1 mL on the apical and basal side) in the following order:

- (1) 10 min 30 % ethanol
- (2) 10 min 50 % ethanol
- (3) 10 min 70 % ethanol
- (4) 10 min 95 % ethanol
- (5) 10 min absolute ethanol
- (6) 10 min absolute ethanol
- (7) 10 min Roticlear[®] (Carl Roth)
- (8) 10 min Roticlear[®]

Subsequently the paraffin infiltration was performed as follows: The inserts were transferred to a receiver plate, covered with liquid paraffin from the apical and basal side and incubated for 2 h at 60 °C in a heating cabinet.

The paraffin was renewed after 1 h and the inserts were incubated for another hour at 60 °C. The excessive paraffin was afterwards removed and the paraffin-coating of the inserts was allowed to cool down by incubating the inserts upside down at room temperature.

The membrane was then isolated carefully from the plastic housing of the insert with a scalpel.

Prior to the paraffin embedding, the following preparations were made: Each two braces of conventional embedding cassettes were removed with a cable cutter. The cassettes were labelled with the respective culture name or number with a pencil and afterwards placed into metal embedding moulds.

The paraffin embedding was performed as follows: The embedding moulds were filled with liquid paraffin and the membranes were positioned into the prepared gap of the embedding cassette by means of a pair of self-spanning tweezers. The tweezers were fixed, for example by placing them on a staple of receiver plate lids, in order to keep the membrane in the right position within the paraffin during hardening. The embedding moulds were left on the heating plate until the thin paraffin-coating of the membranes was melted and mixed with the liquid paraffin in the moulds. Subsequently, the embedding moulds were placed on a small cooling plate for a minimum of 20 min for cooling down, before they were transferred to another cooling plate which reaches temperatures of -5 °C, for final hardening. During the hardening, the membranes remained fixed in their position by the tweezers.

Prior to the generation of histological sections, the tweezers were removed and the paraffin blocks were detached from the embedding mould. The paraffin blocks were trimmed with a microtome in 50 μ m steps until reaching the desired cutting level of the membrane. Sections of each 5 – 7 μ m were then generated from the paraffin block containing cross sections of the embedded cell culture. These sections were transferred to a 40 °C tissue floating waterbath for stretching and placed on silanized glass slides for HE staining or poly-l-lysine coated glass slides for immunohistochemistry.

The sections on glass slides were allowed to dry by placing them on the warm edge of the tissue floating waterbath for a minimum of 1 h. Afterwards they were either stored at room temperature overnight or incubated for 30 min in a heating cabinet at 60 °C before use.

The hematoxylin-eosin staining was routinely used for visualization of the cells in the histological sections. Hematoxylin stains basophilic structures bluish-purple. The nucleic acids are usually basophilic and therefore cell compartments and organelles which are rich in DNA or RNA, like the cell nucleus, the ribosomes and some cytoplasmic regions are stained by hematoxylin. The alcohol-based acidic eosin Y, on the other hand, colors eosinophilic structures bright pink. These eosinophilic structures generally include intracellular or extracellular proteins and most of the cytoplasm.

The HE staining was performed according to a standard protocol. The sections were deparaffinized by applying decreasing concentrations of alcohol in the following order:

- (1) 5 min Roticlear
- (2) 5 min Roticlear®
- (3) 5 min absolute ethanol
- (4) 5 min 95 % ethanol
- (5) 5 min 70 % ethanol
- (6) 5 min 50 % ethanol
- (7) 5 min 30 % ethanol

Afterwards the slides were stored in distilled water for at least 5 min before staining.

Before use, the eosin Y and the mayers haematoxylin solutions were gently shaked and subsequently filtered. The eosin Y solution was supplemented with 0.01 % acetic acid (e.g. 10 μ L of concentrated acetic acid are added to 100 mL eosin Y). The sections were first incubated in haematoxylin for 5 min. The blue color was developed by rinsing the sections in warm tap water for up to 20 min (the water was replaced occasionally). Afterwards the slides were incubated in eosin Y for 5 min and shortly rinsed in distilled water.

After staining, the sections are dehydrated as follows: Short elution in 95 % ethanol followed by incubation in 99 % ethanol for 3 min and in xylene twice for each 5 min. The mounting medium (Roti[®]-Histokitt, Carl Roth) was subsequently applied onto the slides and the sections were covered with a cover slip and dried in the fume hood.

The microscopic evaluation was performed with a light microscope (Axiophot, Carl Zeiss) and a microscope camera (Axiocam, Carl Zeiss)

6.1.5.2 Immunohistochemistry

Indirect immunohistochemistry utilizes a specific primary antibody raised against a certain protein and a labeled secondary antibody that binds to the primary antibody. Thus, proteins of interest in histological sections of cell cultures or tissues can be visualized.

There are several methods for the detection of this cascade. The method described here is the most common one. The secondary antibody is conjugated to a horseradish peroxidase which forms a brown precipitate with the 3.3'-diaminobenzidine (DAB) substrate that can be detected by light microscopy. To prevent unspecific bindings of the antibodies and undesirable reactions, potential binding sites as well as endogenous peroxidases have to be blocked. All necessary reagents for immunohistochemical analyses can be obtained separately or as a kit. In commercially available kits, all reagents are optimally diluted and harmonized resulting in an optimized protocol.

Due to cross-links often caused by formaldehyde fixation, the antigenic sites of the proteins (antibody binding sites) can be masked. There are several antigen retrieval procedures for breaking these cross-links. Heat Induced antigen (Epitope) Retrieval (HIER) is a common procedure. Mostly, the sections are boiled in sodium citrate buffer at pH 6.0 or alternatively in Tris/EDTA buffer at pH 9.0.

The buffers were prepared as follows:

Sodium citrate buffer (pH 6.0):

2.94 g tri-sodium citrate dihydrate ($C_6H_5Na_3O_7 + 2H_2O$) were dissolved in 1000 mL distilled water on a magnetic stirrer. The pH was adjusted to 6.0 with 5 M hydrochloric acid (HCl). Tween 20 was added to a final concentration of 0.05 % (50 µL of 10 % Tween 20 solution).

Tris/EDTA buffer (pH 9.0):

1.21 g Tris(hydroxymethyl)aminomethane $(NH_2C(CH_2OH)_3)$ and 0.37 g EDTA (Ethylenediaminetetraacetic acid) tetrasodium salt were dissolved in 1000 mL distilled water

on a magnetic stirrer. The pH was adjusted to 9.0 with 5 M hydrochloric acid (HCl). Tween 20 was added to a final concentration of 0.05 % (50 μ L of 10 % Tween 20 solution).

Heat induced antigen retrieval /HIER)

The HIER was performed according to the following procedure:

Histological sections of the paraffin embedded cell cultures were prepared and deparaffinized as described in section 6.1.5.1. For the immunohistochemistry, poly-L-lysine coated microscope slides were used.

Before starting, a water bath was heated to 97 °C and an appropriate number of coplin staining jars were filled with 55 – 60 mL of the respective buffer. One to two slides were placed into each coplin staining jar and as much space as possible was left between the two sections and between the sections and the housing of the coplin staining jar to guarantee a uniform heating of the sections. The coplin staining jars were placed into the preheated water bath (97 °C) and incubated for 10 min. Afterwards, the caps of the coplin staining jars were removed and the buffer was cooled down to room temperature by placing the coplin staining jars into a cold water bath (incubation for approximately 30 min). The buffer was discarded and the slides were rinsed in running tap water for 1 min and stored in distilled water until starting the following procedure.

Immunohistochemical staining

The NovoLink[™] Polymer Detection System provided by Leica Microsystems, UK, was used for immunohistochemistry. This kit was specifically designed for immunostaining analyses on formalin-fixed, paraffin-embedded histological sections. The kit contains the following components:

- Peroxidase blocking solution (neutralized endogenous peroxidase activity)
- Protein blocking solution (reduces non-specific bindings of primary antibody and polymers)
- Post primary blocking solution (enhances penetration of the polymer reagent)
- Polymer (anti-mouse/rabbit IgG linked to poly-HRP (horseradish peroxidase); detects any tissue-bound mouse or rabbit IgG primary antibody)

- DAB chromogen and DAB substrate buffer (produces a visible brown precipitate when reacting with the peroxidase)
- Hematoxylin solution (0.02 %, for counterstaining of cells)

The sections on the microscope slides were encircled with a PAP Pen. This hydrophobic circle helps to prevent wasting the reagents. Thus, small volumes of the antibody solutions and other reagents could be dropped directly on top of the sections. The immunohistochemistry with the NovoLink[™] Polymer Detection System was performed according to the protocol provided by the manufacturer. All reagents were provided in dropper bottles which allow easy and economical application.

The kit could be used for mouse and rabbit IgG primary antibodies. Almost each step was followed by a washing step in TBS (recommended by manufacturer) or unsterile PBS (mostly used) for 2 x 5 min. The washing was performed in staining boxes (according to Hellendahl) which were filled with approx. 80 mL PBS. After several washing steps, the PBS was partially renewed. After the staining procedure, the sections were dehydrated and mounted using Roti[®]-Histokitt as described in section 6.1.5.1.

6.1.5.3 Immunofluorescence

The immunoflurescence (IF) is a technique that visualizes the distribution of biomolecules of interest in cell cultures or tissues by means of antibodies that are conjugated to fluorescent dyes. The analyses are conducted through the fluorescence microscopy. In contrast to the IHC technique described above, the IF is often used for cell cultures that are not paraffinembedded and cross-sectioned.

In this thesis, the indirect IF method was used which employs two sets of antibodies, a primary and a secondary antibody. The primary antibody binds specifically to the respective antigen and the secondary antibody that is conjugated to the fluorescent dye binds to the primary. The complex of antigen, primary and secondary antibody can be detected through a fluorescence microscope that is equipped with an UV lamp and a suitable filter for the respective wavelength of the fluorescent dye. To avoid unspecific bindings of the antibodies, potential binding sites are blocked by incubating the samples in normal serum that has been

isolated from the same organism as the secondary antibody. To enhance the penetration of the antibodies the cell membrane are permeablized by e.g. Triton X.

The immunofluorescence analyses of the cilia marker acetylated- α -tubulin in fresh, nonparaffin embedded NHBE cultures (section 4.2.2) was performed according to the following protocol:

After rinsing the surface of the cell cultures on insert membranes with PBS, the cells were fixed with 3.7 % formaldehyde solution for 20 min at room temperature, in the dark. The formaldehyde was then removed and the cells were washed three times with PBS for each 5 min. Triton-X-100 was diluted with PBS to obtain a 0.25 % solution, and the cells were incubated in this solution for 10 min at room temperature. After removing the Triton Xsolution the cells were washed three times for each 5 min in PBS. In order to block unspecific binding sites, the cells were subsequently incubated in 10 % rabbit serum (the secondary antibody used was raised in rabbit). The serum was diluted with PBS and the incubation was performed overnight at 4 °C. The membranes were then isolated from the plastic insert housing with a scalpel and placed in the wells of the receiver plate. The primary antibody was diluted 1:500 (dilution according to suppliers' recommendation) in a solution of PBS with 1 % BSA (PBS/ 1 % BSA) and applied onto the cells. Incubation was performed for 2 h at room temperature. The cells were subsequently rinsed three times with PBS for each 3 min. All following steps were performed in the dark. The secondary antibody "rabbit anti mouse Igg" (labeled with the rhodamine derivateTRITC) was diluted 1:200 with PBS/ 1 % BSA and applied onto the cells. Incubation was performed for 45 min. After washing three times with PBS for each 3 min, the cell nuclei were stained with $1 \mu g/mL$ DAPI for 1 min. The DAPI stock solution of 1 mg/mL was prepared in methanol and for the working solution the DAPI was further diluted with PBS. Subsequently, the cells were rinsed with PBS two times for each 2 min. The membranes were then placed on uncoated glass slides and the coverslips were mounted with Roti[®]-Mount FluorCare (Carl Roth). The microscopic analysis was performed using the Axiophot and Axiocam MRc (Carl Zeiss).

The immunofluorescence analyses of the Ki-67 marker in cross sections of paraffinembedded 16HBE14o- cultures (section 4.1.1.3) were performed according to the following protocol:

169

The sections were placed on poly-I-lysine coated glass slides, dried and deparaffinised according to the standard procedures described in section 6.1.5.1. Heat antigen retrieval in sodium citrate buffer was performed as described in section 6.1.5.2. The sections were afterwards encircled with the Pap Pen (see section 6.1.5.2), enabling the use of small amounts (approx. 100 µL) of all reagents. The reagents were directly applied onto the sections using a pipette (10 - 200 µL). The washing was performed in staining boxes (according to Hellendahl) which were filled with approx. 80 mL PBS. After several washing steps, the PBS was partially renewed. Permeablization was performed with 0.25 % Triton-X (in PBS) for 3 min followed by washing in PBS for three times for each 3 min. To block unspecific binding sites, the sections were afterwards incubated with 10 % rabbit serum (in PBS) for 30 min. Subsequently, the antibody against the Ki-67 antigen was applied in appropriate dilution (1:100 diluted in PBS/ 1 % BSA, as recommended by supplier). The sections were washed in PBS three times for each 3 min before the secondary antibody was applied. All subsequent steps were performed in the dark. The anti-mouse IgG -TRITC antibody was used in a dilution of 1:200 (in PBS/ 1% BSA) and incubated for 25 min. The sections were afterwards washed in PBS three times for each 3 min. For visualization of the cell nuclei, DAPI was applied in a concentration of $1 \mu g/mL$ (in PBS) and incubated for 1 min. After rinsing the sections with PBS, the coverslips was mounted using Roti[®]-Mount FluorCare (Carl Roth). The microscopic analysis was performed using the Axio Observer Z.1 (equipped for fluorescence analyses with the VivaTome technique) and Axiocam MRm (Carl Zeiss).

6.1.5.4 TEER measurement

In an intact epithelial cell layer, the cells are linked to each other by cell-to-cell contacts. These tight junctions occlude the extracellular space between the cells, forming a tight, beltlike adhesive seal, that selectively limits the diffusion of water, ions and larger solutes between the apical and the basolateral sides of the cell layer ⁽²⁵¹⁾.

This feature, called barrier function, leads to a high electrical resistance between the apical and the basolateral regions of the cell. The functionality of the barrier can be determined by measuring the electrical resistance between the two compartments. It is called transepithelial electrical resistance (TEER) and can be measured non-invasively using a special electrode system. If the tight junctions are disrupted, ions can cross the cell barrier more easily, resulting in loss of the high electrical resistance ⁽²⁵²⁾.

In this thesis, the TEER value of the cell cultures that were cultivated on insert membranes was determined by means of the Millicell[®] ERS-2 system (Millipore Inc.). For that purpose, the cell culture medium was removed and replaced by fresh, serum-free culture medium that was previously prewarmed at room temperature. The culture medium was applied on both sides of the insert membranes with a volume of 1000 μ L at the basal side (in the wells) and 500 μ L at the apical side (in the inserts). For TEER measurement, the shorter electrode tip was dipped in the apical medium and the longer tip in the basal medium. Both silver chloride pellets on the tips had to be completely immersed in the medium. To ensure stable and reproducible results, the electrode was held steady and at an angle of 90° to the membrane. Contact between the electrode tips and the cell layer or the bottom of the plate was avoided.

The resistance was inversely proportional to the growth area (resistance $[\Omega]$ x effective membrane area $[cm^2]$). A blank value was determined by measuring the TEER value of coated inserts without cells and this blank value was subtracted from the TEER values of the cell cultures.

6.1.5.5 ELISA

The Enzyme-linked immunosorbent assay (ELISA) was performed to analyze the secretion of IL-6 and IL-8, two cytokines that are associated with inflammation processes in the lung. For that purpose the waste medium of the cell cultures was sampled at different time points and used as template for the analyses.

In general, the ELISA technique enables the detection of substances of interest in liquid samples by means of specific antibodies. The procedure, which is carried out in 96-well plates, is divided into five sequential steps.

- Coating: The wells are coated with an antibody which is directed specifically against the desired antigen.
- Blocking: Free non-specific binding sites are blocked on the surface of the wells with bovine serum albumin.

- 3. Application of the sample: The respective antigen in the sample binds specifically to the antibody on the well.
- 4. Application of Detection Antibody: An enzyme-linked antibody binds to the antigen.
- 5. Detection: A substrate is added which is converted by the enzyme that is linked to the detection antibody into a colored reaction product. The amount of converted substrate is proportional to the amount of the respective antigen present in the sample and can be measured in a plate reader (Appliskan, Thermo Scientific).

For the determination of IL-6 and IL-8 secretion, two commercial ELISA kits were used:

- BD OptEIA[™] Set Human IL-6 (BD Biosciences; order no.: 555220)
- BD OptEIA[™] Set Human IL-8 (BD Biosciences; order no.: 555244)

The analyses were performed according to the manufacturer instructions. The optimal dilutions of the samples were determined by trial and error.

The following solutions were prepared for ELISA:

Coating buffer:

3.56 g NaHCO₃
0.795 g Na₂CO₃
Dissolved in 500 ml ddH₂O
pH adjsuted to 9.5 with 10 N NaOH

Assay Diluent (for dilution of detection antibody and samples): 10 % FBS in PBS (pH 7)

Wash buffer:

0.05 % Tween in PBS

Substrate solution:

1 mg TMB (Tetramethyl Benzidine) in 100 μl DMSO

+ 9.9 ml 100 mM NaOAc (in H₂O; pH 6)

+ 1 μl 30 % H₂O₂

Stop solution:

 $1M H_2SO_4$

Each ELISA analysis included a standard of 8 samples with defined concentrations of the protein of interest. Thus, quantification of the IL-6 and IL-8 secretion in the test cultures was enabled. The data were analyzed using Microsoft Excel and displayed in a bar chart.

6.2 Abbreviations

Table 6-3. List of abbreviations

Abbreviation	Meaning
AEC	Alveolar epithelial cells
ALI	Air-liquid interface
ATCC	American Type Cell Collection
approx.	Approximately
BPE	Bovine pituitary extract
BEGM	Bronchial Epithelium Growth Medium
cm	Centimeters
COPD	Chronic obstructive pulmonary diseases
CCM	Compartment Cell Model
CULTEX [®] LTC	CULTEX [®] Long Term Cultivation system
СК	Cytokeratin
°C	Degree celsius
DMSO	Dimethyl sulphoxide
DMEM	Dulbecco's Modified Eagle Medium
DSMZ	Deutsche Sammlung von Mikroorgiansmen und Zellkulturen
EMEM	Eagle's Minimum Essential Medium
ELISA	Enzyme-Linked Immunosorbent Assay
EGF	Epidermal growth factor
EMTU	Epithelial-Mesenchymal Trophic Unit
ECVAM	European Centre for the Validation of Alternative Methods
ECM	Extracellular matrix
FBS	Fetal bovine serum
e.g.	For example
GA	Gentamycin/amphotericin B
g	Gram
HTS	High throughput screening
h	Hour
НВЕрС	Human bronchial epithelial cells
HE	Hematoxylin-eosin
HPV	Human papiloma virus
ICCVAM	Inter-agency Coordinating Center for the Validation of
	Alternative Methods
IL	Interleukin
ISO	International Organization for Standardization
KRH	Klinikum Region Hannover
L	Liter
μg	Microgram
mg	Milligram
MEM	Minimum Essential Medium
Μ	Molar
MUC5AC	Mucin 5AC

Abbreviation	Meaning
MUC5B	Mucin 5B
nM	Nanomolar
ng	Nanogram
NHBE	Normal human bronchial epithelial cells
NHLF	Normal human lung fibroblasts
OECD	Organisation for Economic Co-operation and Development
Pen/Strep	Penicillin/Streptomycin
PBS	Phosphate buffered saline
RA	Retinoic acid
rH	Relative Humidity
RPMI	Roswell Park Memorial Institute medium
SCM	Sandwich Cell Model
cm ²	Square centimeters
m ²	Square meters
3D	Three-dimensional
TJ	Tight junction
TEER	Transepithelial electrical resistance
TGF	Transforming growth factor
TNF	Tumor necrosis factor
2D	Two-dimensional
ZEBET	Zentralstelle zur Erfassung und Bewertung von Ersatz- und Ergänzungsmethoden zum Tierversuch

6.3 Laboratory devices

Table 6-4: List of laboratory devices

Device	Supplier
Appliskan®	Thermo Fisher Scientific Inc.
Axio Observer.Z1 (inverted microscope equipped for fluorescence analyses, incl. VivaTome technology)	Carl Zeiss Microlmaging GmbH
AxioCam MRc	Carl Zeiss MicroImaging GmbH
AxioCam MRm	Carl Zeiss MicroImaging GmbH
Axiophot	Carl Zeiss MicroImaging GmbH
CO ₂ Incubator, Heraeus [®] , BBD6220	Thermo Fisher Scientific Inc.
Fully Automated Rotary Microtome Leica RM2255	Leika Mikrosysteme Vertrieb GmbH
Heating cabinet TDO66	Medite GmbH
Modular tissue embedding center Leica EG 1150	Leika Mikrosysteme Vertrieb GmbH
Multifuge 1S-R, Heraeus [®]	Thermo Fisher Scientific Inc.
Olympus CKX41	Olympus Deutschland GmbH
Olympus SC30	Olympus Deutschland GmbH
Paraffin tissue floating bath 25900	MEDAX GmbH & CO.
Safety cabinet, Heraeus [®] Safe 2020	Thermo Fisher Scientific Inc.
Spectramax [®] 340 PC	Molecular Devices
Water bath GFL 1006	GFL Gesellschaft für Labortechnik mbH
Water bath W16	Harry Gestigkeit GmbH

6.4 Antibodies

Table 6-5: List of primary and secondary antibodies

Gene/Protein	Organsim	Isotype	Company	Order no.	Dilutions
Acetyl-α-tubulin	Mouse	lgG2b	Abcam	ab24610	1:500
Anti-Cytokeratin-6	Mouse	lgG1	Abcam	ab18586	1:10
Anti-Cytokeratin-13	Mouse	lgG	Abcam	ab16112	1:150
Anti-MMP9	Mouse	lgG2a	Abcam	ab119906	1:300
Anti-MUC5AC	Rabbit	lgG	SantaCruz Biotechnology Inc.	sc-20118	1:300
Anti-MUC5B	Rabbit	lgG	SantaCruz Biotechnology Inc.	sc-20119	1:300
Anti-p63	Mouse	lgG2	Abcam	ab3239	1:200
Anti-MKI67	Mouse	lgG1	Leica Biosystems	NCL-Ki67- MM1	1:100
Name	Label	Organism	Company	Order no.	Dilutions
Anti-Mouse IgG (whole molecule)	TRITC	Rabbit	Sigma	T2402	1:200

6.5 Laboratory equipment

Table 6-6: List of laboratory equipment

Product	Supplier	Order no.
CASY Model TT	Roche Diagnostics Deutschland GmbH	40017
Coplin staining jars with screw cap	Sigma-Aldrich	S5641-12EA
Mechanical pipette, 0.5 - 10 µL; mLINE	Sartorius Weighing Technology GmbH	12595585
Mechanical pipette, 10 - 100 μL; mLINE	Sartorius Weighing Technology GmbH	12595546
Mechanical pipette, 100 - 1000 μ L; mLINE	Sartorius Weighing Technology GmbH	12606243
Mechanical pipette, 20 - 200 μL; mLINE	Sartorius Weighing Technology GmbH	12596853
Mini Vac Eco (vaccum pump)	Peqlab Biotechnologie GmbH	90-6030
PAP pen	Kisker Biotech GmbH & Co. KG	MKP1
Scalpel	Carl Roth GmbH & Co. KG	AH88.1
Scissors	Carl Roth GmbH & Co. KG	3569.1
Serological pipette supporter, CellMate [®] II	Thermo Fisher Scientific Inc.	1036-7302
Staining box	Carl Roth GmbH & Co. KG	H554.1
Staining boxes according to Hellendahl	Carl Roth GmbH & Co. KG	H550.1
Staining rack	Carl Roth GmbH & Co. KG	H552.1
Transferpette [®] S-12, 20 - 200 μL	Brand GmbH & Co. KG	703730
Tweezers, self-spanning	Carl Roth GmbH & Co. KG	LH73.1
Vortex Genie [®] 2	Carl Roth GmbH & Co. KG	P505.1
Wire handle for staining rack	Carl Roth GmbH & Co. KG	H553.1

6.6 Laboratory consumables

Product	Supplier	Order no.
Cap mats for deepwell plate	Carl Roth GmbH & Co. KG	EN10.1
CASY clean	Roche Diagnostics Deutschland GmbH	43052
CASY cups	Roche Diagnostics Deutschland GmbH	43003
CASY ton	Roche Diagnostics Deutschland GmbH	43001
Cell culture flask (vented), 25 cm ²	BD Biosciences	353109
Cell culture flask (vented), 75 cm ²	BD Biosciences	353136
Cell culture inserts, 12-well, 0.4 µm pores	BD Biosciences	353180
Cell culture inserts, 12-well, 1.0 μm pores	BD Biosciences	353103
Cell culture petri dishes, 35 mm (Nunc)	Thermo Fisher Scientific Inc.	150318
Centrifuge tubes, 15 mL	BD Biosciences	352096
Centrifuge tubes, 50 mL	BD Biosciences	352070
Centrifuge tubes, self-standing, 50 mL	Corning Inc.	430921
Companion plate, 6-well	BD Biosciences	353502
Companion plate, 12-well	BD Biosciences	353503
Cover slips, 18 x 18 mm	Carl Roth GmbH & Co. KG	0657.2
Cover slips, 24 x 60 mm	Carl Roth GmbH & Co. KG	H878
Deepwell plate	Eppendorf AG	0030505301
Disposable Scalpel Cutfix®	Carl Roth GmbH & Co. KG	T998.1
Disposable syringe, 60 mL	B.Braun Melsungen AG	4616502F
Eppendorf tubes, Safe lock, 1.5 mL	Eppendorf AG	30.120.086
Immuno plates, MaxiSorp surface (Nunc)	Thermo Fisher Scientific Inc.	460984
Microscope slides Polysine® (Menzel-Gläser)	Thermo Fisher Scientific Inc.	J2800AMNZ
Microscope slides, HISTOBOND [®]	Medite GmbH	46-8101-00
Pasteur pipettes, 150 mm	Carl Roth GmbH & Co. KG	4518
Petri dishes, SureGrip, 100 mm	Sarstedt AG & Co.	831802003
Pipette tips (with filter), 0.1 - 10 μ L	Sartorius AG	790301F
Pipette tips (with filter), 5 - 300 μL	Sartorius AG	790011F
Pipette tips (with filter), 50 - 1000 μL	Sartorius AG	790001F
Reagent reservoir	Carl Roth GmbH & Co. KG	HT66.1
riplate [®] - deepwell plate, 2 mL	Carl Roth GmbH & Co. KG	EN07.1
Scalpel blades	Carl Roth GmbH & Co. KG	AH89.1
Serological pipettes, 10 mL	Sarstedt AG & Co.	86.1254.001
Serological pipettes, 25 mL	Sarstedt AG & Co.	86.1685.001
Serological pipettes, 5 mL	Sarstedt AG & Co.	86.1253.001
Serological pipettes, 50 mL	Sarstedt AG & Co.	86.1689.001
Syringe Filter, 0.2 μm (Nalgene)	Thermo Fisher Scientific Inc.	190-2520
Transwell [®] Inserts, 12-well, 0.4 µm pores	Corning Inc.	3460

6.7 Chemicals and solutions

Table 6-8: List of chemicals and solutions

Product	Supplier	Order no.
Acetic acid 100%	AppliChem GmbH	A0820
BD OptEIA™ Set Human IL-6	BD Biosciences	555220
BD OptEIA™ Set Human IL-8	BD Biosciences	555244
Bovine serum albumin (BSA)	Sigma-Aldrich	A9418
Bovine serum albumin (BSA, cell culture grade)	Sigma-Aldrich	A2153
Collagen type I (from bovine skin, 3mg/ml)	Sigma-Aldrich	C4243
Collagen type IV (human placenta, Bornstein and Traub)	Sigma-Aldrich	C5533
DAPI BioChemica	AppliChem GmbH	A1001
Disodium hydrogen phosphate (Na2HPO4)	AppliChem GmbH	A3567
Dimethyl sulphoxide (DMSO)	Sigma-Aldrich	D2650
EDTA (Ethylenediaminetetraacetic acid) tetrasodium salt dihydrate	Sigma-Aldrich-Aldrich	ED4SS
Eosin G solution	Carl Roth GmbH & Co. KG	X883.2
Ethanol, absolute	AppliChem GmbH	A4930
Fibronectin (human, 1mg/mL)	Biochrom AG	L7117
Formaldehyde solution, 37 %	AppliChem GmbH	A3592
G418 (Genetiticin)	AppliChem GmbH	A6798
H_2SO_4 Sulfuric acid (2N)	Carl Roth GmbH & Co. KG	X873.1
Hemalum solution (hemalum acc. to Mayer)	Carl Roth GmbH & Co. KG	T865.2
Hydrochloric acid (HCl), 5M	AppliChem GmbH	A2668
Immersion Oil	AppliChem GmbH	A3494
Isopropyl alcohol (absolute)	AppliChem GmbH	A0995
LHC Basal Medium	Life Technologies Corp.	12677-019
NovoLink [®] Polymer Detection System	Leica Microsystems	RE7140-K
Paraplast® (Leica)	Carl Roth GmbH & Co. KG	X880.1
PBS (phosphate buffered saline) tablets	Life Technologies Corp.	518912-014
PBS (phosphate buffered saline, sterile; without Mg^{2+} and Ca^{2+})	Biochrom AG	L1825
Recovery Cell Culture Freezing Medium	Life Technologies Corp.	12648-010
Roti [®] -Histokitt	Carl Roth GmbH & Co. KG	6638.1
Roti [®] -Mount FluorCare	Carl Roth GmbH & Co. KG	HP19.1

Product	Supplier	Order no.
Sodium acetate (CH ₃ COONa)	Carl Roth GmbH & Co. KG	6773.1
Sodium carbonate (Na ₂ CO ₃)	Sigma-Aldrich	57795-500G
Sodium hydrogen carbonate (NaHCO ₃)	Carl Roth GmbH & Co. KG	HNO1.1
Sodium hydroxide pellets (Na ₂ CO ₃)	AppliChem GmbH	A1551
Sodiumhydrogenphosphate (NaH ₂ PO ₄₎	AppliChem GmbH	A3559
3,3',5,5'-Tetramethylbenzidin (TMB)	Carl Roth GmbH & Co. KG	6350.1
Tris(hydroxymethyl)aminomethane (Sigma-Aldrich 7-9)	Sigma-Aldrich-Aldrich	T1378
Trisodium citrate dihydrate	AppliChem GmbH	A3901
Triton [®] -X-100	AppliChem GmbH	A1388
Trypsin inhibitor from soybean (50X)	AppliChem GmbH	A8339
Trypsin/EDTA solution (0.05%/0.02% (w/v))	Biochrom AG	L 2143
Trypsin/EDTA solution (0.25%/0.02% (w/v))	Biochrom AG	L 2163
Tween 20 (10% solution)	BioRad Laboratories Inc.	161-0781
Water, bidistilled (ddH ₂ O)	AppliChem GmbH	A4042
Xylene, 98,5%	Carl Roth GmbH & Co. KG	CN80.2

7 References

- 1. Afonso ASM, Verhamme KMC, Sturkenboom MCJM, & Brusselle GGO (2011) COPD in the general population: Prevalence, incidence and survival. *Respiratory Medicine* 105(12):1872-1884.
- 2. Evans MJ, Van Winkle LS, Fanucchi MV, & Plopper CG (1999) The attenuated fibroblast sheath of the respiratory tract epithelial-mesenchymal trophic unit. *Am J Respir Cell Mol Biol* 21(6):655-657.
- 3. Myerburg MM, *et al.* (2007) Hepatocyte growth factor and other fibroblast secretions modulate the phenotype of human bronchial epithelial cells. *Am J Physiol Lung Cell Mol Physiol* 292(6):L1352-1360.
- 4. Skibinski G, Elborn JS, & Ennis M (2007) Bronchial epithelial cell growth regulation in fibroblast cocultures: the role of hepatocyte growth factor. *Am J Physiol Lung Cell Mol Physiol* 293(1):L69-76.
- 5. Gruenert DC, Finkbeiner WE, & Widdicombe JH (1995) Culture and transformation of human airway epithelial cells. *Am J Physiol* 268(3 Pt 1):L347-360.
- 6. James Kirkpatrick C, Fuchs S, Iris Hermanns M, Peters K, & Unger RE (2007) Cell culture models of higher complexity in tissue engineering and regenerative medicine. *Biomaterials* 28(34):5193-5198.
- 7. Stewart CE, Torr EE, Mohd Jamili NH, Bosquillon C, & Sayers I (2011) Evaluation of differentiated human bronchial epithelial cell culture systems for asthma research. *J Allergy* (*Cairo*) 2012:943982.
- 8. Forbes II (2000) Human airway epithelial cell lines for in vitro drug transport and metabolism studies. *Pharm Sci Technolo Today* 3(1):18-27.
- 9. Sachs LA, Finkbeiner WE, & Widdicombe JH (2003) Effects of media on differentiation of cultured human tracheal epithelium. *In Vitro Cell Dev Biol Anim* 39(1-2):56-62.
- 10. SOT SoT (2012).
- 11. AltTox (2011) Non-animal Methods for Toxicity Testing (www.alttox.org.).
- 12. Trevan JW (1927) The Error of Determination of Toxicity. *Pr Soc B* 101(712):483-514.
- 13. Draize JH, et al. (1948) Toxicological Investigations of Compounds Proposed for Use as Insect Repellents. *Journal of Pharmacology and Experimental Therapeutics* 93(1):26-39.
- 14. Hogan MD (1983) Extrapolation of animal carcinogenicity data: limitations and pitfalls. *Environ Health Perspect* 47:333-337.
- 15. Knight A (2007) Systematic reviews of animal experiments demonstrate poor human clinical and toxicological utility. *Altern Lab Anim* 35(6):641-659.
- 16. Dourson ML & Stara JF (1983) Regulatory history and experimental support of uncertainty (safety) factors. *Regul Toxicol Pharmacol* 3(3):224-238.
- 17. Andersen ME & Krewski D (2009) Toxicity Testing in the 21st Century: Bringing the Vision to Life. *Toxicological Sciences* 107(2):324-330.
- 18. Russell WMS (1959) *The principles of humane experimental technique [by] W.M.S. Russell and R.L. Burch* (Methuen, London).
- 19. Stephens ML, Conlee K, Alvino G, & Rowan AN (2002) Possibilities for refinement and reduction: future improvements within regulatory testing. *ILAR J* 43 Suppl:S74-79.
- 20. Richmond J (2002) Refinement, reduction, and replacement of animal use for regulatory testing: future improvements and implementation within the regulatory framework. *ILAR J* 43 Suppl:S63-68.
- 21. Kilkenny C, et al. (2009) Survey of the Quality of Experimental Design, Statistical Analysis and Reporting of Research Using Animals. *PLoS ONE* 4(11):e7824.
- 22. Liebsch M & Spielmann H (2002) Currently available in vitro methods used in the regulatory toxicology. *Toxicol Lett* 127(1-3):127-134.

- 23. Hayes A & Bakand S (2007) Inhalation Toxicology. *Occupational and Environmental Medicine* 64(8):461-488.
- 24. Barnes PJ (2010) New therapies for chronic obstructive pulmonary disease. *Med Princ Pract* 19(5):330-338.
- 25. Lopez AD & Murray CC (1998) The global burden of disease, 1990-2020. *Nat Med* 4(11):1241-1243.
- 26. Aufderheide M, Scheffler S, Mohle N, Halter B, & Hochrainer D (2011) Analytical in vitro approach for studying cyto- and genotoxic effects of particulate airborne material. *Anal Bioanal Chem*.
- 27. Aufderheide M (2008) An efficient approach to study the toxicological effects of complex mixtures. *Exp Toxicol Pathol* 60(2-3):163-180.
- 28. Aufderheide M & Mohr U (2000) CULTEX- an alternative technique for cultivation and exposure of cells of the respiratory tract to airborne pollutants at the air/liquid interface. *Exp Toxicol Pathol* 52(3):265-270.
- 29. EU (2010) Directive 2010/63/EU of the European Parliament and of the Council of 22 September 2010 on the protection of animals used for scientific purposes. *Official Journal of the European Union* 276:33 79.
- 30. Balls M (2002) The establishment of ECVAM and its progress since 1993. *Altern Lab Anim* 30 Suppl 2:5-11.
- 31. Hartung T, *et al.* (2004) A modular approach to the ECVAM principles on test validity. *Altern Lab Anim* 32(5):467-472.
- 32. Spielmann H & Liebsch M (2002) Validation successes: chemicals. *Altern Lab Anim* 30 Suppl 2:33-40.
- 33. European Commision (2011) Consumer Affairs; Executive Summary to the 7th Amendment to the Cosmetic Directive. in *www.ec.europa.eu*.
- 34. Combes RD & Balls M (2011) Integrated testing strategies for toxicity employing new and existing technologies. *Altern Lab Anim* 39(3):213-225.
- 35. Jaworska J & Hoffmann S (2010) Integrated Testing Strategy (ITS) Opportunities to Better Use Existing Data and Guide Future Testing in Toxicology. *21st Century Validation Strategies for 21st Century Tools*, (ALTEX), pp 231-242.
- 36. Levenberg S & Langer R (2004) Advances in Tissue Engineering. *Current Topics in Developmental Biology*, (Academic Press), Vol Volume 61, pp 113-134.
- 37. Stark Y, et al. (2006) Application of collagen matrices for cartilage tissue engineering. Exp Toxicol Pathol 57(4):305-311.
- 38. Kasper C, et al. (2007) A Newly Developed Rotating Bed Bioreactor for Bone Tissue Engineering. *Topics in Tissue Engineering*, eds Ashammakhi N, Reis R, & Chiellini E, Vol 3.
- 39. Alberts B, et al. (2002) Molecular Biology of The Cell, 4th Edition (Garland Science, New York).
- 40. Harris H (1995) Somatic Cells Outside the Body. *The Cells of the Body: A History Of Somatic Cell Genetics*, ed Harris H (Cold Spring Habor Laboratory Press, Cold Spring Habor), pp 31-56.
- 41. Freshney I, R. (2010) *Culture of Animal Cells: A Manual of Basic Technique and Specialized Applications* (Wiley-Blackwell, Hoboken, New Jersey) 6 Ed.
- 42. Theerakittayakorn K & Bunprasert T (2011) Differentiation Capacity of Mouse L929 Fibroblastic Cell Line Compare With Human Dermal Fibroblast. *World Academy of Science, Engineering and Technology* 74:378-381.
- 43. Hayflick L & Moorhead P (1961) The serial cultivation of human diploid cell strains. *Exp Cell Res.* 25:585-621.
- 44. Pezzulo AA, et al. (2010) The air-liquid interface and use of primary cell cultures are important to recapitulate the transcriptional profile of in vivo airway epithelia. Am J Physiol Lung Cell Mol Physiol 300(1):L25-31.
- 45. Lodish H, Berk A, Zipursky SL, & al. e (2000) Growth of Animal Cells in Culture. *Molecular Cell Biology, 4th edition,* (W.H. Freeman, New York).

- 46. Cozens AL, *et al.* (1994) CFTR expression and chloride secretion in polarized immortal human bronchial epithelial cells. *Am J Respir Cell Mol Biol* 10(1):38-47.
- 47. Bodnar AG, *et al.* (1998) Extension of life-span by introduction of telomerase into normal human cells. *Science* 279(5349):349-352.
- 48. Vaziri H & Benchimol S (1998) Reconstitution of telomerase activity in normal human cells leads to elongation of telomeres and extended replicative life span. *Curr Biol* 8(5):279-282.
- 49. Goodman TT, Ng CP, & Pun SH (2008) 3-D tissue culture systems for the evaluation and optimization of nanoparticle-based drug carriers. *Bioconjug Chem* 19(10):1951-1959.
- 50. Yamada KM, Pankov R, & Cukierman E (2003) Dimensions and dynamics in integrin function. *Brazilian Journal of Medical and Biological Research* 36:959-966.
- 51. Sun T, Jackson S, Haycock JW, & MacNeil S (2006) Culture of skin cells in 3D rather than 2D improves their ability to survive exposure to cytotoxic agents. *Journal of Biotechnology* 122(3):372-381.
- 52. Abbott A (2003) Cell culture: Biology's new dimension. *Nature* 424:870.
- 53. Huang S, Wiszniewski L, & Constant S (2011) The Use of In Vitro 3D Cell Models in Drug Development for Respiratory Diseases. *Drug Discovery and Development Present and Future*, ed Kapetanovic IM (Intech).
- 54. Gwak S-J, *et al.* (2006) A method for the effective formation of hepatocyte spheroids using a biodegradable polymer nanosphere. *Journal of Biomedical Materials Research Part A* 78A(2):268-275.
- 55. Härmä V, *et al.* (2010) A Comprehensive Panel of Three-Dimensional Models for Studies of Prostate Cancer Growth, Invasion and Drug Responses. *PLoS ONE* 5(5):e10431.
- 56. Oktem G, *et al.* (2006) Effect of apoptosis and response of extracellular matrix proteins after chemotherapy application on human breast cancer cell spheroids. *Oncol Rep* 15(2):335-340.
- 57. Drewitz M, et al. (2011) Verfahren zur automatisierten Tropfenbildung für die Massenproduktion von organotypischen Mikrogeweben (Novel Production Technology to Automate the Generation of Hanging Drops for Mass Production of Organotypic Microtissues). Chemie Ingenieur Technik 83(12):2170-2176.
- 58. Tibbitt MW & Anseth KS (2009) Hydrogels as extracellular matrix mimics for 3D cell culture. *Biotechnology and Bioengineering* 103(4):655-663.
- 59. Auger FA, *et al.* (1998) Tissue-engineered human skin substitutes developed from collagenpopulated hydrated gels: clinical and fundamental applications. *Med Biol Eng Comput* 36(6):801-812.
- 60. Reichl S, Bednarz J, & Muller-Goymann CC (2004) Human corneal equivalent as cell culture model for in vitro drug permeation studies. *Br J Ophthalmol* 88(4):560-565.
- 61. Vaughan MB, et al. (2004) A reproducible laser-wounded skin equivalent model to study the effects of aging in vitro. *Rejuvenation Res* 7(2):99-110.
- 62. Artym VV & Matsumoto K (2010) Imaging cells in three-dimensional collagen matrix. *Curr Protoc Cell Biol* Chapter 10:Unit 10 18 11-20.
- 63. Lee J, Cuddihy MJ, & Kotov NA (2008) Three-dimensional cell culture matrices: state of the art. *Tissue Eng Part B Rev* 14(1):61-86.
- 64. Kreger ST, et al. (2010) Polymerization and matrix physical properties as important design considerations for soluble collagen formulations. *Biopolymers* 93(8):690-707.
- 65. Rhee S (2009) Fibroblasts in three dimensional matrices: cell migration and matrix remodeling. *Exp Mol Med* 41(12):858-865.
- 66. Rhee S & Grinnell F (2007) Fibroblast mechanics in 3D collagen matrices. *Adv Drug Deliv Rev* 59(13):1299-1305.
- 67. Parenteau-Bareil R, Gauvin R, & Berthod F (2010) Collagen-Based Biomaterials for Tissue Engineering Applications. *Materials* 3(3):1863-1887.
- 68. Hulmes DJS (2008) Collagen Diversity, Synthesis and Assembly. *Collagen Structure and Mechanics*, ed Fratzl P (Springer Science+Business Media LLC, New York), pp 15-48.

- 69. Ehrhardt C, Forbes B, & Kim K-J (2008) In Vitro Models of the Tracheo-Bronchial Epithelium. *Drug Absorption Studies: In situ, In Vitro and In Silicio Models (Biotechnology: Pharmaceutical Aspects),* eds Ehrhardt C & Kim K-J (Springer), Vol VII, pp 235-257.
- 70. Bergman R, A., Afifi A, K., & Heidger P, M. (1999) Atlas of Microscopic Anatomy, Section 11: Respiratory System. (www.anatomyatlases.org).
- 71. Horsfield K & Cumming G (1968) Morphology of the bronchial tree in man. *Journal of Applied Physiology* 24(3):373-383.
- 72. Jackson CL & Huber JF (1943) Correlated Applied Anatomy of the Bronchial Tree and Lungs With a System of Nomenclature. *Chest* 9(4):319-326.
- 73. Hayward J & Reid ML (1952) The cartilage of the intrapulmonary bronchi in normal lungs, in bronchiectasis, and in massive collapse. *Thorax* 7(1):98-110.
- 74. Moran & Rowley Histology Teaching Series. (www.visualhistology.com).
- 75. Claire M, et al. (2002) Bronchial epithelial cell cultures. *Cell Culture Models of Biological Barriers*, (CRC Press), pp 211-227.
- 76. Knight DA & Holgate ST (2003) The airway epithelium: Structural and functional properties in health and disease. *Respirology* 8(4):432-446.
- 77. French CA (2009) Respiratory Tract. *Cytology: Diagnostic Principles and Clinical Correlates*, eds Cibas ES & Ducatman BS (Saunders Elsevier, Philadelphia, PA), pp 65-104.
- 78. BeruBe K, et al. (2009) In vitro models of inhalation toxicity and disease. The report of a FRAME workshop. Altern Lab Anim 37(1):89-141.
- 79. Reynolds SD, *et al.* (2004) Airway injury in lung disease pathophysiology: selective depletion of airway stem and progenitor cell pools potentiates lung inflammation and alveolar dysfunction. *Am J Physiol Lung Cell Mol Physiol* 287(6):L1256-1265.
- 80. Brown B, Lindberg K, Reing J, Stolz DB, & Badylak SF (2006) The basement membrane component of biologic scaffolds derived from extracellular matrix. *Tissue Eng* 12(3):519-526.
- 81. Knight D (2001) Epithelium-fibroblast interactions in response to airway inflammation. *Immunol Cell Biol* 79(2):160-164.
- 82. Snyder JC, Zemke AC, & Stripp BR (2009) Reparative capacity of airway epithelium impacts deposition and remodeling of extracellular matrix. *Am J Respir Cell Mol Biol* 40(6):633-642.
- 83. Joo NS, Wu JV, Krouse ME, Saenz Y, & Wine JJ (2001) Optical method for quantifying rates of mucus secretion from single submucosal glands. *American Journal of Physiology Lung Cellular and Molecular Physiology* 281(2):L458-L468.
- 84. Ballard ST & Inglis SK (2004) Liquid secretion properties of airway submucosal glands. *J Physiol* 556(Pt 1):1-10.
- 85. Ehrhardt C, Laue M, & Kim K-J (2008) In Vitro Models of the Alveolar Epithelial Barrier. *Drug Absorption Studies: In situ, In Vitro and In Silicio Models (Biotechnology: Pharmaceutical Aspects),* eds Ehrhardt C & Kim K-J (Springer), pp 258-282.
- 86. Fehrenbach H (2001) Alveolar epithelial type II cell: defender of the alveolus revisited. *Respir Res* 2(1):33-46.
- 87. Reynolds HY (2005) Lung inflammation and fibrosis: an alveolar macrophage-centered perspective from the 1970s to 1980s. *Am J Respir Crit Care Med* 171(2):98-102.
- 88. Soleas JP, Paz A, Marcus P, McGuigan A, & Waddell TK (2012) Engineering airway epithelium. *J Biomed Biotechnol* 2012:982971.
- 89. Lutter R & Spiteri M (2005) Current perspectives in epithelial cell injury and repair: consequences for epithelial functions. *European Respiratory Review* 14(97):126-130.
- 90. Jeffery PK (2004) Remodeling and inflammation of bronchi in asthma and chronic obstructive pulmonary disease. *Proc Am Thorac Soc* 1(3):176-183.
- 91. Fahy JV & Dickey BF (2010) Airway mucus function and dysfunction. N Engl J Med 363(23):2233-2247.
- 92. Houtmeyers E, Gosselink R, Gayan-Ramirez G, & Decramer M (1999) Regulation of mucociliary clearance in health and disease. *Eur Respir J* 13(5):1177-1188.

- 93. Kirkham S, Sheehan JK, Knight D, Richardson PS, & Thornton DJ (2002) Heterogeneity of airways mucus: variations in the amounts and glycoforms of the major oligomeric mucins MUC5AC and MUC5B. *Biochem J* 361(Pt 3):537-546.
- 94. Rogers DF (2003) The airway goblet cell. Int J Biochem Cell Biol 35(1):1-6.
- 95. Pilewski JM & Albelda SM (1993) Adhesion Molecules in the Lung: An Overview. *American Journal of Respiratory and Critical Care Medicine* 148(6 Pt 2):S31-S37.
- 96. Vareille M, Kieninger E, Edwards MR, & Regamey N (2011) The Airway Epithelium: Soldier in the Fight against Respiratory Viruses. *Clinical Microbiology Reviews* 24(1):210-229.
- 97. Coyne CB, et al. (2002) Regulation of Airway Tight Junctions by Proinflammatory Cytokines. Molecular Biology of the Cell 13(9):3218-3234.
- 98. Crosby LM & Waters CM (2010) Epithelial repair mechanisms in the lung. *American Journal of Physiology Lung Cellular and Molecular Physiology* 298(6):L715-L731.
- 99. Crystal RG, Randell SH, Engelhardt JF, Voynow J, & Sunday ME (2008) Airway epithelial cells: current concepts and challenges. *Proc Am Thorac Soc* 5(7):772-777.
- 100. Hiemstra PS (2005) Repair and Defense Systems at the Epithelial Surface in the Lung: Integrative Physiology in the Proteomics and Post-Genomics Age. ed Walz W (Humana Press), pp 201-214.
- 101. Holgate ST (2008) The Epithelium and Airway Remodelling. *The Pulmonary Epithelium in Health and Disease*, ed Proud D (John Wiley & Sons Ltd, Chichester), pp 351-366.
- 102. Herfs M, et al. (2012) Proinflammatory Cytokines Induce Bronchial Hyperplasia and Squamous Metaplasia in Smokers. *American journal of respiratory cell and molecular biology* 47(1):67-79.
- 103. Donnelly LE (2008) The Epithelium as a Target. *The Pulmonary Epithelium in Health and Disease*, ed Proud D (John Wiley & Sons Ltd, Chichester), pp 57-74.
- 104. Coraux C, Roux J, Jolly T, & Birembaut P (2008) Epithelial cell-extracellular matrix interactions and stem cells in airway epithelial regeneration. *Proc Am Thorac Soc* 5(6):689-694.
- 105. Puchelle E, Zahm JM, Tournier JM, & Coraux C (2006) Airway epithelial repair, regeneration, and remodeling after injury in chronic obstructive pulmonary disease. *Proc Am Thorac Soc* 3(8):726-733.
- 106. Franklin WA, Veve R, Hirsch FR, Helfrich BA, & Bunn PA, Jr. (2002) Epidermal growth factor receptor family in lung cancer and premalignancy. *Semin Oncol* 29(1 Suppl 4):3-14.
- 107. Kerr KM (2012) The Classification of Pre-invasive Lesions. *Molecular Pathology of Lung Cancer*, eds cagle PT, Allen TC, Beasley MB, Chirieac LR, Dacic S, & Borczuk AC (Springer, New York).
- 108. Giangreco A, Groot KR, & Janes SM (2007) Lung cancer and lung stem cells: strange bedfellows? *Am J Respir Crit Care Med* 175(6):547-553.
- 109. Bergeron C & Boulet LP (2006) Structural changes in airway diseases: characteristics, mechanisms, consequences, and pharmacologic modulation. *Chest* 129(4):1068-1087.
- 110. Haswell LE, Hewitt K, Thorne D, Richter A, & Gaca MD (2010) Cigarette smoke total particulate matter increases mucous secreting cell numbers in vitro: a potential model of goblet cell hyperplasia. *Toxicol In Vitro* 24(3):981-987.
- 111. Cohn L (2006) Mucus in chronic airway diseases: sorting out the sticky details. *J Clin Invest* 116(2):306-308.
- 112. Thornton DJ, Rousseau K, & McGuckin MA (2008) Structure and Function of the Polymeric Mucins in Airways Mucus. *Annual Review of Physiology* 70(1):459-486.
- 113. Fahy JV (2002) Goblet Cell and Mucin Gene Abnormalities in Asthma. *CHEST* 122 (6_suppl):320S-326S.
- 114. Orcdoñez CL, *et al.* (2001) Mild and Moderate Asthma Is Associated with Airway Goblet Cell Hyperplasia and Abnormalities in Mucin Gene Expression. *American Journal of Respiratory and Critical Care Medicine* 163(2):517-523.
- 115. Rogers DF (2002) Airway goblet cell hyperplasia in asthma: hypersecretory and antiinflammatory? *Clinical & Experimental Allergy* 32(8):1124-1127.

- 116. Rogers DF (2004) Airway mucus hypersecretion in asthma: an undervalued pathology? *Curr Opin Pharmacol* 4(3):241-250.
- 117. Araya J, et al. (2007) Squamous metaplasia amplifies pathologic epithelial-mesenchymal interactions in COPD patients. J Clin Invest 117(11):3551-3562.
- 118. Lagente V, et al. (2005) Role of matrix metalloproteinases in the development of airway inflammation and remodeling. *Braz J Med Biol Res* 38(10):1521-1530.
- 119. Darby IA & Hewitson TD (2007) Fibroblast differentiation in wound healing and fibrosis. *Int Rev Cytol* 257:143-179.
- 120. Hallgren O, et al. (2010) Altered fibroblast proteoglycan production in COPD. Respiratory Research 11(1):55.
- 121. Pini L, *et al.* (2007) Differences in proteoglycan deposition in the airways of moderate and severe asthmatics. *European Respiratory Journal* 29(1):71-77.
- 122. Proud D ed (2008) *The Pulmonary Epithelium in Health and Disease* (John Wiley & Sons LTD., Chichester).
- 123. Van Lint P & Libert C (2007) Chemokine and cytokine processing by matrix metalloproteinases and its effect on leukocyte migration and inflammation. *Journal of Leukocyte Biology* 82(6):1375-1381.
- 124. Konigshoff M, Kneidinger N, & Eickelberg O (2009) TGF-beta signaling in COPD: deciphering genetic and cellular susceptibilities for future therapeutic regimen. *Swiss Med Wkly* 139(39-40):554-563.
- 125. Makinde T, Murphy RF, & Agrawal DK (2007) The regulatory role of TGF-beta in airway remodeling in asthma. *Immunol Cell Biol* 85(5):348-356.
- 126. Siegel PM & Massague J (2003) Cytostatic and apoptotic actions of TGF-[beta] in homeostasis and cancer. *Nat Rev Cancer* 3(11):807-820.
- 127. Ito J, et al. (2011) Wound-induced TGF-beta1 and TGF-beta2 enhance airway epithelial repair via HB-EGF and TGF-alpha. *Biochem Biophys Res Commun* 412(1):109-114.
- 128. Doerner AM & Zuraw BL (2009) TGF-beta1 induced epithelial to mesenchymal transition (EMT) in human bronchial epithelial cells is enhanced by IL-1beta but not abrogated by corticosteroids. *Respir Res* 10:100.
- 129. Masui T, *et al.* (1986) Type beta transforming growth factor is the primary differentiationinducing serum factor for normal human bronchial epithelial cells. *Proc Natl Acad Sci U S A* 83(8):2438-2442.
- 130. Takizawa H, *et al.* (2001) Increased expression of transforming growth factor-beta1 in small airway epithelium from tobacco smokers and patients with chronic obstructive pulmonary disease (COPD). *Am J Respir Crit Care Med* 163(6):1476-1483.
- 131. Tanabe T, Kanoh S, Moskowitz WB, & Rubin BR (2012) Transforming Growth Factor-β From the Failing Heart Leads to Squamous Metaplasia in Human Airway Cells and in the Murine Lung. *CHEST* 142(5):1274-1283.
- 132. Kondo M, Tamaoki J, Takeyama K, Nakata J, & Nagai A (2002) Interleukin-13 induces goblet cell differentiation in primary cell culture from Guinea pig tracheal epithelium. *Am J Respir Cell Mol Biol* 27(5):536-541.
- 133. Thavagnanam S, et al. (2011) Effects of IL-13 on mucociliary differentiation of pediatric asthmatic bronchial epithelial cells. *Pediatr Res* 69(2):95-100.
- 134. Zhen G, et al. (2007) IL-13 and epidermal growth factor receptor have critical but distinct roles in epithelial cell mucin production. *Am J Respir Cell Mol Biol* 36(2):244-253.
- 135. Tyner JW, et al. (2006) Blocking airway mucous cell metaplasia by inhibiting EGFR antiapoptosis and. *The Journal of Clinical Investigation* 116(2):309-321.
- 136. Thomas LH, Wickremasinghe MI, & Friedland JS (2007) IL-1 beta stimulates divergent upper and lower airway epithelial cell CCL5 secretion. *Clin Immunol* 122(2):229-238.
- 137. O'Gorman MT, Jatoi NA, Lane SJ, & Mahon BP (2005) IL-1beta and TNF-alpha induce increased expression of CCL28 by airway epithelial cells via an NFkappaB-dependent pathway. *Cell Immunol* 238(2):87-96.

- 138. Baggiolini M & Clark-Lewis I (1992) Interleukin-8, a chemotactic and inflammatory cytokine. *FEBS Letters* 307(1):97-101.
- 139. Li J, et al. (2002) Regulation of human airway epithelial cell IL-8 expression by MAP kinases. American Journal of Physiology - Lung Cellular and Molecular Physiology 283(4):L690-L699.
- 140. Linden A (1996) Increased interleukin-8 release by beta-adrenoceptor activation in human transformed bronchial epithelial cells. *Br J Pharmacol* 119(2):402-406.
- 141. Neveu W, et al. (Elevation of IL-6 in the allergic asthmatic airway is independent of inflammation but associates with loss of central airway function. *Respiratory Research* 11(1):28.
- 142. Ge Q, Moir LM, Black JL, Oliver BG, & Burgess JK (TGFβ1 induces IL-6 and inhibits IL-8 release in human bronchial epithelial cells: The role of Smad2/3. *Journal of Cellular Physiology* 225(3):846-854.
- 143. Loddenkemper R, Gibson GJ, & Sibille Y (2003) The burden of lung disease in Europe: why a European White Book on lung disease? *Eur Respir J* 22(6):869.
- 144. O'Neil SE, Lundbäck B, & Lötvall J (2011) Proteomics in asthma and COPD phenotypes and endotypes for biomarker discovery and improved understanding of disease entities. *Journal of Proteomics* 75(1):192-201.
- 145. WAO (World Allergy Organization: Disease Summaries; Chronic Obstructive Pulmonary Disease (COPD) and Asthma: Similarities and Differences. (*www.worldallergy.org*).
- 146. van der Strate BWA, et al. (2006) Cigarette Smoke–induced Emphysema. American Journal of Respiratory and Critical Care Medicine 173(7):751-758.
- 147. Johnson M (2004) Interactions between Corticosteroids and β2-Agonists in Asthma and Chronic Obstructive Pulmonary Disease. *Proceedings of the American Thoracic Society* 1(3):200-206.
- 148. Smart BA (2001) When asthma and COPD Coexist. in *www.copd-international.com*.
- 149. Lindberg A, Bjerg-Bäcklund A, Rönmark E, Larsson L-G, & Lundbäck B (2006) Prevalence and underdiagnosis of COPD by disease severity and the attributable fraction of smoking: Report from the Obstructive Lung Disease in Northern Sweden Studies. *Respiratory Medicine* 100(2):264-272.
- 150. Hong KU, Reynolds SD, Watkins S, Fuchs E, & Stripp BR (2004) Basal cells are a multipotent progenitor capable of renewing the bronchial epithelium. *Am J Pathol* 164(2):577-588.
- 151. Evans MJ, Cabral-Anderson LJ, & Freeman G (1978) Role of the Clara cell in renewal of the bronchiolar epithelium. *Lab Invest* 38(6):648-653.
- 152. Allen DD, et al. (2005) Cell Lines as In Vitro Models for Drug Screening and Toxicity Studies. Drug Development and Industrial Pharmacy 31(8):757-768.
- 153. Epithelix (www.epithelix.com).
- 154. Mattek (www.mattek.com).
- 155. Forbes B & Ehrhardt C (2005) Human respiratory epithelial cell culture for drug delivery applications. *Eur J Pharm Biopharm* 60(2):193-205.
- 156. Coecke S, *et al.* (2005) Guidance on good cell culture practice. a report of the second ECVAM task force on good cell culture practice. *Altern Lab Anim* 33(3):261-287.
- 157. Karunaratne DN, *et al.* (2005) Cell Culture models For Drug Transport Studies. *DRUG DELIVERY: Principles and Applications*, eds Wang B, Siahaan T, & Soltero R (John Wiley & Sons Inc., New Jersey), pp 103 124.
- 158. Gray TE, Guzman K, Davis CW, Abdullah LH, & Nettesheim P (1996) Mucociliary differentiation of serially passaged normal human tracheobronchial epithelial cells. *Am J Respir Cell Mol Biol* 14(1):104-112.
- 159. Pohl C, et al. (2009) Acute morphological and toxicological effects in a human bronchial coculture model after sulfur mustard exposure. *Toxicol Sci* 112(2):482-489.
- 160. Zhu Y, Chidekel A, & Shaffer TH (2010) Cultured human airway epithelial cells (calu-3): a model of human respiratory function, structure, and inflammatory responses. *Crit Care Res Pract*.

- 161. Foster KA, Avery ML, Yazdanian M, & Audus KL (2000) Characterization of the Calu-3 cell line as a tool to screen pulmonary drug delivery. *Int J Pharm* 208(1-2):1-11.
- 162. Zhu Y, Chidekel A, & Shaffer TH (2010) Cultured human airway epithelial cells (calu-3): a model of human respiratory function, structure, and inflammatory responses. *Crit Care Res Pract* 2010.
- 163. Tashjian AH, Jr., Weintraub BD, Barowsky NJ, Rabson AS, & Rosen SW (1973) Subunits of human chorionic gonadotropin: unbalanced synthesis and secretion by clonal cell strains derived from a bronchogenic carcinoma. *Proc Natl Acad Sci U S A* 70(5):1419-1422.
- 164. Dahiya R, *et al.* (1993) Mucin Synthesis and Secretion in Various Human Epithelial Cancer Cell Lines That Express the MUC-1 Mucin Gene. *Cancer Research* 53(6):1437-1443.
- 165. Chen E, Ruvalcaba M, Araujo L, Chapman R, & Chin W-C (2008) Ultrafine titanium dioxide nanoparticles induce cell death in human bronchial epithelial cells. *Journal of Experimental Nanoscience* 3(3):171-183.
- 166. Chen EYT, Garnica M, Wang Y-C, Chen C-S, & Chin W-C (2011) Mucin Secretion Induced by Titanium Dioxide Nanoparticles. *PLoS ONE* 6(1):e16198.
- 167. Foster KA, Oster CG, Mayer MM, Avery ML, & Audus KL (1998) Characterization of the A549 cell line as a type II pulmonary epithelial cell model for drug metabolism. *Exp Cell Res* 243(2):359-366.
- 168. Nardone LL & Andrews SB (1979) Cell line A549 as a model of the type II pneumocyte. Phospholipid biosynthesis from native and organometallic precursors. *Biochim Biophys Acta* 573(2):276-295.
- 169. Swain RJ, Kemp SJ, Goldstraw P, Tetley TD, & Stevens MM (2010) Assessment of cell line models of primary human cells by Raman spectral phenotyping. *Biophys J* 98(8):1703-1711.
- 170. Bitterle E, et al. (2006) Dose-controlled exposure of A549 epithelial cells at the air–liquid interface to airborne ultrafine carbonaceous particles. *Chemosphere* 65(10):1784-1790.
- 171. Deschl U, Vogel J, & Aufderheide M (2011) Development of an in vitro exposure model for investigating the biological effects of therapeutic aerosols on human cells from the respiratory tract. *Experimental and Toxicologic Pathology* 63(6):593-598.
- 172. Ritter D, Knebel JW, & Aufderheide M (2004) Comparative Assessment of Toxicities of Mainstream Smoke from Commercial Cigarettes. *Inhalation Toxicology* 16(10):691-700.
- 173. Nakamura Y, et al. (1995) Bronchial epithelial cells regulate fibroblast proliferation. American Journal of Physiology - Lung Cellular and Molecular Physiology 269(3):L377-L387.
- 174. Pohl C, et al. (2009) Barrier functions and paracellular integrity in human cell culture models of the proximal respiratory unit. *Eur J Pharm Biopharm* 72(2):339-349.
- 175. Swartz MA, Tschumperlin DJ, Kamm RD, & Drazen JM (2001) Mechanical stress is communicated between different cell types to elicit matrix remodeling. *Proc Natl Acad Sci U S A* 98(11):6180-6185.
- 176. Choe MM, Tomei AA, & Swartz MA (2006) Physiological 3D tissue model of the airway wall and mucosa. *Nat Protoc* 1(1):357-362.
- 177. Pageau SC, Sazonova OV, Wong JY, Soto AM, & Sonnenschein C (2011) The effect of stromal components on the modulation of the phenotype of human bronchial epithelial cells in 3D culture. *Biomaterials* 32(29):7169-7180.
- 178. Tomei AA, Choe MM, & Swartz MA (2008) Effects of dynamic compression on lentiviral transduction in an in vitro airway wall model. *Am J Physiol Lung Cell Mol Physiol* 294(1):L79-86.
- 179. Muller L, et al. (2010) Oxidative stress and inflammation response after nanoparticle exposure: differences between human lung cell monocultures and an advanced threedimensional model of the human epithelial airways. J R Soc Interface 7 Suppl 1:S27-40.
- 180. Rothen-Rutishauser BM, Kiama SG, & Gehr P (2005) A three-dimensional cellular model of the human respiratory tract to study the interaction with particles. *Am J Respir Cell Mol Biol* 32(4):281-289.

- 181. Wottrich R, Diabate S, & Krug HF (2004) Biological effects of ultrafine model particles in human macrophages and epithelial cells in mono- and co-culture. *Int J Hyg Environ Health* 207(4):353-361.
- 182. Hermanns MI, et al. (2010) An impaired alveolar-capillary barrier in vitro: effect of proinflammatory cytokines and consequences on nanocarrier interaction. J R Soc Interface 7 Suppl 1:S41-54.
- 183. Hermanns MI, Unger RE, Kehe K, Peters K, & Kirkpatrick CJ (2004) Lung epithelial cell lines in coculture with human pulmonary microvascular endothelial cells: development of an alveolocapillary barrier in vitro. *Lab Invest* 84(6):736-752.
- 184. Papritz M, et al. (2010) Side-specific effects by cadmium exposure: apical and basolateral treatment in a coculture model of the blood-air barrier. *Toxicol Appl Pharmacol* 245(3):361-369.
- 185. de Jong PM, *et al.* (1994) Ciliogenesis in human bronchial epithelial cells cultured at the airliquid interface. *American journal of respiratory cell and molecular biology* 10(3):271-277.
- 186. Kikuchi T, Shively JD, Foley JS, Drazen JM, & Tschumperlin DJ (2004) Differentiationdependent responsiveness of bronchial epithelial cells to IL-4/13 stimulation. *American Journal of Physiology - Lung Cellular and Molecular Physiology* 287(1):L119-L126.
- 187. Berube K, Prytherch Z, Job C, & Hughes T (2010) Human primary bronchial lung cell constructs: the new respiratory models. *Toxicology* 278(3):311-318.
- 188. Ryan JA (2008) Introduction to Animal Cell Culture. *Technical Bulletin, Corning Inc.* (3/08).
- 189. Chambon P (1996) A decade of molecular biology of retinoic acid receptors. *The FASEB Journal* 10(9):940-954.
- 190. Million K, et al. (2001) Effects of retinoic acid receptor-selective agonists on human nasal epithelial cell differentiation. Am J Respir Cell Mol Biol 25(6):744-750.
- 191. Biesalski HK & Nohr D (2003) Importance of vitamin-A for lung function and development. *Mol Aspects Med* 24(6):431-440.
- 192. Yoon JH, *et al.* (1999) Lysozyme expression during metaplastic squamous differentiation of retinoic acid-deficient human tracheobronchial epithelial cells. *Am J Respir Cell Mol Biol* 20(4):573-581.
- 193. Glick AB, et al. (1991) Complex regulation of TGF beta expression by retinoic acid in the vitamin A-deficient rat. *Development* 111(4):1081-1086.
- 194. Klann RC & Marchok AC (1982) Effects of retinoic acid on cell proliferation and cell differentiation in a rat tracheal epithelial cell line. *Cell Proliferation* 15(5):473-482.
- 195. Koo JS, *et al.* (1999) Restoration of the Mucous Phenotype by Retinoic Acid in Retinoid-Deficient Human Bronchial Cell Cultures: Changes in Mucin Gene Expression. *American Journal of Respiratory Cell and Molecular Biology* 20(1):43-52.
- 196. Takatsuka J, Takahashi N, & De Luca LM (1996) Retinoic Acid Metabolism and Inhibition of Cell Proliferation: An Unexpected Liaison. *Cancer Research* 56(4):675-678.
- Albright CD, Grimley PM, Jones RT, & Resau JH (2002) Differential effects of TPA and retinoic acid on cell-cell communication in human bronchial epithelial cells. *Exp Mol Pathol* 72(1):62-67.
- 198. Wu R, Zhao YH, & Chang MM (1997) Growth and differentiation of conducting airway epithelial cells in culture. *Eur Respir J* 10(10):2398-2403.
- 199. Burgel PR & Nadel JA (2004) Roles of epidermal growth factor receptor activation in epithelial cell repair and mucin production in airway epithelium. *Thorax* 59(11):992-996.
- 200. Lee J, et al. (2011) Prevention of bronchial hyperplasia by EGFR pathway inhibitors in an organotypic culture model. *Cancer Prev Res (Phila)* 4(8):1306-1315.
- 201. Fulcher ML, Gabriel S, Burns KA, Yankaskas JR, & Randell SH (2005) Well-differentiated human airway epithelial cell cultures. *Methods Mol Med* 107:183-206.
- 202. Gstraunthaler G (2003) Alternatives to the use of fetal bovine serum: serum-free cell culture. *ALTEX* 20(4):275-281.

- 203. Brunner D, et al. (2010) Serum-free cell culture: the serum-free media interactive online database. *ALTEX* 27(1):53-62.
- 204. Wiesel JM, Gamiel H, Vlodavsky I, Gay I, & Ben-Bassat H (1983) Cell attachment, growth characteristics and surface morphology of human upper-respiratory tract epithelium cultured on extracellular matrix. *European Journal of Clinical Investigation* 13(1):57-63.
- 205. Kempner ME & Felder RA (2002) A Review of Cell Culture Automation. *Journal of the Association for Laboratory Automation* 7(2):56-62.
- 206. Gryseels (2008) Considering Cell Culture Automation in Upstream Bioprocess Development. *BioProcess International* 6(11):12-16.
- 207. TAP Biosystems (www.tapbiosystems.com).
- 208. Diederichs S, et al. (2009) Dynamic cultivation of human mesenchymal stem cells in a rotating bed bioreactor system based on the Z[®]RP platform. *Biotechnology Progress* 25(6):1762-1771.
- 209. Anton F, et al. (2008) Design and Characterization of a Rotating Bed System Bioreactor for Tissue Engineering Applications. *Biotechnology Progress* 24(1):140-147.
- 210. Minuth WW, Denk L, & Glashauser A (2010) A modular culture system for the generation of multiple specialized tissues. *Biomaterials* 31(11):2945-2954.
- 211. Sbrana T & Ahluwalia A (2012) Engineering Quasi-Vivo[®] in Vitro Organ Models New Technologies for Toxicity Testing. Advances in Experimental Medicine and Biology, eds Balls M, Combes RD, & Bhogal N (Springer US), Vol 745, pp 138-153.
- 212. Aufderheide M, Knebel JW, & Ritter D (2003) An improved in vitro model for testing the pulmonary toxicity of complex mixtures such as cigarette smoke. *Experimental and Toxicologic Pathology* 55(1):51-57.
- 213. Ritter D, Knebel JW, & Aufderheide M (2001) In vitro exposure of isolated cells to native gaseous compounds Development and validation of an optimized system for human lung cells. *Experimental and Toxicologic Pathology* 53(5):373-386.
- 214. Giardinello FE, Mc LW, & Rake GW (1958) The apparent toxicity of metallic materials of construction and antifoam agents for mammalian cell lines. *Appl Microbiol* 6(1):30-35.
- 215. ECC (1993) Council Directive 93/42/EEC of 14 June 1993 concerning medical devices. *Official Journal of the European Communities* L169:1.
- 216. International_Standard (2009) Biological evaluation of medical devices- Part 5: Tests for *in vitro* cytotoxicity. *ISO 10993-5:2009(E)*
- 217. Anderson JM, *et al.* (2004) Biological Testing of Biomaterials. *Biomaterials science: an introduction to materials in medicine*, ed Ratner BD (Elsevier Academic Press, London), 2nd Ed, pp 355-359.
- 218. Hopp M, Rogaschewski S, & Groth T (2003) Testing the cytotoxicity of metal alloys used as magnetic prosthetic devices. *Journal of Materials Science: Materials in Medicine* 14(4):335-345.
- 219. Baba AI & Câtoi C (2007) Chapter 3: Tumor Cell Morphology. *Comparative Oncology*, (The Publishing House of the Romanian Academy, Bucharest).
- 220. Florea BI, Cassara ML, Junginger HE, & Borchard G (2003) Drug transport and metabolism characteristics of the human airway epithelial cell line Calu-3. *Journal of Controlled Release* 87(1–3):131-138.
- 221. Shen BQ, Finkbeiner WE, Wine JJ, Mrsny RJ, & Widdicombe JH (1994) Calu-3: a human airway epithelial cell line that shows cAMP-dependent Cl- secretion. *American Journal of Physiology Lung Cellular and Molecular Physiology* 266(5):L493-L501.
- 222. Wan H, et al. (2000) Tight junction properties of the immortalized human bronchial epithelial cell lines Calu-3 and 16HBE140. *European Respiratory Journal* 15(6):1058-1068.
- 223. Yoshikawa T, *et al.* (2010) Dynamic Innate Immune Responses of Human Bronchial Epithelial Cells to Severe Acute Respiratory Syndrome-Associated Coronavirus Infection. *PLoS ONE* 5(1):e8729.

- 224. Kaufmann O, Fietze E, Mengs J, & Dietel M (2001) Value of p63 and cytokeratin 5/6 as immunohistochemical markers for the differential diagnosis of poorly differentiated and undifferentiated carcinomas. *Am J Clin Pathol* 116(6):823-830.
- 225. Forbes B, Shah A, Martin GP, & Lansley AB (2003) The human bronchial epithelial cell line 16HBE14o- as a model system of the airways for studying drug transport. *Int J Pharm* 257(1-2):161-167.
- 226. Ehrhardt C, et al. (2002) Influence of apical fluid volume on the development of functional intercellular junctions in the human epithelial cell line 16HBE14o-: implications for the use of this cell line as an in vitro model for bronchial drug absorption studies. *Cell and Tissue Research* 308(3):391-400.
- 227. Young HW, et al. (2007) Central role of Muc5ac expression in mucous metaplasia and its regulation by conserved 5' elements. *Am J Respir Cell Mol Biol* 37(3):273-290.
- 228. Ben Khalifa Y, et al. (2011) The Human Papillomavirus E6 Oncogene Represses a Cell Adhesion Pathway and Disrupts Focal Adhesion through Degradation of TAp63î² upon Transformation. *PLoS Pathog* 7(9):e1002256.
- 229. Merrifield J, O'Donnell R, Davies DE, Djukanovic R, & Wilson SJ (2011) A panel of antibodies for identifying squamous metaplasia in endobronchial biopsies from smokers. *Biotech Histochem* 86(5):340-344.
- 230. Bahadduri PM, *et al.* (2005) Functional Characterization of the Peptide Transporter PEPT2 in Primary Cultures of Human Upper Airway Epithelium. *American journal of respiratory cell and molecular biology* 32(4):319-325.
- 231. Lin H, et al. (2007) Air-liquid interface (ALI) culture of human bronchial epithelial cell monolayers as an in vitro model for airway drug transport studies. *Journal of Pharmaceutical Sciences* 96(2):341-350.
- 232. Bals R, Beisswenger C, Blouquit S, & Chinet T (2004) Isolation and air-liquid interface culture of human large airway and bronchiolar epithelial cells. *J Cyst Fibros* 3 Suppl 2:49-51.
- 233. Nichols WW, et al. (1977) Characterization of a new human diploid cell strain, IMR-90. *Science* 196(4285):60-63.
- 234. Asaithamby A, et al. (2011) Irreparable complex DNA double-strand breaks induce chromosome breakage in organotypic three-dimensional human lung epithelial cell culture. Nucleic Acids Res 39(13):5474-5488.
- 235. Delgado O, et al. (2011) Multipotent Capacity of Immortalized Human Bronchial Epithelial Cells. *PLoS ONE* 6(7):e22023.
- 236. Koller MR, Palsson B, & Masters JRW eds (2001) *Human Cell Culture: Primary Mesenchymal Cells* (Kluwer Academic Publishers, Dordrecht), Vol V.
- 237. Bell E, et al. (1983) The Reconstitution of Living Skin. J Investig Dermatol 81(s1):2s-10s.
- 238. Mio T, et al. (1998) Human bronchial epithelial cells modulate collagen gel contraction by fibroblasts. *Am J Physiol* 274(1 Pt 1):L119-126.
- 239. Sung KE, et al. (2009) Control of 3-dimensional collagen matrix polymerization for reproducible human mammary fibroblast cell culture in microfluidic devices. *Biomaterials* 30(27):4833-4841.
- 240. Pourjavadi A & Kurdtabar M (2007) Collagen-based highly porous hydrogel without any porogen: Synthesis and characteristics. *European Polymer Journal* 43(3):877-889.
- 241. Achilli M & Mantovani D (2010) Tailoring Mechanical Properties of Collagen-Based Scaffolds for Vascular Tissue Engineering: The Effects of pH, Temperature and Ionic Strength on Gelation. *Polymers* 2(4):664-680.
- 242. Karamichos D, Brown RA, & Mudera V (2007) Collagen stiffness regulates cellular contraction and matrix remodeling gene expression. *J Biomed Mater Res A* 83(3):887-894.
- 243. Sarsour EH, Agarwal M, Pandita TK, Oberley LW, & Goswami PC (2005) Manganese superoxide dismutase protects the proliferative capacity of confluent normal human fibroblasts. *J Biol Chem* 280(18):18033-18041.

- 244. Davies DE (2009) The role of the epithelium in airway remodeling in asthma. *Proc Am Thorac Soc* 6(8):678-682.
- 245. Bolton SJ, Pinnion K, Oreffo V, Foster M, & Pinkerton KE (2009) Characterisation of the proximal airway squamous metaplasia induced by chronic tobacco smoke exposure in spontaneously hypertensive rats. *Respir Res* 10:118.
- 246. Koch L-K, et al. (2008) Stem cell marker expression in small cell lung carcinoma and developing lung tissue. *Human Pathology* 39(11):1597-1605.
- 247. Lachowicz-Scroggins ME, Boushey HA, Finkbeiner WE, & Widdicombe JH (2010) Interleukin-13–Induced Mucous Metaplasia Increases Susceptibility of Human Airway Epithelium to Rhinovirus Infection. *American Journal of Respiratory Cell and Molecular Biology* 43(6):652-661.
- 248. Travis WD (2001) Lung. *Pathology of incipient neoplasia*, eds Henson DE & Albores-Saavedra J (Oxford University Press, Inc., New York), Third Edition Ed.
- 249. Rach J, Halter B, & Aufderheide M (2013) Importance of material evaluation prior to the construction of devices for *in vitro* techniques. *EXP TOXICOL PATHOL in press*.
- 250. Cooke MJ, *et al.* (2008) Enhanced cell attachment using a novel cell culture surface presenting functional domains from extracellular matrix proteins. *Cytotechnology* 56(2):71-79.
- 251. Mullen JB, Wright JL, Wiggs BR, Pare PD, & Hogg JC (1987) Structure of central airways in current smokers and ex-smokers with and without mucus hypersecretion: relationship to lung function. *Thorax* 42(11):843-848.
- 252. Rogers DF & Rubin BK (2007) Mucolytics for COPD. *Chronic Obstructive Pulmonary Disease*, eds Stockley RA, Rennard SI, Rabe KF, & Celli B (Blackwell Publishing Ltd. , Oxford), 1. Ed, p 892.

Jessica Rach

PERSONAL DATA: Date of Birth: January Place of Birth: Gehrde Gender: Female Family Status: unmarr	y, 8, 1977 en, Germany ried, no childs
WORK EXPERIENCE:	
03/2009 – present	 <i>Cultex Laboratories GmbH</i>, Hannover, Germany Research Assistant, Focus: Cell culture
11/2005 – 02/2009	 Stiftung Tierärztliche Hochschule Hannover, Germany Graduate research in Biological Science, Topic: "Development of character-based DNA barcodes for Odonata"
01/1999 – 02/2009	 Agility Logistics GmbH (form. Geologistics GmbH), Hannover, Germany Employee (11/2000 – 02/2009 temporary worker)
UNIVERSITY EDUCATIO	DN:
10/2012 – 06/2013	 Gottfried Wilhelm Leibniz Universität Hannover, Germany Dissertation; Title: "Development of suitable <i>in vitro</i> models of the bronchial epithelium for inhalation toxicology" Graduation with doctoral degree (Dr. <i>rer. nat.</i>)
01/2005 – 10/2005	 Stiftung Tierärztliche Hochschule Hannover, Germany Diploma thesis: "Character-based DNA barcoding: Development of an DNA-based identification system for Odonata" Graduation with University-Diploma in Biological Science
2000 – 2005	 Gottfried Wilhelm Leibniz Universität Hannover, Germany Undergraduate studies in Biological Science, concentration in Molecular Biology, Zoology and Ethology
PROFESSIONAL EDUCA	TION:
01/1999 – 10/2000	 Geologistics GmbH, Hannover, Germany Training as freight forwarding agent Certified by Industrie und Handelskammer Hannover, Germany
SCHOOL EDUCATION: 1983 – 1989	Grundschule Am Castrum, Gehrden, Germany
1989 – 1991	Orientierungsstufe, Gehrden, Germany
1991 – 1996	 Matthias-Claudius-Gymnasium, Gehrden, Germany Higher education entrance qualification

List of publications and posters

Publications:

- Rach J, Budde J, Mohle N, and Aufderheide M. 2013. Direct exposure at the air-liquid interface: evaluation of an in vitro approach for simulating inhalation of airborne substances. J Appl Toxicol.
- <u>Rach J</u>, Halter B, and Aufderheide M. 2013. Importance of material evaluation prior to the construction of devices for in vitro techniques. *Exp Toxicol Pathol*.
- Bergmann T & <u>Rach J</u>, Damm S, Desalle R, Schierwater B, and Hadrys H. 2013. The potential of distance-based thresholds and character-based DNA barcoding for defining problematic taxonomic entities by CO1 and ND1. *Mol Ecol Resour*.
- Rach J, Desalle R, Sarkar IN, Schierwater B, and Hadrys H. 2008. Character-based DNA barcoding allows discrimination of genera, species and populations in Odonata. *Proc Biol Sci* 275(1632):237-247.

Posters:

Rach_J, Aufderheide M. 2011. Importance of biomaterial testing prior to the construction of devices for *in vitro* techniques. *EuroTox 2011, Paris;* Abstract Book.

Möhle N, <u>Rach J</u>, Scheffler S, Budde D-J, Aufderheide M. 2011. Need and Perspectives for the Implementation of Relevant In Vitro Methods in the Field of Inhalation Toxicology. *The 8th World Congress on Alternatives and Animal Use in the Life Sciences, Montreal;* Abstract Book.

Rach J, Schierwater B, Hadrys H. 2007. A two-marker approach for character-based DNA barcoding extends and hardens identification success in Odonata. *DNA Barcoding in Europe, Naturalis, Leiden, The Netherland;,* Conference Book.

<u>Rach J,</u> Schierwater B, Hadrys H. 2005. Identification of dragonflies: DNA barcoding. 98 Jahresversammlung der Deutschen Zoologischen Gesellschaft, Universität Bayreuth; Abstract Book.